The zoogeomorphology of case-building caddisfly larvae

by

Richard Mason

A Doctoral thesis submitted in partial fulfilment of the requirements for the award of

Doctor of Philosophy of Loughborough University

(June 2020)

© Richard Mason 2020

Abstract

Caddisfly (Trichoptera) are an abundant and widespread aquatic insect group. Caddisfly larvae of most species build cases from silk and fine sediment at some point in their lifecycle. Casebuilding caddisfly have the potential to modify the distribution and transport of sediment by: 1) altering sediment properties through case construction, and 2) transporting sediment incorporated into cases over the riverbed. This thesis investigates, for the first time, the effects of bioconstruction by case-building caddisfly on fluvial geomorphology. The research was conducted using two flume experiments to understand the mechanisms of caddisfly zoogeomorphology (case construction and transporting sediment), and two field investigations that increase the spatial and temporal scale of the research. Caddisfly cases varied considerably in mass between species (0.001 g - 0.83 g) and grain sizes used ($D_{50} = 0.17$ mm - 4 mm). As a community, caddisfly used a wide range of grain-sizes in case construction (0.063 mm - 11 mm), and, on average, the mass of incorporated sediment was 38 g m⁻², in a gravel-bed stream. This sediment was aggregated into biogenic particles (cases) which differed in size and shape from their constituent grains. A flume experiment determined that empty cases of some caddisfly species (tubular case-builders; Limnephilidae and Sericostomatidae) were more mobile than their incorporated sediment, but that dome shaped Glossosomatidae cases moved at the same entrainment threshold as their constituent grains, highlighting the importance of case design as a control on caddisfly zoogeomorphology. A second flume experiment found that crawling Glossosomatidae larvae transport sand vertically upwards, from sheltered interstices below gravel particles, onto the surface of armoured gravel-beds, increasing the hydraulic exposure of this sediment. As a result of both larvae crawling and case entrainment, case-building caddisfly were responsible for over 30% of coarse sand transport during low to medium discharge conditions in a gravel-bed stream. Tubular case-builders (especially Limnephilidae) and dome case-builders (Glossosomatidae) were particularly important zoogeomorphic agents, using and transporting substantial amounts of coarse sand and fine gravel. This research has shown that case bioconstruction by caddisfly larvae may have a destabilising effect on sand in gravel-bed rivers. The thesis uses case design to conceptualise and understand caddisfly zoogeomorphology under differing biotic and abiotic controls. Future research should consider sediment stabilisation associated with caddisfly pupation, and the relative importance of destabilisation and stabilisation across spatial and temporal gradients. The broad geographic spread, functional diversity, and commonly high abundances of case-building caddisfly mean that they may have important and widespread implications for sediment dynamics in rivers.

> ... 11

Acknowledgements

Thank you to everybody who has made my PhD experience special. To my supervisors, thank you for this opportunity and for the time and experience you have invested in this project and in me as a researcher. Thank you to Steve Rice, for your enthusiasm which has kept me motivated, your expert guidance, and your dedication to puzzling over the challenges of my PhD. Thank you to Paul Wood, for your humour and care during the stressful times, going above and beyond and offering the practical advice that kept the thesis on track (well nearly!). Thank you to Matt Johnson, for sparking my initial interest in zoogeomorphology and for the help and support which you have provided, during my undergraduate and postgraduate research.

Thanks also go to everyone at Loughborough Geography who have made the experience so rewarding. Particular thanks to those who have solved my (many) obscure research obstacles. To Richard Harland who solved most of them, and to Rebecca McKenzie, Daniel Gschwentner and Bailey Stickings for their incredible support with laboratory work. Thank you also to Davide Vettori for lending your flume and hydraulics expertise. Thank you to NERC CENTA for providing the funding for this project and to my CENTA cohort, a fantastic group of fellow PhD students.

A huge thank you also to my PhD colleagues. From day one, sitting in the pub at 3 pm because we didn't know what we supposed to be doing, to today, you have been there for me. Special thanks go to Harry Sanders, for the crazy escapades and injecting into the PhD experience so much enthusiasm. I will never forget eating spaghetti hoops using only a cable tie, whilst sat in the boot of a departmental car in a Scottish layby. Thanks to Ellen Goddard for your support, for the fun times at King Edward Rd. and for help with statistics. Thanks to Keechy Akkerman for the climbing, cats and whisky. Thank you also to a host of other people who have helped with fieldwork or laboratory work; Beth Worley, Ciara Dwyer, Maud van Soest, Tom Stanton, and everyone else who has been an important part of the last 4 years.

To my Mum and Dad, thank you for the support over my PhD years and for my initial excitement for the natural environment. Without you patiently waiting for 3-year-old me to stomp through puddles or inspect beetles, I might not have an interest in wildlife and rivers and might have a 'proper job' by now! Last, a special thank you to my partner, Hazel Wilson, who has been my rock and inspiration throughout the PhD, here's to many more adventures!

Table of contents

Acknowledgements iii Table of contents iv List of tables viii List of figures viii Chapter 1. Introduction 1 1.1 Research context 1 1.2 Huvial zoogcomorphology 6 1.3.1 Nerretbrate zoogcomorphology 6 1.3.1 Warrine 7 1.3.3 Sediment detabilization in freshwater babitats: biotarbation and bioerosion 8 1.4.4 Caddisfly biology and case-building behaviour 12 1.4.1 Caddisfly diversity and life bistory 12 1.4.2 Caddisfly arbiteture and bioconstruction 14 1.5 Aim and objectives 17 1.6 Justification of aim and objectives 17 1.6 Justification of sediment use by the case building caddisfly 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly 23 2.1 Introduction 24 2.2 Methods 26 2.2.1 Laboratory analysis 27 28 23.3 Spatial variability in sediment use by the caddify community 29 2.3.3 Spatial variability in sediment use by the caddify community 29 23.4 Is caddify abundance a control on the mass of sediment used? 29	Abstract	ii
Table of contents iv List of tables vii List of figures viii Chapter 1. Introduction 1 1.1. Research context 1 1.2. Fluvial zoogeomorphology 6 1.3.1. Terrstrial 6 1.3.2. Marine 7 1.3.3.3 Sediment distabilisation in freshwater babitats: bioturbation and bioerosion 8 1.3.4. Sediment stabilisation in freshwater babitats: bioturbation and bioprotection 10 1.4. Caddisfly diversity and life bitary 12 1.4.1. Caddisfly diversity and life bitary 12 1.4.2. Caddify grachitecture and bioconstruction 14 1.5.1 Sim and objectives 17 1.6 Justification of aim and objectives 17 1.7 Thesis structure 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly 23 2.1.1 Introduction 24 2.2.2 Laboratory analysis 27 2.3.3 Spatial renability in sediment use by the case building caddisfly 28 2.3.1 Sediment use by individual taxa 28 2.3.2 Sediment use by the whole caddify community in a Surber sample 28 2.3.3 Spatial renab	Acknowledgements	iii
List of fables vii List of figures viii List of figures viii Chapter 1. Introduction 1 1.1 Research context 1 1.2 Fluvial zoogeomorphology 3 1.3 Invertebrate zoogeomorphology 6 1.3.1 Terrestrial 6 1.3.2 Marine 7 1.3.3 Sediment distilization in freshwater babitats: biourbation and bioprotection 10 1.4 Caddisfly diversity and life bistory 12 1.4.1 Caddisfly diversity and life bistory 12 1.4.2 Caddisfly diversity and life bistory 17 1.5 Aim and objectives 17 1.6 Justification of aim and objectives 17 1.7 Thesis structure 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community 23 2.1 Introduction 24 2.2 Methods 26 2.3.1 Sediment use by individual taxa 28 2.3.2 Sediment use by individual taxa 28 2.3.3 Spatial variability in sediment use 30 2.4.1 Sediment use by individual caddisfly community 30 2.5.1 Selation between sediment availabilit	Table of contents	iv
List of figures viii Chapter 1. Introduction 1 1.1 Research context 1 1.2 Fluvial zoogeomorphology 6 1.3.1 Ivertebrate zoogeomorphology 6 1.3.1 Warine 7 1.3.3 Sediment destabilisation in freshwater babitats: bioturbation and bioerosion 8 1.4.4 Caddisfly diversity and life bistory 12 1.4.1 Caddisfly diversity and life bistory 12 1.4.1 Caddisfly diversity and life bistory 12 1.4.1 Caddisfly diversity and life bistory 12 1.4.2 Caddisfly architecture and bioconstruction 14 1.5 Justification of aim and objectives 17 1.6 Justification of aim and objectives 17 1.7 Thesis structure 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community 23 2.1 Introduction 24 2.2.1 Vield samphing 26 2.3.2 Sediment use by individual taxa 28 2.3.1 Sediment use by individual taxa 28 2.3.1 Sediment use by the whole caddisfly community 29 2.3.4 Li caddisfly andmater a control on the mass of sediment use 30 2.4.1 S	List of tables	vii
Chapter 1. Introduction 1 1.1 Research context 1 1.2 Fluvial zoogeomorphology 3 1.3 Invertebrate zoogeomorphology 6 1.3.1 Terrestrial 6 1.3.2 Marine 7 1.3.3 Sediment destabilisation in freshwater babitats: biotonstruction and bioerosion 8 1.4.4 Caddisfly diversity and life bistory 12 1.4.1 Caddisfly diversity and life bistory 12 1.4.2 Caddisfly architecture and bioconstruction 14 1.5 Aim and objectives 17 1.6 Justification of aim and objectives 17 1.7 Thesis structure 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community community 23 2.1 Introduction 24 2.2 Methods 26 2.2.1 Field sampling 26 2.3.2 Sediment use by individual taxa 28 2.3.3 Statial variability in sediment use by the caddisfly community 29 2.3.4 is caddisfly abundance a control on the mass of sediment used? 29 2.4 Results 30 24 2.5 Discussion 40 2.4.1 Sediment	List of figures	viii
1.1 Research context 1 1.2 Fluvial zoogeomorphology 3 1.3 Invertebrate zoogeomorphology 6 1.3.1 Terrestrial 6 1.3.2 Marine 7 1.3.3 Sediment destabilisation in freshwater babitats: bioconstruction and bioerosion 8 1.3.4 Sediment stabilisation in freshwater babitats: bioconstruction and bioerosion 10 1.4 Caddisfly diversity and life history 12 1.4.1 Caddisfly diversity and life history 12 1.4.1 Caddifyf architecture and bioconstruction 14 1.5 Aim and objectives 17 1.6 Justification of aim and objectives 17 1.7 Thesis structure 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community 23 2.1 Introduction 24 2.2.1 Vield sampling 26 2.2.2.1 Vield sampling 26 2.3.2 Sediment use by individual taxa 28 2.3.1 Sediment use by individual taxa 29 2.3.4 Is caddisfly abundance a control on the mass of sediment used? 29 2.3.5 Relation between sediment availability and sediment use? 30 2.4 Results 30 2.	Chapter 1. Introduction	1
1.2 Fluvial zoogeomorphology 3 1.3 Invertebrate zoogeomorphology 6 1.3.1 Terrestrial 6 1.3.2 Marine 7 1.3.3 Sediment destabilisation in freshwater babitats: bioturbation and bioprotection 10 1.4 Caddisfly biology and case-building behaviour 12 1.4.1 Caddisfly diversity and life bistory 12 1.4.2 Caddisfly architecture and bioconstruction 14 1.5 A im and objectives 17 1.6 Justification of aim and objectives 17 1.7 Thesis structure 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community 2.2 Methods 26 2.2.1 Field sampling 26 2.2.2 Laboratory analysis 27 2.3 Data Analysis 28 2.3.1 Sediment use by the whole caddisfly community in a Surber sample 28 2.3.3 Spatial variability in sediment use by the caddisfly community 29 2.3.4 Lie caddisfly abundance a control on the mass of sediment use? 29 2.3.5 Relation between sediment availability and sediment use? 30 2.4.1 Sediment use by the whole caddisfly community 30 2.4.1 Sediment use by the whole	1.1 Research context	1
1.3 Invertebrate zoogeomorphology 6 1.3.1 Terrestrial 6 1.3.2 Marine 7 1.3.3 Cadiment detabilisation in freshwater babitats: bioturbation and bioprotection 7 1.3.4 Sediment detabilisation in freshwater babitats: bioturbation and bioprotection 70 1.4 Caddisfly diversity and life bistory 12 1.4.1 Caddisfly diversity and life bistory 12 1.4.2 Caddisfly architecture and bioconstruction 14 1.5 Aim and objectives 17 1.6 Justification of aim and objectives 17 1.7 Thesis structure 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community 2.1 Introduction 24 2.2 Methods 26 2.2.1 Field sampling 26 2.2.2 Laboratory analysis 27 2.3 Data Analysis 28 2.3.1 Sediment use by the vibole caddisfly community in a Surber sample 28 2.3.3 Spatial variability in seliment use if y the caddisfly community 29 2.3.4 Is ediment use by individual caddisfly community 30 2.4.1 Sediment use by individual caddisfly taxa 30 2.4.1 Sediment use by individual caddisfly taxa<	1.2 Fluvial zoogeomorphology	3
1.3.1 Terrestrial 6 1.3.2 Marine 7 1.3.3 Sediment destabilisation in freshwater babitats: bioturbation and bioerosion 8 1.3.4 Sediment stabilisation in freshwater babitats: bioturbation and bioprotection 10 1.4 Caddisfly diversity and life bistory 12 1.4.1 Caddisfly diversity and life bistory 12 1.4.2 Caddisfly architecture and bioconstruction 14 1.5 Aim and objectives 17 1.6 Justification of aim and objectives 17 1.7 Thesis structure 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community community 23 2.1 Introduction 24 2.2 Methods 26 2.2.1 Field sampling 26 2.3.1 Sediment use by individual taxa 28 2.3.2 Sediment use by the whole caddisfly community in a Surber sample 28 2.3.3 Spatial variability in sediment use by the eaddisfly community 29 2.3.4 Results 30 2.4.1 Sediment use by individual taxa 30 2.4.1 Sediment use by individual caddisfly community 30 2.4.2 Sediment use by individual caddisfly community 30 <	1.3 Invertebrate zoogeomorphology	6
1.3.2 Marine 7 1.3.3 Sediment destabilisation in freshmater babitats: bioturbation and bioerosion 8 1.3.4 Sediment stabilisation in freshmater babitats: biotonstruction and bioerosion 10 1.4 Caddisfly biology and case-building behaviour 12 1.4.1 Caddisfly diversity and life bistory 12 1.4.2 Caddisfly architecture and bioconstruction 14 1.5 Aim and objectives 17 1.6 Justification of aim and objectives 17 1.7 Thesis structure 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community 23 2.1 Introduction 24 2.2 Methods 26 2.3.1 Stediment use by individual taxa 28 2.3.2 Sediment use by the whole caddisfly community in a Surber sample 28 2.3.2 Sediment use by the whole caddisfly community 29 2.3.4 Is caddisfly abundance a control on the mass of sediment used? 29 2.3.5 Relation between sediment availability and sediment use 30 2.4 Results 30 2.4.1 Sediment use by individual caddisfly community 30 2.4.1 Sediment use by the whole caddisfly community 30 2.4.1 Sediment use by individual caddis	1.3.1 Terrestrial	6
1.3.3 Sediment destabilisation in freshwater babitats: bioturbation and bioerosion 8 1.3.4 Sediment stabilisation in freshwater babitats: bioconstruction and bioprotection 10 1.4 Caddisfly diversity and life bistory 12 1.4.1 Caddisfly diversity and life bistory 12 1.4.2 Caddisfly architecture and bioconstruction 14 1.5 Aim and objectives 17 1.6 Justification of aim and objectives 17 1.7 Thesis structure 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community community 23 2.1 Introduction 24 2.2 Methods 26 2.2.1 Field sampling 26 2.2.2 Laboratory analysis 27 2.3 Data Analysis 28 2.3.3 Spatial variability in sediment use by the caddisfly community 29 2.3.4 Is caddisfly abundance a control on the mass of sediment used? 29 2.3.5 Relation between sediment availability and sediment use? 30 2.4.1 Sediment use by individual caddisfly community 30 2.4.1 Sediment use by the whole caddisfly community 30 2.3.5 Relation between sediment availability and sediment use? 30 </td <td>1.3.2 Marine</td> <td>7</td>	1.3.2 Marine	7
1.3.4 Sediment stabilisation in freshwater habitats: bioconstruction and bioprotection 10 1.4 Caddisfly biology and case-building behaviour 12 1.4.1 Caddisfly diversity and life bistory 12 1.4.2 Caddisfly architecture and bioconstruction 14 1.5 Aim and objectives 17 1.6 Justification of aim and objectives 17 1.7 Thesis structure 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community 24 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community 2.1 Field sampling 2.2.2 Laborato	1.3.3 Sediment destabilisation in freshwater habitats: bioturbation and bioerosion	8
1.4 Caddisfly biology and case-building behaviour 12 1.4.1 Caddisfly dirersity and life bistory 12 1.4.2 Caddisfly architecture and bioconstruction 14 1.5 Aim and objectives 17 1.6 Justification of aim and objectives 17 1.7 Thesis structure 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community 23 2.1 Introduction 24 2.2 Methods 25.1 Field sampling 26 2.2.1 Field sampling 26 2.2.2 Laboratory analysis 27 2.3 Data Analysis 28 2.3.1 Sediment use by individual taxa 28 2.3.2 Sediment use by the whole caddisfly community in a Surber sample 28 2.3.3 Spatial pariability in sediment use by the caddisfly community 29 2.3.4 Is caddisfly abundance a control on the mass of sediment used? 29 2.4.1 Sediment use by individual caddisfly community 30 2.4.1 Sediment use by individual caddisfly community 30 2.4.2 Sediment use by individual caddisfly community 30 2.4.3 Spatial nariability in community sediment use 36 2.4.1 Sediment use by individual caddi	1.3.4 Sediment stabilisation in freshwater habitats: bioconstruction and bioprotection	10
1.4.1 Caddisfly diversity and life bistory 12 1.4.2 Caddisfly arbitecture and bioconstruction 14 1.5 Aim and objectives 17 1.6 Justification of aim and objectives 17 1.7 Thesis structure 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community 23 21 1.1 Introduction 24 2.2 Methods 26 2.2.1 Field sampling 26 2.2.2 Laboratory analysis 27 2.3 Data Analysis 27 2.3 Data Analysis 28 2.3.1 Sediment use by individual taxa 28 2.3.2 Sediment use by the whole caddisfly community in a Surber sample 28 2.3.3 Spatial variability in sediment use by the caddisfly community 29 2.3.4 Is caddisfly abundance a control on the mass of sediment used? 29 2.3.5 Relation between sediment availability and sediment use? 30 2.4.1 Sediment use by individual caddisfly community 30 2.4.2 Sediment use by individual caddisfly community 30 2.4.1 Sediment use by individual caddisfly community 30 2.4.2 Sediment use by individual caddisfly taxa 30 </td <td>1.4 Caddisfly biology and case-building behaviour</td> <td>12</td>	1.4 Caddisfly biology and case-building behaviour	12
1.4.2 Caddisfly architecture and bioconstruction 14 1.5 Aim and objectives 17 1.6 Justification of aim and objectives 17 1.7 Thesis structure 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly community 23 2.1 Introduction 24 2.2 Methods 2.1 Introduction 24 2.2 Methods 26 2.2.1 Field sampling 26 2.2.2 Laboratory analysis 27 2.3 Data Analysis 28 2.3.1 Sediment use by individual taxa 28 2.3.2 Sediment use by individual taxa 29 2.3.4 Is caddisfly avaiability in sediment use by the caddisfly community in a Surber sample 28 2.3.3 Spatial variability in sediment use by the caddisfly community 29 2.3.4 Is caddisfly abundance a control on the mass of sediment used? 29 2.3.5 Relation between sediment availability and sediment use 30 2.4.1 Sediment use by individual caddisfly taxa 30 2.4.2 Sediment use by the whole caddisfly community 30 2.4.3 Spatial variability in community sediment use 36 2.4.1 Sediment use by individual caddisfly taxa	1.4.1 Caddisfly diversity and life history	12
1.5 Aim and objectives171.6 Justification of aim and objectives171.7 Thesis structure20Chapter 2. A spatial quantification of sediment use by the case building caddisfly community232.1 Introduction242.2 Methods262.2.1 Field sampling262.2.2 Laboratory analysis272.3 Data Analysis282.3.1 Sediment use by individual taxa282.3.2 Sediment use by individual taxa282.3.3 Spatial variability in sediment use by the caddisfly community292.3.4 Is caddisfly abundance a control on the mass of sediment use?292.3.5 Relation between sediment availability and sediment use302.4.1 Sediment use by the whole caddisfly community302.4.2 Sediment use by the whole caddisfly taxa302.4.3 Spatial variability in community sediment use362.4.4 Results302.4.5 Relation between sediment availability and sediment use362.4.4 Relation between sediment availability and sediment use362.4.5 Relation between sediment availability and sediment use362.4.5 Relation between sediment availability and sediment use362.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.3 Spatial variability in community sediment use372.5 Discussion402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa3.5.4 Resultion to the abundance of taxa and availability of sedi	1.4.2 Caddisfly architecture and bioconstruction	14
1.6 Justification of aim and objectives 17 1.7 Thesis structure 20 Chapter 2. A spatial quantification of sediment use by the case building caddisfly 23 community 23 2.1 Introduction 24 2.2 Methods 26 2.2.1 Field sampling 26 2.2.2 Laboratory analysis 27 2.3 Data Analysis 28 2.3.1 Sediment use by individual taxa 28 2.3.2 Sediment use by the whole caddisfly community in a Surber sample 28 2.3.3 Spatial variability in sediment use by the caddisfly community 29 2.3.4 Is caddisfly abundance a control on the mass of sediment used? 29 2.3.5 Relation between sediment availability and sediment use 30 2.4.1 Sediment use by individual caddisfly taxa 30 2.4.1 Sediment use by individual caddisfly taxa 30 2.4.2 Sediment use by individual caddisfly community 30 2.4.3 Spatial variability in community sediment use 36 2.4.1 Sediment use by individual caddisfly community 30 2.4.2 Sediment use by the caddisfly community 30 2.4.3 Spatial variability in community sediment use 37 2.5 Discussion	1.5 Aim and objectives	17
1.7 Thesis structure20Chapter 2. A spatial quantification of sediment use by the case building caddisfly community232.1 Introduction242.2 Methods262.2.1 Field sampling262.2.2 Laboratory analysis272.3 Data Analysis282.3.1 Sediment use by individual taxa282.3.2 Sediment use by the whole caddisfly community in a Surber sample282.3.3 Spatial variability in sediment use by the caddisfly community292.3.4 Is caddisfly abundance a control on the mass of sediment used?292.3.5 Relation between sediment availability and sediment use302.4.1 Sediment use by individual caddisfly community302.4.2 Sediment use by individual caddisfly community302.4.3 Spatial variability in community sediment use362.4.4 Relation between sediment availability and sediment use372.5 Discussion402.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment43	1.6 Justification of aim and objectives	17
Chapter 2. A spatial quantification of sediment use by the case building caddisfly 23 community 24 2.1 Introduction 24 2.2 Methods 26 2.2.1 Field sampling 26 2.2.2 Laboratory analysis 27 2.3 Data Analysis 28 2.3.1 Sediment use by individual taxa 28 2.3.2 Sediment use by individual taxa 28 2.3.3 Spatial variability in sediment use by the caddisfly community in a Surber sample 28 2.3.4 Is caddisfly abundance a control on the mass of sediment used? 29 2.3.5 Relation between sediment availability and sediment use 30 2.4.1 Sediment use by individual caddisfly taxa 30 2.4.2 Sediment use by individual caddisfly taxa 30 2.4.3 Spatial variability in community sediment use 36 2.4.4 Relation between sediment availability and sediment use 36 2.4.5 Relation between sediment availability and sediment use 36 2.4.5 Sediment use by individual caddisfly community 30 2.4.5 Sediment use by individual caddisfly community 30 2.4.5 Relation between sediment availability and sediment use 37 2.5.5 Discussion 40	1.7 Thesis structure	20
2.1 Introduction242.2 Methods262.2.1 Field sampling262.2.2 Laboratory analysis272.3 Data Analysis282.3.1 Sediment use by individual taxa282.3.2 Sediment use by the whole caddisfly community in a Surber sample282.3.3 Spatial variability in sediment use by the caddisfly community292.3.4 Is caddisfly abundance a control on the mass of sediment used?292.3.5 Relation between sediment availability and sediment use302.4.1 Sediment use by individual caddisfly taxa302.4.2 Sediment use by the whole caddisfly community302.4.3 Spatial variability in community sediment use362.4.4 Relation between abundance and sediment use362.4.5 Relation between abundance and sediment use372.5 Discussion402.5.1 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	community 2.1 Introduction	23
2.2 Methods262.2.1 Field sampling262.2.2 Laboratory analysis27 2.3 Data Analysis282.3.1 Sediment use by individual taxa282.3.2 Sediment use by the whole caddisfly community in a Surber sample282.3.3 Spatial variability in sediment use by the caddisfly community292.3.4 Is caddisfly abundance a control on the mass of sediment used?292.3.5 Relation between sediment availability and sediment use302.4.1 Sediment use by individual caddisfly taxa302.4.2 Sediment use by individual caddisfly community302.4.3 Spatial variability in community sediment use362.4.4 Relation between abundance and sediment use362.4.5 Relation between sediment availability and sediment use362.4.5 Relation between sediment availability and sediment use362.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by individual taxa and the case-building caddisfly community402.5.3 Spatial variability in community sediment use37 2.5. Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by individual taxa and the case-building caddisfly community402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2.1 Introduction	24
2.2.2 Laboratory analysis272.3 Data Analysis282.3.1 Sediment use by individual taxa282.3.2 Sediment use by the whole caddisfly community in a Surber sample282.3.3 Spatial variability in sediment use by the caddisfly community292.3.4 Is caddisfly abundance a control on the mass of sediment used?292.3.5 Relation between sediment availability and sediment use302.4 Results302.4.1 Sediment use by individual caddisfly taxa302.4.2 Sediment use by individual caddisfly taxa302.4.3 Spatial variability in community sediment use362.4.4 Relation between abundance and sediment use362.4.5 Relation between sediment availability and sediment use362.4.5 Relation between sediment availability and sediment use362.4.5 Relation between sediment availability and sediment use362.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by individual taxa and the case-building caddisfly community402.5.3 Spatial variability in community sediment use372.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2.2 Methods 2.2.1 Field sampling	20 26
2.3 Data Analysis282.3 Data Analysis282.3.1 Sediment use by individual taxa282.3.2 Sediment use by the whole caddisfly community in a Surber sample282.3.3 Spatial variability in sediment use by the caddisfly community292.3.4 Is caddisfly abundance a control on the mass of sediment used?292.3.5 Relation between sediment availability and sediment use302.4 Results302.4.1 Sediment use by individual caddisfly taxa302.4.2 Sediment use by the whole caddisfly community302.4.3 Spatial variability in community sediment use362.4.4 Relation between abundance and sediment use362.4.5 Relation between sediment availability and sediment use362.4.5 Relation between abundance and sediment use362.4.5 Relation between sediment availability and sediment use372.5 Discussion402.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2.2.1 I icia sumpling 2.2.2 I aboratory analysis	20
2.3.1 Sediment use by individual taxa282.3.2 Sediment use by the whole caddisfly community in a Surber sample282.3.3 Spatial variability in sediment use by the caddisfly community292.3.4 Is caddisfly abundance a control on the mass of sediment used?292.3.5 Relation between sediment availability and sediment use302.4 Results302.4.1 Sediment use by individual caddisfly taxa302.4.2 Sediment use by individual caddisfly community302.4.3 Spatial variability in community sediment use362.4.4 Relation between abundance and sediment use362.4.5 Relation between sediment availability and sediment use362.4.5 Relation between sediment availability and sediment use362.4.5 Relation between abundance and sediment use362.4.5 Relation between sediment availability and sediment use372.5 Discussion402.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2 3 Data Analysis	28
2.3.1 Solument use by the whole caddisfly community in a Surber sample282.3.2 Sediment use by the whole caddisfly community292.3.4 Is caddisfly abundance a control on the mass of sediment used?292.3.5 Relation between sediment availability and sediment use302.4 Results302.4.1 Sediment use by individual caddisfly taxa302.4.2 Sediment use by individual caddisfly community302.4.3 Spatial variability in community sediment use362.4.4 Relation between abundance and sediment use362.4.5 Relation between sediment availability and sediment use372.5 Discussion402.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2 3 1 Sediment use by individual taxa	28
2.3.3 Spatial variability in sediment use by the caddisfly community292.3.4 Is caddisfly abundance a control on the mass of sediment used?292.3.5 Relation between sediment availability and sediment use302.4 Results302.4.1 Sediment use by individual caddisfly taxa302.4.2 Sediment use by the whole caddisfly community302.4.3 Spatial variability in community sediment use362.4.4 Relation between abundance and sediment use362.4.5 Relation between sediment availability and sediment use362.4.6 Relation between abundance and sediment use362.4.7 Sediment use by individual taxa and the case-building caddisfly community402.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2.3.2 Sediment use by the whole caddisfly community in a Surber sample	_0 28
2.3.4 Is caddisfly abundance a control on the mass of sediment used?292.3.5 Relation between sediment availability and sediment use302.4 Results302.4.1 Sediment use by individual caddisfly taxa302.4.2 Sediment use by the whole caddisfly community302.4.3 Spatial variability in community sediment use362.4.4 Relation between abundance and sediment use362.4.5 Relation between abundance and sediment use362.4.5 Relation between sediment availability and sediment use362.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment43	2.3.3 Spatial variability in sediment use by the caddisfly community	29
2.3.5 Relation between sediment availability and sediment use302.4 Results302.4.1 Sediment use by individual caddisfly taxa302.4.2 Sediment use by the whole caddisfly community302.4.3 Spatial variability in community sediment use362.4.4 Relation between abundance and sediment use362.4.5 Relation between sediment availability and sediment use372.5 Discussion402.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2.3.4 Is caddisfly abundance a control on the mass of sediment used?	29
2.4 Results302.4.1 Sediment use by individual caddisfly taxa302.4.2 Sediment use by the whole caddisfly community302.4.3 Spatial variability in community sediment use362.4.4 Relation between abundance and sediment use362.4.5 Relation between sediment availability and sediment use372.5 Discussion402.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2.3.5 Relation between sediment availability and sediment use	30
2.4.1 Sediment use by individual caddisfly taxa302.4.2 Sediment use by the whole caddisfly community302.4.3 Spatial variability in community sediment use362.4.4 Relation between abundance and sediment use362.4.5 Relation between sediment availability and sediment use372.5 Discussion402.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2.4 Results	30
2.4.2 Sediment use by the whole caddisfly community302.4.3 Spatial variability in community sediment use362.4.4 Relation between abundance and sediment use362.4.5 Relation between sediment availability and sediment use372.5 Discussion402.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2.4.1 Sediment use by individual caddisfly taxa	30
2.4.3 Spatial variability in community sediment use362.4.4 Relation between abundance and sediment use362.4.5 Relation between sediment availability and sediment use372.5 Discussion402.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2.4.2 Sediment use by the whole caddisfly community	30
2.4.4 Relation between abundance and sediment use362.4.5 Relation between sediment availability and sediment use372.5 Discussion402.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2.4.3 Spatial variability in community sediment use	36
2.4.5 Relation between sediment availability and sediment use372.5 Discussion402.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2.4.4 Relation between abundance and sediment use	36
2.5 Discussion402.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2.4.5 Relation between sediment availability and sediment use	37
2.5.1 Sediment use by individual taxa and the case-building caddisfly community402.5.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2.5 Discussion	40
2.3.2 Sediment use by key taxa and potential zoogeomorphological importance402.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa412.5.4 Sediment use in relation to the abundance of taxa and availability of sediment422.5.5 Methodological discussion43	2.5.1 Sediment use by individual taxa and the case-building caddisfly community	40
2.3.5 Spanal variabuly in community seament use and the distribution of case-outloing caldisfly taxa 41 2.5.4 Sediment use in relation to the abundance of taxa and availability of sediment 42 2.5.5 Methodological discussion 43	2.3.2 Seatment use by key taxa and potential zoogeomorphological importance	40
2.5.5 Methodological discussion 42	2.3.3 Spanae variability in community seamen use and the distribution of case-building caddisfly taxa	41 12
	2.5.5 Methodological discussion	<i>∓∠</i> 43

iv

44

2.6 Summary

Chapter 3. The effect of caddisfly case construction and case design on the entrainment		
of incorporated sediment.	47	
3.1 Introduction	47	
3.1.1 Controls on sediment transport	47	
3.1.2 Entrainment of caddisfly and the role of the case	48	
3.2 Methods	50	
3.2.1 Selection of species	50	
3.2.2 Flume setup	51	
3.2.3 Flume procedure	53	
3.3 Data Analysis	55	
3.3.1 Hydraulic stages	55	
3.3.2 Entrainment	57	
3.3.3 Caddisfly case design	58	
3.4 Results	61	
3.4.1 Entrainment of cases versus constituent sediment	61	
3.4.2 Differences in case entrainment between species	63	
3.4.3 Differences in constituent sediment entrainment between species	63	
3.4.4 The role of case shape for case entrainment	66	
3.5 Discussion	67	
3.5.1 Transport of caddisfly cases versus sediment (Question 1)	69	
3.5.2 Entrainment of loose sediment	70	
3.5.3 Movement of cases, role of mass and shape (Question 2)	70	
3.5.4 Geomorphological implications of caddisfly case construction	71	
3.6 Summary	72	

Chapter 4. Vertical sand displacement by Glossosomatidae Agapetus fuscipes larvae	75
4.1 Introduction	75
4.1.1 Sand transport in armoured gravel-bed rivers	75
4.1.2 Context of this Chapter	78
4.2 Methods	79
4.2.1 Flume setup	79
4.2.2 Sediment tray setup	80
4.2.3 Experimental procedure	82
4.2.4 Caddisfly and case analysis	83
4.2.5 Data analysis	84
4.3 Results	85
4.4 Discussion	91
4.4.1 Vertical displacement of sand by Agapetus fuscipes larvae	91
4.4.2 Influence of grain protrusion on sand displacement	92
4.4.3 Influence of flow velocity on sand displacement	93
4.4.4 Combined influence of gravel protrusion and high flow velocity on larvae positioning	93
4.4.5 Zoogeomorphic effects of sand displacement	94
4.5 Summary	96

Chapter 5. The relative contributions of hydraulics and case-building caddisfly to	
bedload transport in a gravel-bed stream	98
5.1 Introduction	98
5.2 Methods	100
5.2.1 Field site	100
5.2.2 Pit trap design and deployment	100
5.2.3 Pit trap sampling	103
5.2.4 Environmental measurements	104
5.2.5 Laboratory analysis of detailed samples	104
5.3 Data Analysis	105
5.3.1 Calculating caddisfly sediment flux	105
5.3.2 Calculating hydraulic sediment flux	106
5.3.3 Calculating total sediment flux	107
5.4 Results	107
5.4.1 Seasonal discharge and bedload transport in Black Brook	107
5.4.2 What is the mass and GSD of sediment transport in caddisfly cases, and how does this vary	
seasonally?	109
5.4.3 What is the contribution of individual caddisfly taxa to sediment transport?	110
5.4.4 Hydraulic mineral and organic sediment	113
5.4.5 How does the magnitude of caddisfly sediment transport compare to mineral and organic sedi	ment
transported by hydraulic processes?	114
5.5 Discussion	118
5.5.1 What is the mass and GSD of sediment transport in caddisfly cases, and how does this vary	
seasonally?	118
5.5.2 Contribution of caddisfly families	121
5.5.3 How does the magnitude of caddisfly sediment transport compare to hydraulically transported	mineral
and organic sediment?	123
5.5.4 Consequences for fluvial zoogeomorphology	124
5.6 Summary	125

Chapter 6. Discussion	128
6.1 Introduction	128
6.2 Fulfilment of thesis objectives	129
6.2.1 Objective 1 – Quantifying sediment use by case-building caddisfly	129
6.2.2 Objective 2 – Effects of case construction on sediment entrainment	130
6.2.3 Objective 3 – Sediment redistribution by case-building caddisfly	131
6.3 Bioconstruction by case-building caddisfly larvae	132
6.4 Effect of case architecture on caddisfly zoogeomorphology	134
6.4.1 Tube case-builders	135
6.4.2 Dome case-builders	137
6.4.3 Pupal cases of net and retreat building and free-living taxa	1 <i>38</i>
6.5 Scaling up caddisfly zoogeomorphology and future research directions.	141
6.6 Implications for river geomorphology and ecology	146
6.7 Concluding remarks	149

Reference list

150

List of tables

Table 1.1. Mechanisms by which organisms affect geomorphic systems (adapted from Corenblit <i>et al.</i> , 2011; Viles, 2019) with definitions and the typical effect of each mechanism on
sediment stability
Table 2.1. Characteristics of larvae and cases for each case-building caddisfly taxonomic group and individual species of Limnephilidae. 33
Table 2.2. Caddisfly taxa found in this chapter. Taxa are sorted according to their behaviour (after Wiggins, 1996); (1) taxa that used mineral sediments in their cases were included in sediment analysis and (2) cases that were not analysed for sediment content
Table 2.3. Results of Kruskal-Wallis and Dunn's post-hoc tests to determine the variability in sediment use by caddisfly larvae between sites (upper, mid and lower), relative to within site variability.36
Table 2.4. R ² values for linear regression between abundance (independent) and the mass of sediment used (dependent) for each taxon across all samples (whole river) and for each site (upper, mid and lower) independently
Table 3.1. Flume and hydraulic measurements for each flow stage (1-11). Discharge, velocity and bed shear stress increased with each run, whilst depth remained constant
Table 3.2. Size and shape characteristics for cases and sediment. Mean with standard deviation in brackets. a, b, and c indicate major, intermediate and minor particle axes respectively61
Table 3.3. Results of statistical tests performed on (A) Difference between entrainment of each species, for both cases and loose sediment and (B) Difference between case and
sediment
Table 4.1. Details of each treatment. (A) Four treatments were conducted; 1) small gravel, low velocity, 2) large gravel, low velocity, 3) small gravel, high velocity and 4) large gravel, high velocity. For each treatment (B) physiochemical conditions and, (C) details of larvae which stayed within the sediment tray are given
Table 4.2. Characteristics of <i>A. fuscipes</i> cases and sediment use within each treatment. (A) Case characteristics; Mean case mass and D ₅₀ , (B) Sediment displacement; Mean vertical distance moved (Z distance) and sediment flux
Table 5.1. Details of the six sampling occasions. (A) Detailed bedload trap samples were collected on six occasions over the year (date emptied), having been in place for approximately 14 days (19 in August, 13 in December)

List of figures

Figure	1.1. Examples of documented caddisfly zoogeomorphology to date.	11
Figure	1.2. Lifecycle of a case-building caddisfly.	13
Figure	1.3. Diversity of caddisfly larval and pupal architecture (after Wiggins, 1996).	15
Figure	1.4. A conceptual diagram showing possible zoogeomorphic impacts of caddisfly in	
0	gravel-bed rivers. Previous research has considered nets and retreats and burrowing tax	a.
	This thesis will examine the zoogeomorphic effects of case construction.	19
Fioure	1.5. Structure of the thesis outlining the six chapters	21
Figure	2.1. Study site and sampling strategy	26
Figure	2.2 Particle size distribution of the cases of caddisfly taxa	32
Figure	2.3. Total sediment use by the case-building caddisfly community and constituent taxa	33
Figure	2.4. Variability in sediment use between sites (upper, mid and lower)	35
Figure	2.5. Variability in sediment use by <i>A fuscipes</i> (Glossosomatidae) individuals between sites	55 c
i iguite .	2.5. Variability in securitent use by 21. <i>justifies</i> (Giossosofilatidae) individuals between sites	5. 38
Figure	2.6 Sediment use in relation to available sediment in the river had Particle size	50
riguite.	distributions shown as a mean for the river as a whole (with standard deviation) and ear	ch
	site independently	20
Element	site independently	39
rigure .	5.1. Examples of case architecture from the three caucisity species used in this	ΕO
Б,	experiment	50
Figure .	3.2. The laboratory flume setup	52
Figure .	5.5. Entrainment of cases and sediment from measurement platform.	54
Figure .	3.4. Hydraulic conditions during successional flow stages 1-11	56
Figure 1	3.5. Association between width (b axis) and depth (c axis) of cases of each species.	60
Figure .	3.6. Cumulative particle size distribution for case sediment of each of the three species	62
Figure	3.7. Parameters for the entrainment of cases and $90%$ of constituent sediment for each	
	species	64
Figure	3.8. Percent of sediment entrained by the end of each flow stage for cases (solid line) and	d
	sediment (dashed line) of each species (A-C).	65
Figure	3.9. Zingg classification of particle shape for cases of the three species (Mclean 1982).	
	Distinct differences in shape are evident between A. <i>fuscipes</i> cases, which more closely	
	resemble spheres and the rod-shaped cases of P. latipennis and S. personatum	67
Figure	3.10. Shields criteria for cases and sediment for each species.	68
Figure	4.1. Glossosomatidae larvae in the River Dove, Derbyshire, U.K	76
Figure	4.2. The laboratory flume setup.	80
Figure	4.3. Sediment tray setup. A tray of loose sediment was added to the flume, flush with the	e
	bed	81
Figure -	4.4. Flow velocity profiles for a representative run from each treatment. Indicates a	
	difference in velocity between velocity treatments, but not between gravel protrusion	07
г.	treatments.	82
Figure	4.5. Longitudinal movement of <i>A. juscipes</i> larvae within each treatment. (A) Upstream, af	na
г.	(B) downstream.	86
Figure -	4.6. Vertical sediment transport by Glossosomatidae <i>A. fuscipes</i> larvae	86
Figure	4.7. Case architecture of <i>A. fuscipes</i> larvae from each treatment	88
Figure	4.8. The relative hydraulic exposure of <i>A. fuscipes</i> larvae, interred from the position of	
	larvae (Figure 3B).	89
Figure	4.9. The location of each larvae (which remained within the sediment tray) for each run	
	within each treatment.	90
Figure	4.10. Physical bed armouring processes (blue) and Glossosomatidae movement of	
	sediment (green) occurring in an armoured gravel-bed. Movement of sediment, even	
	small distances can substantially alter the exposure of this sediment to entraining flows	

and may affect downstream sediment flux in rivers. Glossosomatidae transport opposes
vertical winnowing and increases the hydraulic exposure of case sediment
Figure 4.11. The stability of an <i>A. fuscipes</i> case (C) is dependent upon anchoring of the case by the larvae, (A) and hydraulic force (B)
Figure 5.1. Sampling location on Black Brook near Loughborough
Figure 5.2 Depth and velocity profiles for sediment trap cross sections looking downstream (A)
Upper traps and (B) Lower traps
Eight 5.2 Discharge and hadlead transport estimations for the study period
Figure 5.5. Discharge and bedioad transport estimations for the study period
(B) Size distribution and (C) Skewness
Figure 5.5. Grain size distribution of sediment use by caddisfly (<i>gci</i>) in each sampling period.111
Figure 5.6. The mineral content of caddisfly sediment for upper (A) and lower traps (B) and split
into each caddisfly family
Figure 5.7. Percentage of caddisfly sediment use in each size fraction contributed by each
caddisfly family (Mean across all samples)
Figure 5.8. (A) The abundance of caddisfly larvae of each family within each sample for upper
and lower traps (B) the contribution of each caddisfly family to total caddisfly sediment
mass in each sample 113
Figure 5.0. Hydraulic and caddiefly sediment content of trans at each detailed sampling occasion
for upper trops (A) and lower trops (D)
$\mathbf{F} = \mathbf{F} + $
Figure 5.10. Contribution of case-building caddisfly (Cl%) and hydraulic (mineral and organic)
sediment to be load sediment flux as a mean across both sets of traps and all sampling
occasions
Figure 5.11.Percentage of total bedload transport that was caddisfly sediment, for each size
fraction in each sample ($ci\%$)
Figure 5.12. Conceptual model of sediment transport processes applying to caddisfly sediment.
Caddisfly and cases may have been transported into the pit traps either by crawling
(vertical and horizontal locomotion; blue arrows) or by entrainment (black arrows) 120
Figure 5.13. Glossosomatidae A. fuscipes cases of different sizes from the same sample
Figure 6.1. Conceptual zoogeomorphology of caddisfly larvae
Figure 6.2. Examples of sediment stabilisation by case building caddisfly observed in Black
Brook (location of the work reported in Chapter 5) September 2018
Figure 6.3 Eurther examples of case construction architecture by tubular case-building caddisfly
larvae
Figure 6.4. Glossosomatidae (believed to be <i>A fuscibes</i>) attach their eggs to small gravel particles
in the shallow marging of Black Brook (Chapter 5) and cover them with a capstone
fastened in place
$E_{\text{result}} = \left\{ \frac{1}{2} \sum_{i=1}^{n} \frac{1}{$
Figure 6.5. Sediment stabilisation by pupal cases of <i>Hyaropsyche sutatat</i> (Hydropsychidae).
Keproduced from Statzner (2012)
Figure 6.6. Hypothesised relationship between zoogeomorphic effects (sediment transport) of
mobile case building taxa (destabilisation) and fixed cases and nets (stabilisation) with
increasing hydraulic power, in gravel-bed rivers
Figure 6.7. The role of spatial scale in this thesis and zoogeomorphology more broadly
Figure 6.8. Caddisfly case longevity. Preliminary results from an ongoing experiment to
understand how quickly Glossosomatidae A. fuscipes cases break down in the river 143
Figure 6.9. Empty caddisfly cases from Little Stour River, Kent UK provide surfaces for calcium
carbonate deposition144
Figure 6.10. Examples of seeds in caddisfly cases observed in samples collected for Chapter 5.

Chapter 1. Introduction

1.1 Research context

Throughout history, rivers have been integral to human civilisation (Gani and Gani, 2011; Haidvogl, 2018). Rivers provide critical ecosystem services, including water supply, provision of food, power generation, navigation pathways, cultural and aesthetic value, and enhance human well-being (Böck *et al.*, 2018; Kaval, 2019). Estimates suggest that freshwater ecosystems deliver \$6.5 trillion USD a⁻¹ worth of services, which is 20% of the global total, despite covering only 0.8% of the Earth's surface (Constanza *et al.*, 1997; Strayer and Dudgeon, 2010). Freshwater habitats are also extraordinarily biodiverse, supporting 9.5% of total described animal species (Balian *et al.*, 2008).

However, multiple pressures on rivers, including morphological and hydrological modification, urbanisation, pollution and climate change, have caused a global decline in freshwater biodiversity (Dudgeon *et al.*, 2006; Malmqvist and Rundle, 2002; Reid *et al.*, 2019; Strayer and Dudgeon, 2010). Freshwater flora and fauna have incurred an 81% reduction in abundance, compared to a global average of 58% across all habitats (1970-2012; WWF, 2016). This is concerning not only because of the intrinsic value of species, but also the loss of the ecosystem services that they provide (Cardinale *et al.*, 2012). Consequently, research into the geomorphological and ecological functioning of rivers is required to mitigate and reverse this decline (Auerbach *et al.*, 2014; Palmer and Ruhi, 2019; Wohl *et al.*, 2015).

Sediment (fragmentary material transported by or deposited in water; Edwards and Glysson, 1999), is an important control on river geomorphology and ecology. Sediment transport is a key driver of river morphology (Schumm, 1977), water quality (Hauer *et al.*, 2018), and provides the habitat template for a diverse array of plants and animals (Newson and Newson, 2000; Southwood, 1977). The erosion, deposition and transport of fluvial sediment is, therefore, a primary control on the ecosystem services provided by rivers and understanding sediment transport is essential for river management and restoration (Kondolf *et al.*, 2006; Sear, 1994). The influence of fluvial sediment on biological communities has been fundamental to research in aquatic ecology (Frissell *et al.*, 1986; Vannote *et al.*, 1980). More recently, however, research has demonstrated the reciprocal effects of biology on sediment dynamics in rivers (Atkinson *et al.*, 2018; Reinhardt *et al.*, 2010). Plants and animals have important impacts which can either reduce or increase sediment transport. For example, riparian vegetation stabilises river banks (Brooks

and Brierley, 2002; Gurnell *et al.*, 2019) while bioturbating invertebrates mix and erode sediment from the riverbed (Nogaro *et al.*, 2006).

Zoogeomorphology is the field of research that specifically considers the role of animals in geomorphology (Butler, 1995, 1992). It is increasingly recognised that animals can have substantial effects on sediment transport in rivers (Rice *et al.*, 2012; Statzner, 2012). Zoogeomorphic processes are important because animals may modify river sediment yields, morphology, and habitat. Signal crayfish, for example, increase suspension of fine sediment (Harvey *et al.*, 2014), which may reduce habitat quality for invertebrates (Mathers *et al.*, 2020). In addition, zoogeomorphic engineers may present a risk to lives or property. Burrowing of porcupines may have contributed to the collapse of a levee containing the Secchia river, Italy (Orlandini *et al.*, 2015; Taccari and Van Der Meij, 2016). In contrast, zoogeomorphic effects can potentially be used to enhance river ecosystem services (e.g. beaver rewilding to restore biodiversity; Law *et al.*, 2017), or to prioritise the conservation of species which usefully affect sediment dynamics or increase habitat diversity. Consequently, there are growing calls to understand the interactions between geomorphology and ecology to better manage and restore rivers (Auerbach *et al.*, 2014; Johnson *et al.*, 2019; Palmer and Ruhi, 2019).

Zoogeomorphic research in rivers has typically focussed on large charismatic taxa, especially beaver and salmonid fish (Atkinson *et al.*, 2018; Rice *et al.*, 2012; Statzner, 2012). Despite the recognised importance of invertebrates to geomorphic processes in terrestrial and marine systems (Cadée, 2001; Wilkinson *et al.*, 2009), they have received far less attention in rivers (Albertson and Allen, 2015). Caddisfly (Trichoptera: Insecta) have the potential to be important zoogeomorphic engineers. Caddisfly are diverse and ubiquitous in lotic habitats and can be extremely abundant, often occurring in very high densities (> 1000 larvae m⁻²; Mcneely and Power, 2007; Wiggins, 2004). Furthermore, caddisfly may have disproportionately large zoogeomorphic effects due to their construction of filter-feeding silk nets and cases from fine sediment (Albertson and Allen, 2015). Whilst caddisfly nets have been found to bind together gravel particles and increase the critical shear stress required to initiate gravel movement by 10– 40% (Statzner *et al.*, 1999; Cardinale *et al.*, 2004), the zoogeomorphic effects of case-construction by caddisfly larvae are unknown.

Most caddisfly species use fine sediment (~0.1 mm up to 5 mm; De Gispert, 2015; Hansell, 1968; Okano & Kikuchi, 2012; Statzner *et al.*, 2005; Tolkamp, 1980) in the construction of mobile cases, fixed retreats or pupal cases at some point in their life cycle (Wiggins, 1996). Both the construction of cases and the movement of sediment by caddisfly may modify the mobility

of incorporated sediment and its exposure to hydraulic forces. As a result, case-building caddisfly may affect sediment transport processes in rivers. This research is the first to consider the zoogeomorphic importance of case construction by caddisfly larvae. The aim of this thesis is to investigate and quantify the effect of case-building caddisfly on the distribution and transport of sand in rivers.

1.2 Fluvial zoogeomorphology

Recognition of the importance of organisms in shaping Earth surface processes is not new (Darwin, 1881) but it has seen a surge of research in the last 30 years, within the discipline of *biogeomorphology* (Viles, 1988). Biogeomorphology considers the two-way interaction between the biosphere and the geosphere to understand how plants and animals shape landforms (Naylor *et al.*, 2002; Viles, 2019). Biogeomorphology and the sub-discipline of zoogeomorphology have largely been studied by geomorphologists. Ecological interest in the closely related discipline of *ecosystem engineering* (Hastings *et al.*, 2007; Jones *et al.*, 1994) has also been growing (Butler and Sawyer, 2012). Ecosystem engineering investigates the physical modification, maintenance or creation of habitat by organisms (Jones *et al.*, 1994) and, as such, there is considerable overlap between the disciplines. Ecosystem engineers usually also have biogeomorphologists and ecologists, which is required to address common interdisciplinary research questions (Reinhardt *et al.*, 2010; Rice *et al.*, 2010; Stallins, 2006).

The biogeomorphic effects of organisms can be broadly divided into five categories (Table 1.1) (Corenblit *et al.*, 2011; Naylor *et al.*, 2002; Viles, 2019). *Bioprotection* reduces erosion or weathering processes, stabilising sediment. *Bioconstruction* typically also has a stabilising effect (Viles, 2019), but may also destabilise sediment (Fei *et al.*, 2014). In contrast, *bioweathering, bioturbation* and *bioerosion* usually act to destabilise material. The diversity of organisms and their range of functional traits and behaviours can make conceptualising their zoogeomorphic effects difficult (Allen *et al.*, 2014). Individual species may also have both stabilising and destabilising effects depending on their biotic and abiotic context (Moore, 2006).

The dynamic nature and abiotic energy of many rivers may suggest that biogeomorphic effects in rivers are less important than in other ecosystems (Statzner, 2012). Biology, however, is proving to have large scale geomorphic effects that can dominate abiotic energy in some systems (Atkinson *et al.*, 2018; Coombes, 2016; Harvey and Bertoldi, 2015). Castro and Thorne (2019)

3

conceptualised biology as a primary driver of river typology (alongside geology and hydrology), highlighting the importance of plants and animals as major controls on river morphology and habitat.

The importance of vegetation and large woody material (large wood) to river geomorphology is evidenced by its now widespread use in river restoration (Cashman *et al.*, 2019; Grabowski *et al.*, 2019; Gurnell *et al.*, 2019). Vegetation and large wood can have large-scale effects on channel planform and longitudinal profiles (for reviews see; Corenblit *et al.*, 2007; Gurnell, 2014; Gurnell *et al.*, 2019; 2002; Montgomery and Piégay, 2003; Wohl, 2013). Usually, vegetation in rivers is associated with sediment stabilisation. Aquatic macrophytes, large-wood structures and riparian vegetation reduce local flow velocities and trap sediment, providing bioprotection (Table 1.1) to river beds and banks (Cotton *et al.*, 2006; Tal and Paola, 2010). When the spatial distribution of vegetation is patchy, however, flow is deflected around these biostabilised areas, leading to increased scour (Vandenbruwaene *et al.*, 2011). Vegetation can therefore increase the spatial complexity of geomorphic processes and the diversity of habitat provided.

Mechanism	Definition	Typical geomorphic effect
(1) Bioconstruction	Allogenic (using material external to the organism) or autogenic (utilising self-generated tissue) structures built by organisms (Jones et al., 1994; Naylor et al., 2002).	Sediment
(2) Bioprotection	Biological effects which lead to the accumulation or stability of sediment (Carter and Viles, 2005).	stabilisation
(3) Bioweathering	Chemical weathering associated with biota (Dorn, 2014).	
(4) Bioturbation	Biotic reworking of lose sediment (Meysman et al., 2006).	Sediment destabilisation
(5) Bioerosion	Degradation of hard surfaces by organisms (Davidson et al., 2018).	

Table 1.1. Mechanisms by which organisms affect geomorphic systems (adapted from Corenblit *et al.,* 2011; Viles, 2019) with definitions and the typical effect of each mechanism on sediment stability.

An increasing number of animals are documented to affect hydraulics, bed sediment sorting and sediment transport processes in rivers (see reviews; Rice *et al.*, 2012; Statzner, 2012; Albertson and Allen, 2015). Beaver (*Castor canadensis* and *Castor fiber*) have received the most research attention due to their construction of dams (bioconstruction; Table 1.1), which act as traps for water and sediment (Butler and Malanson, 2005; Gurnell, 1998; Naimen *et al.*, 1988; Puttock *et al.*, 2018; Ruedemann and Schoonmaker, 1938). Beaver dams can create multiple channels and a diverse mosaic of habitats, fundamentally changing the geomorphology of smaller rivers and riparian zones (Gurnell, 1998). Beaver are being reintroduced within their historical geographical range to restore habitat diversity, biodiversity and slow the downstream transport of sediment and water (Law *et al.*, 2017; Pollock *et al.*, 2015, 2014). Due to their dam construction, Beaver are usually sediment stabilisers, but burrowing by beaver can also erode river banks and input fine sediment into river systems (bioturbation, Table 1.1; Meentemeyer *et al.*, 1998).

Spawning Salmonidae (e.g. salmon and trout) also have strong zoogeomorphic effects (DeVries, 2012; Hassan *et al.*, 2011). Female salmon construct redds (nests) within gravel-beds during spawning, using strong undulations of their tail (Burner, 1951). Due to the vertical mixing and disturbance of stabilising gravel-bed sediment structures, redd construction locally destabilises bed sediment (Buxton *et al.*, 2015; Rennie and Millar, 2000). The landscape of pits and tail spills created by spawning salmonids is also vulnerable to erosion in high discharge (Gottesfeld *et al.*, 2004). A study in British Columbia estimated that bioturbation by salmon accounted for 30-50% of total annual bedload flux at spawning locations on four gravel-bed rivers (Hassan *et al.*, 2008). Salmonidae spawning may therefore influence the characteristics of river channels and sediment transport over large spatial and temporal timescales (Fremier *et al.*, 2017).

Benthivorous fish also bioturbate and destabilise sediment while foraging (reviewed by Rice *et al.*, 2019). Foraging by the Sonora sucker (*Catostomus insignis*), for example, was responsible for diurnal cycles of up to 32% of daily suspended matter load (Booth *et al.*, 2020). Pledger *et al.*, (2014) determined that foraging Barbel (*Barbus barbus*) could increase gravel entrainment by 82% in a laboratory setting. Density (Barton *et al.*, 2000; Breukelaar *et al.*, 1994; Parkos III *et al.*, 2003), body size (Pledger *et al.*, 2016) and specific feeding behaviours (Rice *et al.*, 2019), are all important controls on the geomorphic effectiveness of benthivorous fish bioturbation. Consequently, both salmonids and benthivorous fish disturb water worked structures, increasing the likelihood of sediment erosion in subsequent high flows (Pledger *et al.*, 2014; Statzner *et al.*, 2003). Nest building (bioconstruction; Table 1.1) by non-salmonid species from river-bed sediment may also have geomorphic implications (Rice *et al.*, 2019).

1.3 Invertebrate zoogeomorphology

"Worms have played a more important part in the history of the world than most persons would at first suppose" (Charles Darwin, 1881 p. 305).

Moore (2006) hypothesised that the magnitude of an organism's zoogeomorphic effect is a function of its density, behaviour and body size. Using a meta-analysis of biogeomorphic studies in lotic environments, Albertson and Allen (2015) tested this hypothesis. While the biomass of individuals was associated with the scale of their potential zoogeomorphic work, when populations were considered, the effects of invertebrates were comparable to much larger vertebrates (Albertson and Allen, 2015). After controlling for biomass, the individual effects of some aquatic insects were 550 times greater than those of fish and crustaceans (Albertson and Allen, 2015). Therefore, whilst zoogeomorphic research in rivers has prioritised larger taxa with evident geomorphic impacts (Statzner, 2012), the less noticeable effects of invertebrates can be equally important. Rice *et al.*, (2012) called these 'Cinderella' taxa because their important geomorphic effects are not easily visible in rivers.

In terrestrial and marine environments, the zoogeomorphic importance of invertebrates has received more research than in lotic systems. Functionally, there are many similarities between invertebrates in these habitats with those in rivers (Meysman *et al.*, 2006), therefore a brief description and examples of terrestrial and marine invertebrate sediment engineers is included, as well as a more exhaustive review of literature on the zoogeomorphology of freshwater invertebrates. Literature from other disciplines, including ecosystem engineering, ichnotaxonomy (fossil animals) and geochemistry, is also re-interpreted from a zoogeomorphic context (following Statzner, 2012).

1.3.1 Terrestrial

In his seminal text on zoogeomorphology, Butler (1995 pp. 11) stated "*the geomorphic effects of invertebrates are apparent to even the most casual observer who has seen an ant mound or earthworm casting*". Perhaps because of their increased visibility, the zoogeomorphic effects of terrestrial invertebrates have been recognised for a considerable time. Darwins' (1881) book concerning the importance of earthworms in soil is often cited as the first reference to animal geomorphic agents (Coombes, 2016; Meysman et al., 2006; Rice et al., 2019). Darwin recognised, not only that worms bioturbate sediment, moving 0.53 t ha⁻¹ a⁻¹ upwards to the soil surface, but that this contributed to landform development (Feller *et al., 2003*). Subsequently, considerable research

has considered the bioturbation effects of worms and their role in soil formation (*pedogenesis*; see reviews by Meysman *et al.*, 2006; Wilkinson *et al.*, 2009). Ants and termites also contribute to bioturbation (Dangerfield *et al.*, 1998; Whitford and Eldridge, 2013). Sediment excavation by ants is often 1–5 t ha⁻¹ a⁻¹ soil and occasionally exceeds 50 t ha⁻¹ a⁻¹ (Seal and Tschinkel, 2006; Wilkinson *et al.*, 2009).

Worms, ants, and termites construct sediment structures (mounds; Wilkinson *et al.*, 2009). Mounds of loose sediment may be easily eroded (such as Darwin's earthworm casts; Darwin, 1881) but some mounds, especially those of ants and termites, are maintained (Humphreys and Mitchell, 1983) and can be stabilising components of terrestrial landscapes (Wilkinson *et al.*, 2009). Ant mounds can persist as long as 100 years (Cowan *et al.*, 1985). Terrestrial invertebrates, therefore, bioturbate sediment below the ground surface, create mound structures above the surface and influence both soil formation and the flux and downslope erosion of fine sediment (Gabet *et al.*, 2003; Richards and Humphreys, 2010; Wilkinson *et al.*, 2009).

1.3.2 Marine

The significance of invertebrate zoogeomorphology is also recognised in marine environments (see reviews by Mermillod-Blondin and Rosenberg, 2006; Murray *et al.*, 2002; Shull, 2009). As for the terrasphere, bioturbation is an important process in marine environments (Mermillod-Blondin, 2011; Murray *et al.*, 2002). Bioturbators convey sediment upwards, downwards or internally within the substrate (François *et al.*, 1997; Shull, 2009). For example, burrowing crabs moved 0.375 kg m⁻² of sediment per day to the surface of unvegetated mudflats (Wang *et al.*, 2010).

Filter feeding Polychaete worms build tubes from secreted material and often occur at high densities, creating tube 'lawns' and modifying the physical structure of soft sediment environments (Dudgeon, 1990). The effect of Polychaete tube lawns on sediment stability is still debated, however, with several studies documenting a destabilising effect caused by flow acceleration around the tubes (Eckman *et al.*, 1981; Luckenbach, 1986; Xie *et al.*, 2018) and others a stabilising effect (Meadows *et al.*, 1990; Meadows and Tait, 1989). Friedrichs *et al.*, (2000) suggest that polychaete tubes transition from having destabilising to stabilising effects based on population density, at 8.8 % of sediment surface coverage by their tubes, regularly recorded in the field. Similar zoogeomorphic activities are reported for amphipods, such as *Haploops nirae* which constructs tubes and stabilises soft bed sediment, facilitating colonisation by other invertebrates (Chaalali *et al.*, 2017).

1.3.3 Sediment destabilisation in freshwater habitats: bioturbation and bioerosion

Bioturbators redistribute sediment, changing both the physical structure of substrate and the chemical and biological processes occurring within it (Gautreau *et al.*, 2020; Mermillod-Blondin, 2011). The effects of bioturbators vary based on the functional characteristics of the organisms (Mermillod-Blondin *et al.*, 2002; Michaud *et al.*, 2005; Nogaro *et al.*, 2009) and the abiotic environment. Mermillod-Blondin and Rosenberg (2006) distinguish between diffusion dominated habitats: fine grained sediments with low hydraulic conductivity (typical of lakes and marine systems); and advection dominated substrates with high hydraulic connectivity and usually larger grain sizes (rivers). Bioturbation in diffusion dominated habitats has particularly strong effects on microbial and geochemical processes (Mermillod-Blondin, 2011). In advection dominated substrates, such as the hyporheic zone of most rivers, bioturbators can affect hydrological exchange with the surface (Nogaro *et al.*, 2009) and may be especially important geomorphologically by altering the exposure of sediment to erosive hydraulics at the sediment-water interface. Bioturbation in rivers is particularly strong when stabilising grain structures (such as armoured gravel-beds) are disturbed (Statzner, 2012).

Several species of crayfish cause substantial bioturbation via locomotion (foraging and fighting) including the rusty crayfish (*Faxonius rusticus*; Albertson and Daniels, 2016; Statzner *et al.*, 2000; Statzner and Peltret, 2006), red swamp crayfish (*Procambarus clarkii*; Arce and Diéguez-Uribeondo, 2015) and the signal crayfish (*Pacifastacus leniusculus*; Harvey *et al.*, 2014; Johnson *et al.*, 2010; Rice *et al.*, 2014). Crayfish locomotion in gravel substrates results in pits and mounds (Albertson and Daniels, 2016; Johnson *et al.*, 2010), affects near bed hydraulics (Han *et al.*, 2019) and leads to the suspension of fine sediment (Harvey *et al.*, 2014). Diurnal signals of high turbidity at night, when crayfish are more active, allowed Rice *et al.* (2016) to isolate the contribution of nocturnal signal crayfish to baseflow sediment yield for a low-order, lowland, UK stream. Signal crayfish added at least 32% (474 kg) to monthly baseflow sediment (Rice *et al.*, 2016). Bioturbation by signal crayfish can also can increase the infiltration (downwards transport) of fine sediment into interstices between gravel particles during foraging (Mathers *et al.*, 2019).

In the UK, where they are invasive, signal crayfish also contribute to bioturbation via burrowing into river banks. Faller *et al.* (2016) estimated that signal crayfish add approximately 3 t km⁻¹ of sediment per burrowed bank in the Thames catchment, UK, but burrowing was patchy and occurred at only 10% of banks surveyed. Crayfish burrowing may also destabilise banks, causing

collapse and further fine sediment input into river channels (Arce and Diéguez-Uribeondo, 2015; Harvey *et al.*, 2019).

Other crustaceans also have bioturbation effects including Atyidae (shrimp) which reduce sedimentation (Pringle *et al.*, 1993; Pringle and Hamazaki, 1998) and Asellidae (water hoglouse) which mix and homogenise sediment within the river bed (Mermillod-Blondin *et al.*, 2002). As in terrestrial habitats, Oligochaeta (worms) convey sediment upwards, feeding on material up to 20 cm below the sediment surface, and transporting this to the surface as faecal pellets (Ciutat *et al.*, 2006; Dafoe *et al.*, 2011). Networks of burrows constructed by Tubificidae (Oligochaeta) increase interstitial flow within fine-grained benthic sediments (Nogaro *et al.*, 2006), modify lake sediment structure, and increase suspended sediment (Zhang *et al.*, 2014; Ciutat *et al.*, 2006).

Predatory stonefly larvae can transport substantial quantities of sand from gravel interstices whilst foraging for prey (up to 200 – 400 kg m⁻² yr⁻¹ for *Dinocras cephalotes;* Statzner *et al.*, 1996), with reduced prey availability increasing foraging activity, and therefore erosion (Statzner *et al.*, 1996; Zanetell and Peckarsky, 1996). Similarly, *Pseudiron centralis* (mayfly) larvae have been documented eroding sand furrows while hunting for buried Chironomidae (non-biting midge) prey (Soluk and Craig, 1990). Mayfly in lakes are also documented to have bioturbation effects (Charbonneau and Hare, 1998; Edmunds and Mccafferty, 1996), by burrowing or dwelling in interstices (Bae and McCafferty, 1995). Individual *Hexagenia limbate* (mayfly) larvae can move an average of 4 cm³ sediment per day (Charbonneau and Hare, 1998).

Consequently, bioturbation is an important zoogeomorphological process in both lotic and lentic environments. In a pool of a small, low-order stream, bioturbation by the resident invertebrate communities resulted in 85% of the surface reworked after 4 days (De Nadaï-Monoury *et al.,* 2013). A number of taxa were responsible, but these included cranefly (Tipulidae), dragonfly (Odonata) and two caddisfly species; *Odontocerum albicorne* (Odontoceridae) and *Sericostoma sp.* (Sericostomatidae; De Nadaï-Monoury *et al.,* 2013).

Bioerosion (the removal of material from hard surfaces; Table 1.1; Davidson *et al.*, 2018), is documented for only a small number of taxa in freshwater habitats, largely within the fields of paleoecology and ichnotaxonomy. Uchman *et al.* (2017) documented recent U-shaped burrows of mayfly in consolidated sediment from the rivers Drwęca (Poland) and Ohře (Czech Republic). Similarly, Savrda (2019) found Chironomidae burrows, up to 40 mm deep in claystone in riffle habitats of the Conecuh River, USA. Caddisfly burrows were also documented, in both sandstone and claystone in riffles, with burrows also up to 40 mm deep and extended above the burrow by chimneys of fine-sediment (Figure 1.1; Savrda, 2019).

1.3.4 Sediment stabilisation in freshwater habitats: bioconstruction and bioprotection

There are fewer examples of stabilisation than destabilisation in lotic zoogeomorphology (Albertson and Allen, 2015), but thanks to their diverse behaviours, a number of aquatic invertebrates have stabilising effects. Silk is a protein secreted by several aquatic insects (Trichoptera, Diptera and Lepidoptera) and some arthropods (Wotton, 2011). Silk is used by organisms in tube construction, feeding nets, and to attach to substrate (Wotton, 2011). Silk can also affect the stability of benthic substrates, providing bioprotection. For example, the silk tubes constructed by Chironomidae larvae act to reduce suspension of fine sediment (Edwards, 1962).

Similarly, the construction of nets and retreats from silk by caddisfly larvae has been found to increase the critical shear stress required to initiate gravel movement by 10 to 40% (Albertson *et al.*, 2014a; Cardinale *et al.*, 2004; Johnson *et al.*, 2009; Statzner *et al.*, 1999; Figure 1.1). Several biotic factors are important controls on these zoogeomorphic effects. Larger taxa, with stronger silk result in greater sediment stabilisation. For example, *Arctopsyche californica* result in greater stabilisation than *Ceratopsyche oslari* (Albertson *et al.*, 2014b). *Stenopsyche marmorata* are even larger and can secure larger gravel particles (Nunokawa *et al.*, 2008; Takao *et al.*, 2006). Species interactions are also important. Further increases in sediment stability occur with *A. californica* and *C. oslari* polycultures, possibly because they inhabit different hydraulic niches (Albertson *et al.*, 2014b; 2019).

Caddisfly nets therefore provide bioprotection by binding gravel particles together, increasing their effective size (Albertson *et al.*, 2014a), and modelling suggests nets may reduce near bed flow velocities by nearly 60% (Juras *et al.*, 2018). *Stenopsyche marmorata* retreats fixed with silk between large cobbles can increase the mean force required to move a cobble 2 - 3 fold for a single retreat (Nunokawa *et al.*, 2008). Furthermore, many caddisfly species fix gravel particles to cobbles in pupation cases (Statzner et al., 2005), which may stabilise sediment (Statzner, 2012). Attempts to upscale the zoogeomorphic effects of net-construction by caddisfly have included modelling (Albertson *et al.*, 2014b; Tashiro *et al.*, 2005) and large scale experiments. Albertson *et al.*, (2019) used 50 m² semi-natural outdoor river channels and found that *A. californica* and *C. oslari* increased gravel entrainment thresholds by 20%.

Freshwater mussels also stabilise sediment via the physical weight of the mussels reducing entrainment (Statzner, 2012), and the use of byssal threads by some species to attach themselves to substrate (Peyer *et al.*, 2009). Zimmerman and de Szalay, (2007) found that mucket (*Actinonaias ligamentina*) and kidney shell mussels (*Ptychobranchus fasciolaris*) stabilised sediment when they were

dormant but also had bioturbation effects whilst burrowing and reproducing. Mussel aggregations can be extensive in freshwater habitats (e.g. several thousand invasive zebra mussels (*Dreissena polymorpha*) per m⁻² in the Hudson river, USA; Strayer, 1999) and can provide stable habitat for other invertebrates (Ricciardi *et al.*, 1997; Vaughn and Spooner, 2006). Furthermore, zebra mussels may reduce suspended sediment levels by filter feeding (Skubinna *et al.*, 1995; Strayer, 1999).

Invertebrate biostructures, including cases of caddisfly, can provide a stable spot for calcium carbonate to deposit (Roche *et al.*, 2019), resulting in bioherms (accretionary mounds consisting of fossilised remains surrounded by different geology; Cummings, 1932; Figure 1.1). In Western China, He *et al.* (2015) recorded caddisfly bioherms 3-5 m high stretching 400-500 m laterally. Similarly, bioherms predominantly composed of layers of caddisfly cases and calcium deposits line the shores of Lake Gosiute, Wyoming, USA (Leggitt and Cushman, 2001; Figure 1.1). These bioherms are extensive; 70 km in length, 9 m in height and 40 m in diameter (Leggitt and Cushman, 2001).



Figure 1.1. Examples of documented caddisfly zoogeomorphology to date. (A) Sediment stabilisation by caddisfly filter-feeding nets. (B) Caddisfly retreats can stabilise both coarse and fine gravel. (C) Bioerosion by caddisfly burrows which extend 40 mm into mudstone (white outline) accompanied by smaller Chironomidae burrows (white arrows) (Scale bar = 1 cm; Savrda, 2019). (D) Layers of caddisfly form calcium carbonate bioherms. Each caddisfly case is orientated vertically with the anterior end downwards (Leggitt & Cushman, 2001). (A) and (B) photos M. Johnson.

Therefore, many lotic invertebrates have been demonstrated to have zoogeomorphic effects, acting to both stabilise and destabilise sediment. A significant challenge for biogeomorphology is to understand the net effects of a vast range of organisms that vary widely in behaviour and may have complicated and often contradictory zoogeomorphic effects under different biotic and abiotic conditions. Zoogeomorphic research is usually undertaken at limited spatial and temporal scales (short term field or flume experiments) and few studies attempt to understand the effects of organisms at scales typically studied by geomorphologists (e.g. Rice *et al.*, 2016). There is also a need to study the zoogeomorphic effects of more taxa, particularly aquatic invertebrates which have received less research in rivers (Albertson and Allen, 2015; Rice *et al.*, 2012). Taxa which create bioconstructions or provide bioprotection using autogenic secretions are particularly effective sediment stabilisers in aquatic ecosystems, including Polychaete worms, mussel byssal threads, Chironomidae tubes and net-building caddisfly. In rivers, net building caddisfly have received a great deal of research for their stabilisation of gravel-beds with filter feeding nets. However, the zoogeomorphic effects of case construction by caddisfly larvae have not been considered.

1.4 Caddisfly biology and case-building behaviour

1.4.1 Caddisfly diversity and life history

Caddisfly (Trichoptera) are one of the most diverse orders of aquatic insects (Holzenthal *et al.*, 2007; Malm *et al.*, 2013). There are currently 16,267 extant caddisfly species recorded (Morse, 2020), of which, 198 are found in the UK (Wallace *et al.*, 2016). Caddisfly are found in all biogeographical regions except for the Antarctic (de Moor and Ivanov, 2008). They are abundant in most freshwater systems including rivers, lakes and ephemeral systems often occurring in densities of several thousand per meter square (Mcneely and Power, 2007; Wiggins, 2004; de Moor and Ivanov, 2008). Caddisfly larvae vary in size but are usually < 30 mm long (Wallace *et al.*, 2003). Caddisfly larvae are typically aquatic (with one exception in the UK; Wallace, 2016), and the adults are terrestrial (Figure 1.2).

The lifecycle of a caddisfly usually takes approximately one year in temperate latitudes (univoltine), although this varies with taxa and environment (Becker, 2005; Cudney and Wallace, 1980). Most of a caddisfly's lifecycle is spent as a larvae, during which time caddisfly go through a number of instars (larvae development stages; usually 5; Wiggins, 2004; Figure 1.2). For most species at temperate latitudes, metamorphosis from pupae to adult takes about three weeks

(Wallace *et al.*, 2003). The new adult uses its pupal mandibles (powered by the muscles of the adult) to cut its way out of the cocoon and case and swim (using pupal swimming legs) to the surface where it climbs onto emergent substrate or vegetation (Figure 1.2). Adult caddisfly do not have mandibles so either require liquid food (e.g. nectar or honeydew from aphids) or use energy stored from larval feeding (Wiggins, 2004). Typically female caddisfly dive into the water to lay eggs, or fly down to the water surface and release eggs (Figure 1.2).



Figure 1.2. Lifecycle of a case-building caddisfly. Larvae are typically aquatic whilst the adult is terrestrial. Timings, number of larval instars and number of generations per year vary between species.

Caddisfly larvae are diverse in their functional behaviour. For example, caddisfly larvae vary in their feeding strategies (Cummins, 1995; Wiggins and Mackay, 1978). Scrapers and grazers use their specially adapted mandibles to feed on microscopic organisms (e.g. diatoms, other algae and protozoans; Wiggins, 2004). Shredders utilise vascular plant material and aid in leaf litter decomposition, a major source of energy in streams (Vannote *et al.*, 1980). Filter feeders construct nets to catch food from the water column. The abundance of filter feeders can be particularly high because they do not rely on resources from their immediate environment (Wallace *et al.*, 1977). A number of caddisfly are also predators of other invertebrates and occasionally larger organisms (e.g. eggs and tadpoles of wood frogs; Rowe *et al.*, 1994). Due to their feeding behaviour, caddisfly larvae perform essential roles in stream ecosystems, including processing of organic matter by shredders (Graça, 2001; Wallace and Webster, 1996), secondary

productivity as food for higher organisms (including fish), and affecting the structure of algal communities (Hart, 1985; Kohler, 1992; McNeely and Power, 2007). In addition caddisfly provide a number of ecosystem services (See Morse *et al.*, 2019). However, it is for their bioconstruction that caddisfly are of most interest to zoogeomorphologists.

1.4.2 Caddisfly architecture and bioconstruction

Caddisfly (Trichoptera) have evolved diverse strategies to survive and thrive in lotic habitats and the production of silk is believed to be central to their evolutionary success (Wiggins, 2004). Silk is a protein synthesized by the larvae and produced from silk glands at the tip of the labium (a mouth part; Holzenthal *et al.*, 2007). Caddisfly silk has high specific strength and ability to stretch and can more than double in length before breaking (Brown *et al.*, 2004; Tszydel *et al.*, 2009). Therefore, caddisfly structures can be strong and long lasting (Albertson and Daniels, 2016; Maguire *et al.*, 2020).

Silk allows caddisfly larvae to create a wide variety of structures. Wiggins (1996) classified larvae into five groups (Figure 1.3, i - v) which: i) construct retreats and silk filtration nets; ii) bind together fine sediments into portable tubular cases; iii) build domed (saddle shaped) cases; iv) construct purse-type cases, primarily composed of silk; and v) are free living, building a case only for pupation. These architecture types broadly fit with the three typically recognised taxonomic caddisfly suborders; Annulipalpia (retreat builders), Integripalpia (tube case-builders) and Spicipalpia (free living, dome and purse cased caddis; de Moor and Ivanov, 2008; Holzenthal *et al.*, 2007; Wiggins, 2004). Most caddisfly species (Figure 1.3, all groups i-v including net builders and free-living taxa) construct or modify cases for pupation (Wiggins, 1996). Therefore, most caddisfly taxa build a case at some point in their lifecycle.



Figure 1.3. Diversity of caddisfly larval and pupal architecture (after Wiggins, 1996). (i) Nets filter food from flowing water, often accompanied by a static retreat in which the larvae resides (e.g. Hydropsychidae). (ii) Tubular cases which are usually mobile but may be fixed to stable particles for pupation (e.g. Limnephilidae). (iii) Domed/saddle cases are also usually mobile but fixed for pupation (e.g. Glossosomatidae). (iv) Purse cases (e.g. Hydroptilidae). (v) Pupal cases of free living or retreat building taxa (e.g. Rhyacophilidae or Hydropsychidae). The scale bar indicates approximately 5 mm.

Case-building caddisfly use mineral and/or organic sediment in the construction of a wide range of case designs (e.g. Figure 1.3). The design of cases is largely dependent on the species and instar of caddisfly (Wallace et al., 2003). The size of the overall case, and the individual sediment particles incorporated, typically becomes larger as the larvae grows (Statzner, 2011). Case design may also change with larvae instar, with some taxa only building cases in later instars or changing from organic to mineral sediment as they develop. For example, Hansell (1972) described the tubular case building (Figure 1.3ii) of Lepidostoma hirtum (Lepidostomatidae). L. hirtum builds a case from fine mineral grains shortly after hatching and periodically extends its case by adding material to the anterior (front end) and cutting material off the posterior (back end), during each instar. At the third instar, L. hirtum larvae added rectangular leaf panels, thus transitioning from a mineral to organic case (Hansell, 1972). Whilst most tube case-builders construct one case and modify it during their larval life stage (as for L. hirtum), dome/ saddle case-builders such as Glossosomatidae (Figure 1.3iii) build a new case at the start of each instar (Becker, 2005; Houghton and Stewart, 1998). Some caddisfly species are highly selective of the sediment that they incorporate into their cases and will sort through and discard many particles (Hansell 1968), assessing their size, shape, density and surface texture (Okano & Kikuchi 2012). Caddisfly larvae

can adapt their case design based on sediment availability (Okano *et al.*, 2010; Statzner and Dolédec, 2011), including the use of artificial particles such as plastic (Ehlers *et al.*, 2019). Some evidence suggests that larvae may also change their case design based on biotic context, such as the presence of predators (Boyero *et al.*, 2006), or abiotic context, such as differences in flow velocity (Delgado and Carbonell, 1997).

During pupation (Figure 1.2), tubular case-builders (Figure 1.3ii) modify their cases slightly, fixing them to stable substrate and adding silk to the anterior and posterior ends to reduce risk of predation but maintain water flow for respiration (Wiggins, 2004). Net and retreat taxa construct a chamber of mineral or plant material with a silk lining. This may be adapted from the retreat although most species abandon their existing retreat and build a new cocoon (Figure 1.3v; Holzenthal *et al.*, 2007; Wiggins, 2004). Dome case-builders often construct a new case, while free-living and net-building taxa build cases only for pupation. These pupal cases are attached to the river bed with silk (Holzenthal *et al.*, 2007).

Building cases is extremely costly for caddisfly larvae, requiring substantial energy investment to collect and orientate the constituent particles and for silk production. Otto (1975) determined that 12% of the total energy expended in the final instar of *Potamophylax cingulatus* (Limnephilidae; tubular case-builder) was used in silk production for case building and this comes at a cost of larvae growth (Stevens *et al.*, 1999). Cases must provide a benefit to caddisfly larvae to justify this investment. Cases act as camouflage, deterrent and provide protection if larvae are attacked (Johansson, 1991). For example Ferry *et al.* (2013) found that the presence of a case aided in caddisfly survival against larval dragonfly. Cases of many tube case-builders also aid in respiration because larvae perform undulations to pump water through the case (Feldmeth, 1970; Williams *et al.*, 1987). However, dome case-builders (e.g. Glossosomatidae) do not undulate within their case, instead building cases with many gaps and preferring fast flowing habitats to ensure that water circulates through the case and over the larvae's gills (Okano *et al.*, 2016; Williams *et al.*, 1987). Heavy, mineral cases may also provide useful ballast in lotic habitats, reducing the effort the larvae needs to put into avoiding entrainment and drift (Otto and Johansson, 1995; Statzner and Holm, 1989; Webster and Webster, 1943).

Previous research concerning sediment use by case-building caddisfly has focused largely on individual species and the functional importance of structures to the larvae (De Gispert *et al.,* 2018; Limm and Power, 2011; Otto and Johansson, 1995), rather than the effects of caddisfly case-building on sediment stability and transport.

1.5 Aim and objectives

Based on the literature reviewed above, the aim of this thesis is to investigate and quantify the effect of case-building caddisfly on the distribution and transport of sediment in rivers. To achieve this, the thesis will encompass two in-situ field studies and two ex-situ flume experiments. Research will be conducted across a range of spatial and temporal scales from the transport of individual caddisfly cases up to understanding seasonal variability in sediment transport by caddisfly at the reach scale. The thesis aims to address three specific objectives:

- 1) To quantify the mass and grain-size of sediment incorporated into caddisfly cases and identify the individual taxa responsible, in riffle habitats of a lowland stream (Chapter 2).
- 2) To determine the effect that case construction by caddisfly larvae has on the hydraulic force required to transport incorporated sediment (Chapter 3).
- To ascertain the importance of caddisfly in transporting and redistributing sediment over the riverbed (Chapters 4 and 5).

1.6 Justification of aim and objectives

Animals can have important zoogeomorphic effects on the transport of sediment in rivers, with consequences for river morphology, habitat and ecosystem services. Whilst the zoogeomorphic effects of a wide range of lotic taxa have been examined, the effects of aquatic invertebrates are understudied in rivers, compared to terrestrial and marine habitats. Case-building caddisfly are widespread and often extremely abundant in rivers and exhibit behaviours with the potential to have important effects on sediment distribution and transport. However, the zoogeomorphic effects of caddisfly case-construction have received little research.

The mass and grain size of sediment used by caddisfly is known for individual cases of some species (e.g. Tolkamp, 1980). However, the quantity of sediment used by the whole caddisfly community at a single site has never been measured. The grain-size used by caddisfly varies between species from fine sand to fine gravel (Statzner *et al.*, 2005; Tolkamp, 1980). Therefore, it is important to identify the mass and grain size distribution of sediment which may be directly affected by case-building caddisfly (i.e. tied up in cases). Objective 1 will address this research gap by quantifying both the mass and grain size characteristics of mineral sediment used by case-building caddisfly. It also helps to understand the potential spatial scale and variability of case-building caddisfly zoogeomorphic effects.

Some case-building caddisfly are known to bioturbate sediment by burrowing (De Nadaï-Monoury *et al.*, 2013) and caddisfly cases can promote precipitation of tufa in calcareous environments (Leggitt and Cushman, 2001), but case-building caddisfly may have a much broader range of unstudied zoogeomorphic mechanisms, hypothesised in Figure 1.4. Case construction involves binding sediment particles together with silk (Hansell, 1968; Okano *et al.*, 2016), producing a composite particle of reduced density and relatively large volume. Caddisfly may therefore modify the entrainment characteristics of sediments incorporated into their cases (Figure 1.4A). The shear stress required to move caddisfly cases varies with species, case size and whether the case is occupied (Otto & Johansson, 1995; Statzner and Holm, 1989; Waringer, 1993), but there has been no systematic examination of the mobility of cases relative to their constituent grains. It is unknown whether case construction will increase or reduce the mobility of constituent sediment; fundamental to understanding the zoogeomorphic effects of case construction. Objective 2 will address this research gap through ex-situ flume experiments, considering three species of empty caddisfly cases, and their constituent mineral grains.

Unlike caddisfly nets and retreats, or mounds built by ants and worms, caddisfly case bioconstructions are often transported around with the larvae. Many caddisfly species crawl large distances (~ 1 – 4 m day⁻¹, Erman, 1986; Jackson *et al.*, 1999; Lancaster *et al.*, 2006), transporting their case and constituent sediment with them. Case-building caddisfly may therefore directly transport sediment, vertically and horizontally over the river-bed (Figure 1.4B & C) and may move sediment upstream (Williams and Williams, 1993; Bergey, 1989; Hart and Resh, 1980). Sediment movement by caddisfly may be an important flux of sediment itself or it may modify the exposure of this sediment to hydraulic forces and affect sediment transport. Therefore, it is important to understand the transport of sediment by caddisfly larvae and the consequences of this movement for sediment distributions and bedload transport in rivers (objective 3).



Figure 1.4. A conceptual diagram showing possible zoogeomorphic impacts of caddisfly in gravel-bed rivers. Previous research has considered nets and retreats and burrowing taxa. This thesis will examine the zoogeomorphic effects of case construction. (A) All case-building caddisfly change the size, shape, mass and density of sediment particles by combining them into cases with potential implications for the mobility of this sediment. (B) Case-building taxa may transport fine sediment vertically, with potential consequences for the flow exposure and mobility of this sediment. (C) Case-building taxa may also transport sediment horizontally over the riverbed.

1.7 Thesis structure

This thesis is comprised of 6 chapters (Figure 1.5). Following the introduction and literature review (This Chapter), the thesis research is divided into four separate projects (Chapters 2-5) which address the thesis aim and objectives. Chapter 2 measures the mass and spatial variability of sediment used by the case-building caddisfly larvae community. This was achieved using a field study to quantify sediment in caddisfly case samples collected from riffle habitats, spaced along a UK stream. Furthermore, Chapter 2 identifies the caddisfly taxa which have the potential to be important to zoogeomorphology, based on their abundance, sediment mass used and behaviour, providing the focus of future chapters. Chapter 2 also considers the importance of abundance and sediment availability in controlling sediment used by the caddisfly population. Chapter 2 therefore addresses thesis objective 1.

Chapters 3 and 4 are both laboratory flume experiments designed to better understand specific mechanisms by which case-building caddisfly may affect sediment transport. Chapter 3 addresses objective 2, investigating whether the construction of cases by caddisfly alters the hydraulic force required to move the sand incorporated into the case. This study is a small-scale flume experiment which compares critical entrainment thresholds for individual empty cases to the sand the cases were constructed from. This was examined for three species with differing case designs and considered the importance of case mass and shape in their entrainment thresholds.

Based on the importance of one species, Glossosomatidae *Agapetus fuscipes*, in Chapters 2 and 3, it was hypothesised that sediment movement by this taxon may be substantial and change the hydraulic exposure of transported sediment. Chapter 4 employs a flume study to measure sediment movement by *A. fuscipes* larvae, under low and high flow velocity and in gravel-beds with high and low grain protrusion. Chapter 4 therefore focusses on objective 3, documenting sediment movement by one caddisfly species in detail.

Chapter 5 expands the temporal and spatial scale of previous chapters to determine the role of case-building caddisfly in sediment transport, seasonally and at the reach scale. Chapter 5 uses bedload traps placed in a lowland stream to isolate and compare the geophysical and case-building caddisfly contributions to bedload transport over an annual period. Chapter 5 further addresses objective 3; considering seasonal variations and the contribution of different caddisfly taxa to sediment transport.

Chapter 6 discusses the results of Chapters 2-5. The fulfilment of thesis objectives are reviewed, followed by an updated version of Figure 1.4; discussing additional understanding of the

20

mechanisms of case-building caddisfly zoogeomorphology as a result of this thesis research. Subsequently the zoogeomorphic effects of caddisfly taxa from each architecture group (Figure 1.3) are discussed. Last, the challenges in upscaling the thesis results and the potential implications of caddisfly zoogeomophology for river geomorphology and ecology are examined.



Figure 1.5. Structure of the thesis outlining the six chapters.



Caddisfly in an alpine river, near Briançon, France.

Note: Figures at the head of each chapter are not included in the figure list or referred to in the text because they are not part of the scientific argument, rather are used to enhance the presentation and provide a break between each chapter.

Chapter 2. A spatial quantification of sediment use by the casebuilding caddisfly community

This chapter is based largely on a published manuscript but has been adapted to fit the thesis.

Published as: Mason, R.J., Rice, S.P., Wood, P.J., Johnson, M.F., 2019. The zoogeomorphology of case-building caddisfly: Quantifying sediment use. Earth Surface Processes and Landforms. 44, 2510 – 2525. DOI: 10.1002/esp.4670.

This chapter differs from the published version in the following ways:

- 1) Paper introduction has been shortened and elements of the published version have been included in the thesis introduction.
- 2) Minor edits have been made to the methods, results and discussion.

Author contributions:

Richard J Mason: Study design, field work, laboratory and data analysis, and chapter write-up.

Stephen P Rice, Paul J Wood and Matthew F Johnson: Advised with fieldwork, laboratory procedure, data analysis, and reviewed the published manuscript and chapter.

2.1 Introduction

Gravel-bed rivers are characterised by coarse sediment (>2 mm diameter) but often contain substantial quantities of fine sediment (< 2 mm), with bed sediment sometimes exceeding 30% sand by mass (Church, 2010). This fine sediment is important to sediment transport (Wilcock and Kenworthy, 2002), near-bed hydraulics (Laronne *et al.*, 2001) and habitat structure and availability (Wharton *et al.*, 2017; Wilkes *et al.*, 2019). Sand fractions constitute the majority of the sediment flux in many rivers and the sand content of gravel-beds is a control on the movement of gravels as bedload (Jackson and Beschta, 1984; Ikeda and Iseya, 1988; Wilcock *et al.*, 2001; Wilcock and Crowe, 2003). Therefore, understanding the transport and distribution of sand in rivers is important.

During case construction, caddisfly cement sand-sized sediment together with silk, creating aggregate particles, with potential zoogeomorphic implications (Figure 1.4). In many streams and rivers where caddisfly are abundant, case construction could involve the use of substantial quantities of sand. Previous research has considered the sediment used by individual caddisfly larvae for several species (e.g. Hansell, 1972; Tolkamp, 1980), but there has been little research on sediment used by groups of individuals and communities. Statzner *et al.*, (2005) estimated that the pupal cases of Rhyacophilidae and Hydropsychidae (groups v and i respectively, Figure 1.3) use between 1 and 3 kg m⁻² (Statzner, 2011; Statzner, *et al.*, 2005). Preliminary studies indicate that the saddle-cased species *Agapetus fuscipes* (Glossosomatidae, Figure 1.3iii) used approximately 1.4 tonnes km⁻¹ of sand to build their cases on the River Dove, U.K.

From a zoogeomorphological perspective, it is important to measure the mass of sand used by case-building caddisfly communities, to determine the mass of sediment which may be affected by their bioconstruction activities. Furthermore, it is important to understand which grain sizes are most affected by caddisfly bioconstruction. Individual caddisfly species use a wide range of grain sizes (~0.105 mm up to 5 mm; Statzner *et al.*, 2005; Tolkamp, 1980, Hansell, 1968; De Gispert *et al.*, 2018; Okano & Kikuchi, 2012) and its unknown what size range the caddisfly community will predominantly use. Knowing the species of caddisfly which dominate sediment use is also important because different taxa vary considerably in behaviour and therefore their potential zoogeomorphic effects.

The mass and grain sizes used by the caddisfly community will be controlled by the composition and abundance of caddisfly larvae as well as any variability in case design between individuals of the same species. The distribution of caddisfly is a result of environmental controls (including flow velocity, oxygen availability and bed sediment characteristics), biological interactions (e.g. competition and predation) and historical factors (Southwood, 1977, Murphy and Davy-Bowker, 2005). The abundance and case designs of caddisfly larvae varies seasonally (e.g. some species only use mineral sediment grains when building pupal cases; Figure 1.3v; Wiggins, 2004). Furthermore, any impacts of case-building caddisfly on sediment stability will also depend on the geomorphic and hydrological context of the river environment. Consequently, both sediment use and the potential for zoogeomorphic effects are likely to vary spatially and temporally.

This chapter quantifies the mineral sediment use by case-building caddisfly in a UK stream (thesis objective 1). Sampling was restricted to riffle habitats because these typically support a greater abundance and diversity of caddisfly taxa (Brown and Brussock, 1991) and the lowest quantity of fine bed-sediments, due to bed armouring (Statzner, 2011). Specifically, the grain size distribution and quantity of sediment used by each of the case-building taxa present are measured and compared to sediment use between taxa. The spatial variation of sediment was investigated in relation to the abundance of caddisfly taxa and the availability of sediment in the bed; both likely to be important controls on case design and sediment use. These results provide important information necessary to design ex-situ experiments and field studies to investigate the potential zoogeomorphic role of case-building caddisfly (Figure 1.4). Furthermore, knowledge of the controls on sediment use by caddisfly larvae will begin to address the potential extent and importance of case-building caddisfly for fluvial geomorphology. For example, if caddisfly adapt their case design to local conditions their presence would be expected to be less constrained by the availability of specific grain size fractions and consequently they may have a more widespread zoogeomorphic impact. This chapter addresses objective 1 of the thesis via three specific questions:

- 1. What is the mass and grain size of sediment used in cases by different caddisfly taxa and by the whole caddisfly community?
- 2. How does the mass and grain size of sediment used by the caddisfly community and individual taxa vary spatially?
- 3. To what extent do variations in the mass and grain size of sediment used in cases reflect differences in taxon abundance and sediment availability?

2.2 Methods

2.2.1 Field sampling

Caddisfly and bed sediment samples were collected from riffle habitats in Wood Brook, Leicestershire, UK (Figure 2.1). Wood Brook is a $1^{st} - 2^{nd}$ order gravel-bed stream with an average width of 1.5 m and depth of less than 0.1 m in the riffles studied. Wood Brook rises at an elevation of 200 m above sea level and flows for approximately 10 km to its confluence with the River Soar. The surficial geology is largely mudstones, siltstone and alluvium. The median discharge of Wood Brook for the five years prior to sampling was 0.03 m³ s⁻¹. This was calculated from 15 minute resolution stage data and spot discharge measurements, fitted with a rating curve, provided by the Environment Agency. The slope of Wood Brook over the study area was 0.02 and the mean D₅₀ of surface sediment was 38 mm.



Figure 2.1. Study site and sampling strategy. (A) Wood Brook (drawn from EDINA, 2018). Three sites were sampled; upper, mid and lower. Mid site: 1° 13.4124' W, 52° 45.0648' N. (B) Three riffles were sampled, nested within these three sites. (C) Three samples were taken within each riffle (i. – iii.). In each case, caddisfly larvae (L) and sediment (S) were sampled from proximate locations. (D) Wood Brook at the mid site, riffle 2.

Caddisfly and sediment samples were collected from three sites along Wood Brook, (referred to as upper, mid and lower; Figure 2.1) in order to capture variability in bed sediment availability and caddisfly communities, associated with longitudinal variations in grain size, channel morphology and local habitat characteristics. Nested within each site, three riffles were selected (each >6 m in length; Figure 2.1) and three longitudinally distributed samples collected from each, making a total of 27 units (i.e. 3 samples x 3 riffles x 3 sites). Caddisfly communities and case characteristics are known to vary seasonally, therefore all samples were collected during a one-month period between 28^{th} March and 26^{th} April 2017, recognised as an optimum time for instream macroinvertebrate diversity and therefore likely to be the peak time for sediment use within caddisfly cases.

Caddisfly larvae samples were collected using a Surber sampler (Surber 1937) with a basal area of 0.1 m^2 (330 x 310 mm frame with a 1 mm mesh net), providing a fully quantitative measure of macroinvertebrate density (Everall *et al.*, 2017). The surface sediment layer was agitated by hand so that macroinvertebrates were carried into the net by the current and larger particles were individually inspected for larvae and cases. A similar Surber sampler frame, with a 0.1 mm net to retain fine sediments, was used to collect a benthic sediment scoop from a location immediately adjacent to the caddisfly sample (Figure 2.1C). The bed sediment sample was used to estimate the sediment available to caddisfly for case building, although this will vary depending on the mobility of the individual species (Statzner, 2011). Both caddisfly larvae and sediment were sampled to a depth of 0.05 m (approximately 1.4 x bed D₅₀).

2.2.2 Laboratory analysis

Caddisfly samples were preserved in 4% formaldehyde solution, washed through a 0.25 mm mesh sieve and manually processed for caddisfly larvae, pupae and cases, which were stored in 70% industrial methylated spirit. Most larvae were identified to species level (using Edington and Hildrew, 1995 and Wallace *et al.*, 2003), although early instar taxa (e.g. *Athripsodes* sp.) were not identified beyond genus. Larvae and case length were measured to the nearest mm (Tolkamp, 1980). When more than 10 individuals of a species were present, length was estimated for a random subsample (mean n = 12). Cases in each sample were combined within each taxon for grain size analysis (regardless of size or instar). For most taxa, only cases that were complete and retained their original length and width, were analysed, although Rhyacophilidae and Hydropsychidae pupae were included even if incomplete due to low total abundance.
After preliminary trials, a methodology was developed to break down the silk structure of cases to disaggregate the mineral particles without damaging them. First, caddisfly cases were immersed in 30% hydrogen peroxide (H₂O₂) at 80°C for at least five hours and stirred periodically. For cases with high organic content some larger organic particles were removed manually. H₂O₂ was then removed by washing the samples with deionised water through a 38 mm diameter, 0.063 mm aperture, sieve and then dried overnight at 80°C. The disaggregated sediment was finally mechanically sieved through 5.6 mm to 0.063 mm (38 mm diameter sieves), at half phi intervals and weighed. Individual fractions were weighed on a 4 decimal place scale. To reduce sieving error, case samples with mass less than 0.01 g, or samples where the mass lost during the sieving process exceeded 5%, were removed from the analysis (9 and 1 samples removed respectively, leaving 151 samples). Bed sediment samples were dried at 105°C for 12 hours and then sieved into half-phi intervals down to 0.063 mm using a mechanical sieve shaker.

2.3 Data Analysis

2.3.1 Sediment use by individual taxa

The grain size distribution (GSD) of an average individual case of each taxon in each sample, was calculated by dividing the mass of each grain size fraction by the number of cases in that sieved sample. Percentiles were determined for each of these GSDs. The mass of an average individual case was estimated as the sum of all size fractions divided by the number of cases sieved. To compare the mass and size distribution of sediment used in individual cases of different taxa at the river scale (Question 1) the GSDs, percentiles and mass were averaged across all samples in which that taxon used sediment. The mean and standard deviation of larvae length, case length, case mass, $\overline{D50}$, $\overline{D5}$ and $\overline{D95}$ are reported for each taxon, where the bar indicates the mean of all values across the river. The $\overline{D5}$ and $\overline{D95}$ are used to give a measure of the range of grain sizes used by each taxon that excludes the most extreme sizes.

2.3.2 Sediment use by the whole caddisfly community in a Surber sample

To calculate the total mass of sediment used by each taxon in each Surber sample, the sediment use by an individual of that taxon was multiplied by the number of *cases* recorded in the Surber sample (note: not all cases in a sample were subject to sieve analysis). The fragility of Glossosomatidae cases meant that the number of larvae recorded was typically greater than the number of surviving cases. Therefore, the total sediment used by this taxon was determined by multiplying average sediment use per case by the number of *larvae* present. The total sediment

mass used by the case-building caddisfly community in each Surber sample was then calculated as the sum of the totals for each taxon. The GSD was similarly calculated, multiplying the mass of each grain size fraction in an individual case by the number of cases (or larvae for Glossosomatidae) present. The community GSD was calculated as the sum of each grain size fraction used by each taxon. The mean and standard deviation, percentiles, skew and kurtosis were calculated for community sediment use. The mass of sediment used by caddisfly larvae in each sample was scaled from 0.1 m² samples, to 1 m² for ease of interpretation.

2.3.3 Spatial variability in sediment use by the caddisfly community

In order to address Question 2, the mean mass and GSD of sediment used by the caddisfly community in each sample were grouped into the three sites (upper, mid and lower), with nine samples in each (i.e. 3 replications in 3 riffles within each site). A Shapiro-Wilk test revealed that these predominately displayed non-normal distributions and therefore, a non-parametric Kruskal-Wallis test was used to compare the variability in the mass and grain size percentiles (D₅₀, D₁₆ and D₈₄) of the community-level data between sites. Subsequently, Dunn's tests were used to determine which sites were significantly different from each other. Shapiro-Wilk, Kruskal-Wallis and Dunn's tests were conducted using the *stats* package in R Studio (version 3.3.3, R Core Team, 2017).

2.3.4 Is caddisfly abundance a control on the mass of sediment used?

To address this question, I assumed that differences in the total sediment mass used in different locations (i.e. variability between samples) reflect: 1) differences in the abundance and composition of the caddisfly taxa present, and 2) differences in the mass of cases built by individuals of the same species (potentially reflecting unknown biotic and abiotic drivers; including sediment availability). The relative importance of these factors was investigated using linear regression between abundance (independent variable) and the total mass of sediment used (dependant variable) for each taxon and the whole community. A high value (R²) indicates that spatial variability in sediment use is associated with the abundance of each taxon and therefore that within species variability in case mass is low. Samples without sediment use by a particular taxon were excluded from the regression analysis but samples with sediment use (cases), but no larvae (larvae may have been lost in sampling or emerged as adults), were included. *Agapetus fuscipes* (Glossosomatidae) were the focus of further investigation of within species variability because they were the most abundant taxon and accounted for most sediment use at the river scale. To determine if variability in *A. fuscipes* case mass corresponded with variability in the size of sediment used, the mean and standard deviation of GSDs were plotted for each site.

29

Furthermore, the correlations between mean case mass, mean case D_{50} and mean larvae length across all sites for this taxon were used to investigate potential explanations for the variability in case mass.

2.3.5 Relation between sediment availability and sediment use

To investigate how sediment use varied with sediment availability, the GSDs of mean sediment use by the caddisfly community were compared to the bed sediment availability, both at the river scale and for each site. The mass of sediment used by the caddisfly community was divided by the mass of available sediment and multiplied by 100 to give the percentage of sediment used. This was done for each grain size, for each sample, and averaged for each site and the whole river.

2.4 Results

2.4.1 Sediment use by individual caddisfly taxa

There was considerable variability in the mass and size distribution of cases of individual taxa (Table 2.1, Figure 2.2). The tubular cases of Leptoceridae and Lepidostomatidae used fine sand $(D_{50} = 0.17 \text{ mm})$ and mean mass for both taxa was only 0.001 g; less than 1% of the mean Limnephilidae case mass (0.146 g), which used coarser sediment ($D_{50} = 1.12 \text{ mm}$). Pupal cases of Rhyacophilidae (free-living) and Hydropsychidae (net and retreat builder) were substantially heavier (mean mass = 0.83 g) and coarser ($D_{50} = 4 \text{ mm}$) than all other taxa (Table 2.1, Figure 2.2). Goeridae cases were unusual because coarser particles secured to the case edges resulted in a bimodal sediment distribution. The size distribution of Glossosomatidae cases (dome cases, $D_{50} = 0.94 \text{ mm}$) was similar to Limnephilidae (tube cases), but the mean mass of each individual case was less (0.02 g, Table 2.1, Figure 2.2). Therefore, case-building caddisfly exhibited considerable variability in larvae and case size, particle size distribution and case mass, not just between structure groups (Figure 1.3) but also between families and species (Table 2.1, Figure 2.2).

2.4.2 Sediment use by the whole caddisfly community

A diverse caddisfly community was recorded with twenty-four taxa from ten families (Table 2.2) and a mean taxonomic richness of 7.4 taxa per sample (minimum 3, maximum 12). Mean abundance was 2250 larvae m⁻², but this varied from 180 to 7460 larvae m⁻². Caddisfly using mineral sediments for case-building accounted for approximately 50% of the taxa present (Table

2.2), but 94% of individual caddisfly larvae recorded. All taxa combined, the average total mineral mass used by case-building caddisflies was 37.57 g m⁻², but it varied across the 27 samples, ranging from 3.74 to 138.83 g m⁻² (Figure 2.3A).

Due to the differences in case GSDs between taxa, the size range of mineral sediment used by the community was broad, from 0.063 mm to 11 mm (Figure 2.2). Therefore, caddisfly utilise the entire sand size range as well as fine gravel particles for case construction (0.063 mm < sand <2 mm < gravel; Wentworth, 1922). On average, sediment use was 84% sand and 16% fine gravel. The average particle-size distribution of sediment used (Figure 2.3B) was bimodal with a primary mode between 1 - 1.4 mm and another mode at 4 - 5.6 mm. The D₅₀ was 1.06 mm and the distribution is fairly symmetrical (Skew *Sk* : -0.008) and leptokurtic (Kurtosis *K*: 1.238). The total mass of sediment used by case-building caddisfly (all samples combined) was dominated by Glossosomatidae (64%, Figure 2.3B), followed by pupae of Rhyacophilidae and Hydropsychidae (15%), and Limnephilidae (11%) with other taxa using less than 5%.



Figure 2.2. Particle size distribution of the cases of caddisfly taxa. (A) Grain size distribution averaged across all samples, for the six case-building families and for the pupal cases of Rhyacophilidae and Hydropsychidae combined. Particle size for (A) refers to the upper limit or passing sieve. (B) Spread ($\overline{D5} - \overline{D95}$) and median ($\overline{D50}$) of the particle size distribution for each taxonomic group, averaged across all samples. In the legend, brackets indicate the total number of cases analysed and the number of samples in which the taxa were present (out of a total of 27).

Table 2.1. Characteristics of larvae and cases for each case-building caddisfly taxonomic group and individual species of Limnephilidae. Values shown are the mean of all 27 samples, with standard deviation in brackets (only one incidence of *H. radiatus* and *C. villosa*). †Pupal cases of Rhyacophilidae and Hydropsychidae larvae.

	Larvae			Case				
Family	Number	Length (mm)	Length (mm)	Weight (g)	D50 (mm)	D5 (mm)	D95 (mm)	
Glossosomatidae	180 (178)	3.6 (1.04)	4.6 (1.13)	0.020 (0.012)	0.94 (0.24)	0.37 (0.07)	1.82 (0.36)	
Goeridae	6 (5)	5.4 (0.93)	5.7 (1.15)	0.027 (0.011)	1.18 (0.22)	0.22 (0.02)	2.10 (0.31)	
Sericostomatidae	12 (14)	6.5 (3.59)	8.4 (2.55)	0.012 (0.007)	0.27 (0.03)	0.13 (0.03)	0.53 (0.20)	
Lepidostomatidae	18 (15) 21 (23)	3.1 (1.53) 1.8 (1.05)	5.4 (0.60) 3 5 (0.36)	0.001 (0.0003)	0.17 (0.01)	0.09 (0.004) 0.09 (0.01)	0.29 (0.03) 0.30 (0.03)	
Rhy. and Hyd. Pupae [†]	-	-	-	0.83 (0.68)	4.00 (1.29)	1.34 (0.33)	5.95 (2.24)	
Limnephilidae (Average)	4 (3)	16.5 (1.68)	17.2 (2.27)	0.146 (0.050)	1.12 (0.15)	0.46 (0.06)	2.14 (0.26)	
Potamophylax cingulatus	-	-	-	0.16 (0.05)	1.12 (0.14)	0.47 (0.05)	2.13 (0.25)	
Potamophylax latipennis	-	-	-	0.17 (0.01)	1.31 (0.17)	0.48 (0.05)	2.24 (0.36)	
Micropterna sequax	-	-	-	0.11 (0.05)	0.93 (0.37)	0.40 (0.11)	1.77 (0.53)	
Halesus radiatus	-	-	-	0.063	1.39	0.51	2.64	
Chaetopteryx villosa	-	-	-	0.028	0.57	0.30	1.04	



Figure 2.3. Total sediment use by the case-building caddisfly community and constituent taxa. (A) Boxplot of total community sediment use from all samples. The box denotes the median and interquartile range (inclusive of median), whiskers show interquartile range multiplied by 1.5 and the mean is indicated by x (37.57 g m⁻²). (B) Size distribution of mean mass of sediment use across all sites for the entire case-building caddisfly community and for each taxon, including only those which accounted for >3 % of total sediment use. The standard deviation in community sediment use is also shown and indicates substantial variability between samples. Legend indicates community and taxon with percentage of total sediment use in brackets. Particle size indicates passing sieve.

Table 2.2. Caddisfly taxa found in this chapter. Taxa are sorted according to their behaviour (after Wiggins, 1996); (1) taxa that used mineral sediments in their cases were included in sediment analysis and (2) cases that were not analysed for sediment content. [†]Cases for which a larvae was never found and therefore could not be identified were classified as unknown. Unknown cases were all similar in architecture and are believed to be discarded pupal cases of Leptoceridae species. [‡]Pupal cases of Hydropsychidae and Rhyacophilidae were identified only to family level but are likely to be the same species found elsewhere.

Behaviour group		Family	Genus and Species
Saddle case makers		Glossosomatidae	Agapetus fuscipes
Tube case makers		Goeridae	Silo pallipes
	1) (Sericostomatidae	Sericostoma personatum
	Cas	Lepidostomatidae	Lepidostoma hirtum
	es i	Leptoceridae	Athripsodes sp.
	ncli	Limnephilidae	Potamophylax cingulatus
	ude		Potamophylax latipennis
	ă T		Micropterna sequax
	l se		Halesus radiatus
	din		Chaetopteryx villosa
	lent		Empty Limnephilidae cases
	ana	Unknown [†]	
Net spinners	alys	Hydropsychidae	Hydropsyche siltalai,
	<u>v</u> .		pellucidula and instabilis t
Free living		Rhyacophilidae	Rhyacophila dorsalis ‡
Tube case makers		Leptoceridae	Mystatices longicornis
			Mystatices azurea
		Limnephilidae	Stenophylax sp.
	(2)		Limnephilus lunatus
Purse case makers	Z	Hydroptilidae	Hydroptila sp.
Net spinners	tin	Hydropsychidae	Hydropsyche siltalai
	cluc		Hydropsyche pellucidula
	ded		Hydropsyche instabilis
		Polycentropidae	Polycentropus flavomaculatus
			Polycentropus kingi
Free living		Rhyacophilidae	Rhyacophila dorsalis



Figure 2.4. Variability in sediment use between sites (upper, mid and lower). (A) Boxplot showing the mass of sediment used by the caddisfly community at each site. (B) Boxplot showing the D_{50} of sediment used at each site. For both boxplots, the box denotes the median and interquartile range (inclusive of median), whiskers show interquartile range multiplied by 1.5 and the mean is indicated by x. (C)-(E) Particle size distribution of mean total sediment use for the caddisfly community and constituent taxa, for each site. Only taxa which account for more than 3 % of total sediment use are shown. Particle size indicates passing sieve.

2.4.3 Spatial variability in community sediment use

Kruskal-Wallis tests confirmed that there were no significant differences in the total mass of sediment used between the three sites (Table 2.3, mean at the upper site = 29.59, mid = 38.02, lower = 34.11 g m⁻², Figure 2.4A). However, the size of sediment used was significantly finer at the mid site than the upper and lower sites for D_{50} , D_{84} and D_{16} (Table 2.3, Figure 2.4B, mean D_{50} at the upper site = 1.61, mid = 0.79, lower = 1.11 mm). The upper and lower site were not significantly different for any grain size parameter (Table 2.3). The mass of sediment used by each taxon varied between sites (Figure 2.4C-E). At the upper site, Glossosomatidae and Limnephilidae accounted for a similar proportion of the 0.5 - 4 mm size fraction while at the mid and lower sites sediment use was dominated by Glossosomatidae. Both the upper and lower sites have community GSDs skewed towards coarser particles due to: 1) pupal cases of Rhyacophilidae and Hydropsychidae which dominated sediment use at larger grain sizes (2.8 – 11 mm) and, 2) differences in the size-distribution of sediment used by Glossosomatidae.

Table 2.3. Results of Kruskal-Wallis and Dunn's post-hoc tests to determine the variability in sediment use by caddisfly larvae between sites (upper, mid and lower), relative to within site variability. Chi-squared values reported for Kruskal-Wallis and Z statistic for Dunn's test. For both significance is indicated by * 0.05 > P > 0.01, ** 0.01 > P > 0.001, *** P < 0.001.

Parameter	Kruskal- Wallis (K)	Dunns (Z)				
		lower-mid	mid-upper	upper-lower		
D ₅₀	16.5***	3.44***	-3.59***	-0.15		
D ₁₆	7.3*	1.96*	-2.58**	-0.62		
D ₈₄	13.5**	2.94**	-3.39***	-0.45		
Mass	4.0	-1.34	1.96*	0.62		

2.4.4 Relation between abundance and sediment use

Abundance of the caddisfly community (all taxa) was a poor predictor of the mass of sediment used, explaining only 17% of variation (Table 2.4). This means that differences in case mass between individuals are important at a community level. When considering each taxon separately, for some taxa their abundance displayed a strong association with sediment mass used (Leptoceridae, $R^2 = 0.79$ and Goeridae, $R^2 = 0.75$), but for others the association was weak (Glossosomatidae, $R^2 = 0.34$). When grouped by site, the association between abundance and sediment use was stronger for some taxa, including Glossosomatidae, which displayed a strong linear association between sediment mass and abundance at the upper and lower sites but not the mid (Table 2.4, Figure 2.5A). Within each of the three sites, Glossosomatidae larvae built cases

of similar mass and consequently Glossosomatidae abundance had a strong association with total Glossosomatidae sediment use. However, at the river scale there was considerable inter-site variability in case mass.

In addition to having lower mass, Glossosomatidae cases at the mid site contained finer sediment particles (Figure 2.5B). There was a negative association between Glossosomatidae larvae abundance and mean case mass across all sites (R = -0.55, Figure 2.5C). However, in samples with low abundance, case mass varied substantially, whereas case mass was always lower at high abundance. A very strong correlation exists between Glossosomatidae mean case mass and D_{50} (Figure 2.5D) and between case mass and mean larvae length (Figure 2.5E). Overall, the abundance of most taxa, particularly Glossosomatidae, is a poor predictor of sediment mass used by this taxon due to within-taxon variations in the mass of cases built and the grain sizes used.

2.4.5 Relation between sediment availability and sediment use

The mean D_{50} of bed material samples was lowest at the mid site (32 mm) and similar at upper and lower sites (38 and 40 mm respectively). Considering only the size range used by casebuilding caddisfly in this study, the average mass of bed sediment availability increased with grain size (Figure 2.6A). Some samples displayed a bimodal distribution with a second peak in the 0.25 – 0.5 mm size range, particularly at the mid site. Sediment use was also skewed towards finer sediment at the mid site (Figure 2.6B). Sediment availability for all size fractions greatly exceeded the mass of sediment used by caddisfly larvae, resulting in a low percentage used overall (Figure 2.6C). As an average across all samples, the percentage of sediment use, Figure 2.6B). In this size fraction, the percentage used ranged considerably from 0.25% up to 24.86% of available sediment. Therefore, sediment of all grain sizes was abundant and available to taxa at all sites. Despite this, use of sediment by the case-building caddisfly community and Glossosomatidae were both skewed towards finer sediment at the mid site where this sediment was more abundant.



Figure 2.5. Variability in sediment use by *A. fuscipes* (Glossosomatidae) individuals between sites. (A) Plot of the relation between abundance and sediment use by this taxon giving the R² values in Table 2.4. Within each site abundance is a good indicator of sediment mass used but all sites combined it is poor. (B) Particle size distribution for individual *A. fuscipes* cases as a mean of all samples and for each site independently. Particle size refers to passing sieve. (C)-(E) Correlation plots between mass of an individual *A. fuscipes* case and variables expected to explain spatial variability in case mass. (C) mean *A. fuscipes* abundance, (D) mean case D_{50} and (E) mean larvae length.

Table 2.4. R^2 values for linear regression between abundance (independent) and the mass of sediment used (dependent) for each taxon across all samples (whole river) and for each site (upper, mid and lower) independently. The regression was also conducted for the whole caddisfly community. The number of samples included in each regression (n) is shown in brackets (regression not conducted when n < 5). Significance is indicated by * 0.05 > P > 0.01, ** 0.01 > P > 0.001, *** P < 0.001.

	All	Upper	Mid	Lower
Glossosomatidae	0.34(25)**	0.71(7)*	0.15(9)	0.99(9)***
Goeridae	0.75(16)***	0.80(6)*	0.75(9)**	(1)
Sericostomatidae	0.38(24)**	0.05(6)	0.78(9)**	0.25(9)
Lepidostomatidae	0.52(6)	(0)	(0)	0.52(6)
Leptoceridae	0.79(13)***	(2)	0.84(9)***	(2)
Limnephilidae	0.55(18)***	0.25(9)	0.93(8)***	(1)
Community	0.17(27)*	0.09(9)	0.14(9)	0.97(9)***



Figure 2.6. Sediment use in relation to available sediment in the river-bed. Particle-size distributions shown as a mean for the river as a whole (with standard deviation) and each site independently. (A) Mass of bed sediment available to case-building caddisfly. (B) Mass of sediment used by case-building caddisfly communities. (C) Percentage of available sediment that is used by case-building caddisfly. Note that due to the use of 0.1 mm diameter net the quantity of sediment available and percentage of sediment used are not accurate below 0.125 mm. Particle size indicates passing sieve.

2.5 Discussion

2.5.1 Sediment use by individual taxa and the case-building caddisfly community

The caddisfly community of Wood Brook was diverse, both in terms of species richness and the design of silk structures, with all five case categories recorded (Figure 1.3, Table 2.2). About half of taxa were case-building caddisfly utilising mineral sediments, but due to their high abundance they accounted for over 90% of individuals. This suggests that if the mechanisms identified in Figure 1.4 are significant, case-building caddisfly could be important zoogeomorphic engineers in this stream. The considerable variability in the size of sediment used by different taxa in this study (Figure 2.2) probably reflects differences in the function of the larval cases (Otto and Svensson, 1980; Wiggins, 2004). For example, the fine grains used in Sericostomatidae cases (D₅₀ = 0.27, Table 2.1) result in smooth, tapered cases, which may facilitate burrowing. Most taxa in this study used predominantly mineral sediments (Table 2.2), which require a much greater investment of energy than organic cases (Otto and Svensson, 1980). Consequently, using mineral sediment must be advantageous to larvae in gravel-bed rivers, perhaps providing ballast, protection from crushing by moving sediment and reduced risk of predation (Otto and Johansson, 1995; Statzner and Holm, 1989; Webster and Webster, 1943). The results of this study demonstrate that both the size and magnitude of mineral sediment used is dependent on the caddisfly taxa present, and consequently, any geomorphic effect is likely to vary between taxa. As a result of the diversity in grain size used by different caddisfly taxa, the community used sediment from 0.063 to 11 mm in diameter (Figure 2.3B). Consequently, caddisfly have the potential to directly affect the distribution and mobility of a broad sediment size range. However, sediment use was predominantly coarse sand and fine gravel (Figure 2.3B), which are important size fractions geomorphologically, as they are often the most mobile bed-material size fractions (Ashworth and Ferguson, 1989; Jackson and Beschta, 1984; Kuhnle, 1993).

2.5.2 Sediment use by key taxa and potential zoogeomorphological importance

Several taxa are of particular zoogeomorphic interest. Glossosomatidae accounted for over 60% of sediment use and were by far the most abundant taxa (up to 6710 larvae m⁻²). Glossosomatidae cases are built from sediment particles accessible to them on the bed surface (Becker, 2001; Marchant, 1988; Statzner, 2011) and usually reside on the exposed upper surfaces of larger gravel particles where the flow of water facilitates respiration (Becker, 2005; Morris *et*

al., 2015; Morris and Hondzo, 2013a). Glossosomatidae may therefore have important effects on sand availability and hydraulic exposure at the bed-surface, where they are likely to affect downstream sediment transport

Despite their low abundance in this study, Rhyacophilidae and Hydropsychidae pupal cases were constructed from large particles and, therefore accounted for 15% of the overall sediment used by case-building caddisfly by mass (Figure 2.3B), and nearly 50% of sediment used at the upper site (Figure 2.4C). Statzner (2011) and Statzner et al. (2005) measured sediment use by these taxa in the Furan river, France, and found them to use substantial quantities of sediment (up to ~3 kg m⁻²). They found considerably higher abundances of pupal cases (250 m⁻² for *H. siltalai*; Statzner, 2011), compared to the average of 3 cases m^{-2} in this study. This is probably due to both higher abundances of larvae in the Furan river than Wood Brook, and the time of year sampled. Statzner et al. (2005) sampled in June and August while our study was conducted in April, prior to pupation for most species (Wallace et al., 2003). In situations where pupal cases of free-living and net building taxa are abundant (> 1000 larvae m⁻², Cardinale et al., 2004) sediment use is expected to be skewed towards coarser particles. During pupation, these taxa may have geomorphic effects by modifying the hydraulic properties of coarse sand and gravel (Figure 1.1). Pupal cases are usually attached to more stable particles (unlike mobile cases) and consequently the mobility of constituent sediment is likely to be reduced. Cases may also reduce the mobility of the larger particles they are attached to (Nunokawa et al., 2008).

2.5.3 Spatial variability in community sediment use and the distribution of case-building caddisfly taxa

Both the size and mass of sediment used were spatially variable at the river scale (Figure 2.4), even within this relatively narrow range of environmental conditions (riffle habitats in ~3.6 km of stream; Figure 2.1A). There was considerable variability in the mass of sediment used (4 – 139 g m⁻²) but this was not structured by site (Figure 2.4A). This is potentially related to the patchy distribution of benthic invertebrates in rivers (Pringle *et al.*, 1988); a result of habitat variability and biological controls (Southwood, 1977; Cummins and Lauff, 1969). The size of sediment used by the caddisfly community was significantly lower at the mid site than the upper and lower sites (Table 2.3, Figure 2.4B). This is interesting because the greatest environmental difference would be expected between the upper and lower sites (Rice and Church, 1998; Vannote *et al.*, 1980). Nevertheless, the mid site had the lowest slope, bed sediment D₅₀ and depth, suggesting that it was distinct from the other sites in terms of habitat characteristics.

The greater use of fine sediments by the caddisfly community at the mid site is the result of both the absence of Rhyacophilidae and Hydropsychidae pupal cases and the use of finer sediments by Glossosomatidae larvae (Figure 2.4D, Figure 2.5B). The upper site had the coarsest D_{50} where Rhyacophilidae and Hydropsychidae pupae used a greater proportion of coarse sediment grains, resulting in a bimodal sediment distribution with a primary peak in the coarse sand range and a second peak of fine gravel for the site (4 – 5.6 mm, Figure 2.4C). The upper site had, on average, the lowest abundance of Glossosomatidae and the mass of each case was very variable (Figure 2.5C). This could be due to shading by riparian vegetation reducing the availability of their algae food source (Mcneely and Power, 2007) and increasing inputs of allochthonous organic matter, favouring shredders (Vannote *et al.*, 1980), such as the Limnephilidae larvae present.

2.5.4 Sediment use in relation to the abundance of taxa and availability of sediment

The weak association between the abundance of the whole case-building caddisfly community and the total mass of sediment used (Table 2.4) is unsurprising because the mass of cases varied by several orders of magnitude between taxa (Table 2.1). Therefore, to achieve a 0.85 g increase in sediment use would require just one Rhyacophilidae or Hydropsychidae pupal case but 850 Leptoceridae cases (0.001g each, Table 2.1). Interestingly, abundance of many individual taxa was also only weakly related to the mass of sediment used by that taxa at the river scale (Table 2.4). This indicates that within taxon differences in case mass between samples explained more variability in total sediment use than the abundance of that taxon. In this study, differences in case mass between individuals of the same taxon may be explained to some extent by the presence of multiple instars of larvae, which were combined for grain size analysis. Furthermore, within the Limnephilidae family, the presence of multiple species (with different case designs) and differing use of organic material (Table 2.1) also explains this weak association.

A. fuscipes (Glossosomatidae) cases had lower mass and used finer sediment at the mid site than the upper and lower (Figure 2.5B and C). A. fuscipes larvae at the mid site were also smaller (Figure 2.5E), and less-developed (early instar) larvae are likely to build smaller cases with lower mass. The greater abundances of A. fuscipes at this site may increase competition for resources and stunt larvae development. Alternatively, the greater availability of fine sediment at the mid site (Figure 2.6A) may cause larvae to build cases which contain more fine sediment and consequently weigh less (Figure 2.5D). At the river scale only a small percentage of available sediment was used (average 1 - 1.4 mm fraction = 2.99%; Figure 2.6C) and therefore, it is unlikely that taxa are limited in their case design by access to their preferred grain sizes.

42

Glossosomatidae larvae, however, typically use easily accessible sediment (Becker, 2001). Whilst specific size fractions are not limiting at any site, the greater availability of finer fractions at the mid site may, therefore, explain the lower D_{50} of *A. fuscipes* cases here.

Overall, the percentage of available sediment used in this study is considerably lower than that of Statzner (2011), at an average of 2.99% compared to 15 - 25% for the dominant case sediment size. This is likely to be at least partly due to a difference in sampling technique. Statzner (2011) sampled surface sediments and the first subsurface layer and, whilst they do not specify a precise depth, this is unlikely to be as deep as the 5 cm sampled in this study. In addition, Wood Brook drains agricultural lands and consequently contains a high quantity of fine sediment. There is, however, considerable variability in the proportion of sediment used, up to 25% of the 1 - 1.4 mm size fraction was used at one lower sample. This is explained by a low quantity of fine sediment in this sample, rather than high caddisfly use of sediment. Therefore, in rivers with lower quantities of fine sediment, such as mountain streams (Church, 2010) or downstream of reservoirs (Brandt, 2000), case-building caddisfly may use substantial proportions of the available fine sediment.

2.5.5 Methodological discussion

This chapter presents the first quantification of sediment use by a whole case-building caddisfly community, rather than small numbers of individuals or species. This presented several methodological challenges and required amalgamation of techniques used previously to measure GSDs of individual caddisfly cases (Statzner et al., 2009; Tolkamp, 1980), with those used in assessment of much larger quantities of sediment in geomorphology (Bunte and Abt, 2001). Determining the size distribution of very small masses of sediment (as low as 0.01 g) was difficult, but the use of small diameter sieves (38 mm) reduced sediment loss and allowed comparison with sieved bed-sediment samples (unlike photographic methods; Statzner, 2009). Quantifying sediment use for enough individuals and sites necessitated combining taxa of different instars within each sample. However, sediment distributions are known to change with larvae development, which means that the grain size estimates are broader than many in the literature (e.g. Tolkamp, 1980). Last, the mass and size distribution characteristics of Rhyacophilidae and Hydropsychidae pupal cases are likely to be less accurate than those for the other taxa, due to their low abundance and therefore the low number of individuals analysed. This also necessitated combining the two families which are known to differ in pupal case design (Statzner et al., 2005). Nevertheless, within these constraints, the results present an important

step towards understanding the magnitude of sediment use by case-building caddisfly and the effect of cases on fine-sediment sorting and transport in rivers.

2.6 Summary

This study quantifies sediment use by the case-building caddisfly community from riffle habitats in a small stream. In doing so it extends research on both the ecology of case building and the zoogeomorphic role of caddisfly in rivers (Albertson *et al.*, 2014a; Cardinale *et al.*, 2004; Johnson *et al.*, 2009). Considerable variability was found in the mass and size distribution of cases between caddisfly taxa at every taxonomic level, between 1) silk structure design groups (Figure 1.3), 2) families (Figure 2.2), 3) species (Table 2.1) and, 4) individuals of the same species (Table 2.4, Figure 2.5). Case mass ranged from 0.001 g to 0.85 g and D₅₀ from 0.17 mm to 4 mm. Therefore, the mechanisms and extent of caddisfly zoogeomorphic effects are likely to vary between taxa. Community sediment use was on average 37.57 g m⁻² and was 84% sand and 16% fine gravel. *A. fuscipes* (Glossosomatidae), accounted for over 60% of sediment use, Rhyacophilidae and Hydropsychidae (15%) and Limnephilidae (11%). These taxa should therefore be the focus of future research concerning the mechanisms of caddisfly zoogeomorphic impact (Figure 1.4) for small gravel-bed streams.

Due to considerable variability in case mass between sites, abundance was weakly related to sediment use for the caddisfly community and for most taxa at the river scale. In this stream, caddisfly used a small percentage of the total sediment available to them for all size fractions (on average < 3%) suggesting that access to preferred sediment was not a limiting factor in case design. Nevertheless, *A. fuscipes* cases varied in case design between sites, with a lower D₅₀ at the mid site where finer sediment was more readily available. Intraspecific variability is rarely considered in zoogeomorphology (Albertson and Allen, 2015). This raises important questions about the replicability of localised biogeomorphic studies which consider only a few individuals, or individuals from a singular site.

Despite variability in community composition, abundance and case architecture, the caddisfly community used sediment in all 27 samples. Consequently, within the riffle habitats studied, sediment use by the caddisfly community is not constrained in space by the presence of a particular species, their abundance or availability of particular grain-size fractions for case construction. Furthermore, individual taxa appear to be adaptable in their use of sediment. The distribution of *A. fuscipes,* in particular, is not limited by the availability of a specific range of grain sizes and consequently may have widespread zoogeomorphic effects. It follows that if caddisfly

cases do have geomorphic effects (mechanisms in Figure 1.4), then these are likely to be widespread across a range of habitats in which case-building taxa are known to be common.

This study provides insights into the size and quantity of sediment use by case-building caddisfly, essential to guide further research on caddisfly zoogeomorphology. Future work is required to understand the specific mechanisms of zoogeomorphic effects (Figure 1.4) and to quantity the effects of case construction by caddisfly on sediment transport in rivers.



Caddisfly cases during laboratory analysis.

Chapter 3. The effect of caddisfly case construction and case design on the entrainment of incorporated sediment.

3.1 Introduction

3.1.1 Controls on sediment transport

Sediment transport is an important control on the morphology of rivers and fluvial habitat, with implications for the ecosystem services they provide and the risks they pose to society. Therefore, understanding and predicting sediment dynamics remains a primary focus of fluvial geomorphology. The transport of sediment is controlled by many variables reflecting characteristics of both the sediment (e.g. size, shape, grain interactions) and the hydraulic environment (Wilcock et al., 2009; Church et al., 1998). Particle size is a dominant control on sediment transport (e.g. Shields, 1936) because larger particles typically have greater mass and consequently require larger, less frequent, hydraulic forces to entrain them (Bridge and Bennett, 1992; Wilcock et al., 2009). Particle shape is also important because it influences how particles respond to the forces applied to them (Perret et al., 2017). Rods and spherical shapes are transported more easily than disc and blade shapes because they are able to roll which reduces friction with the river bed (Carling et al., 1992; Oakey et al., 2005). In river-beds, particles do not exist in isolation and grain interactions, including imbrication, packing, sorting and hiding, substantially modify the mobility of river-beds (Parker and Sutherland, 1990). These factors affect the *critical* or *threshold* shear stress; the hydraulic force at which sediment particles begin to move (Dey and Papanicolaou, 2008; Simões, 2014).

Animals can affect sediment transport processes by modifying both sediment characteristics and local hydraulics (Chapter 1; Rice *et al.*, 2012; Statzner, 2012). Biological processes can modify the size, shape and density of sediment (Dudgeon, 1990), with implications for sediment transport. For example, faecal pellets produced by aquatic invertebrates, such as those of the marine Polychaete *Amphicteis scaphobranchiata*, are easily entrained due to their relatively high protrusion above the bed (pellet D₅₀ approximately 0.5 - 2.5 mm; Taghon *et al.*, 1984). Pellets produced by *Hydrobia ulvae* in intertidal mudflats of the Danish Wadden Sea, lose their cohesion to the clay mudflats on which this snail lives and are therefore preferentially entrained (Austen *et al.*, 1999).

Aggregate particles are also produced in dryland environments by physical processes including the expansion and contraction of soils containing swelling clays, salt efflorescence and the breakup of mud curls (Rust and Nanson, 1989; Simon and Gibling, 2017). In rivers, these aggregates ($D_{50} = 0.13$ mm; Maroulis and Nanson, 1996) commonly move as bedload; maintaining frequent contact with the bed by rolling, hopping or saltating (Maroulis and Nanson, 1996; Rust and Nanson, 1989), although their constituent clay particles would otherwise usually be transported in suspension. The combination of sediment into an aggregate, whether by biological or physical processes, can therefore substantially alter the entrainment and transport of incorporated particles.

During the construction of cases, caddisfly larvae combine grains of sand and fine gravel into a single aggregate (Figure 1.4A). Caddisfly cases are heavier, but larger and less dense than the original grains used in case construction. Furthermore, caddisfly case designs are often shapes that rarely occur in natural river mineral sediments; such as the long hollow cylinders built by tubular case makers (Figure 3.1A&B). Therefore, case construction is likely to modify the hydraulic force required to transport the constituent grains. Caddisfly are abundant and can incorporate substantial quantities of sediment in their cases (Figure 2.3). If case construction changes the mobility of sediment involved in case construction this could be an important zoogeomorphic effect.

3.1.2 Entrainment of caddisfly and the role of the case

Previous research on the entrainment of caddisfly larvae and cases has been approached primarily from an ecological perspective. Drift (downstream transport) is an important means of migration and relocation for many species of aquatic invertebrates in rivers (Naman *et al.*, 2016). Uncontrolled drift can be detrimental (e.g. increasing vulnerability to predation) and therefore aquatic invertebrates have evolved a wide variety of strategies for reducing unintentional entrainment (Statzner *et al.*, 1988). Caddisfly can resist drift *passively*, via adhesive or static friction (due to weight and case roughness), or *actively* via the effort of the larvae to remain attached (Waringer, 1989).

The role of the caddisfly case in reducing entrainment of larvae is much debated (Dodds and Hisaw, 1925; Limm and Power, 2011; Otto and Johansson, 1995). For mineral case-building caddisfly, an increase in case mass does not necessarily increase passive drift resistance. For example, coarse grains stuck to the edges of Goeridae cases are commonly believed to increase ballast. However, these 'ballast' particles contributed little to total current resistance (empty cases with stones = 10.3 cm s^{-1} , without stones = 8.7 cm s^{-1} ; Otto and Johansson, 1995) because the increase in ballast from the added weight of particles was offset by the increased lift the particles provided (Otto and Johansson, 1995; Statzner and Holm, 1989). Nevertheless, due to their much

greater mass, mineral cases presumably provide more passive resistance to drift than organic cases (Dodds and Hisaw, 1925; Webster and Webster, 1943). This would allow larvae to expend less energy actively resisting drift (Waringer, 1989) and reduce the drift distance if larvae are entrained. *Potamophylax latipennis* (Limnephilidae) larvae build large mineral cases (Figure 3.1B) and are usually only entrained for short periods, moving akin to a saltating grain (Lancaster *et al.,* 2006).

Caddisfly case designs vary considerably in shape, size and weight (Figure 3.1), factors which are likely to effect their entrainment thresholds. Even between taxa which build tubular cases, the velocity required to entrain cased caddisfly larvae varies considerably. For example, passive resistance to drift (case and dead larvae) for 5th instar larvae of tubular case-builders (all Limnephilidae) *Chaetoteryx fusca, Drusus monticola* and *Metanoea rhaetica* was between 8 and 10 cm s⁻¹, *Allogamus auricollis* was 21 cm s⁻¹, whilst the larger *P. cingulatus* required 71 cm s⁻¹ (Waringer, 1993). Therefore, the design of caddisfly cases is an important control on their entrainment thresholds.

Whilst considerable research has been conducted on the transport of sand and gravel in rivers, this past work tends to neglect the importance of biology. Similarly, research conducted on entrainment of caddisfly larvae has not considered the consequences of case construction for geomorphology. Caddisfly create aggregate particles which are likely to vary in mobility from their constituent grains. This chapter investigates the effects of case construction by three species with different case designs, on the entrainment of the constituent sand particles. In doing so it adds a geomorphic perspective to existing literature on caddisfly entrainment and drift. This chapter aims to determine what effect case construction by caddisfly larvae has on the hydraulic force required to entrain the fine sediment incorporated in the cases. Entrainment thresholds were measured for empty cases in order to assess the zoogeomorphic effects of the case construction process (thesis objective 2), rather than sediment transport by live larvae (thesis objective 3). Specifically, the following research questions are investigated:

- 1) Is there a difference between the critical entrainment threshold of caddisfly cases and the loose sediment incorporated in the cases?
- 2) How do critical entrainment thresholds vary between species with differing case designs and what aspects of case design (e.g. mass or shape) influence entrainment?

49



Figure 3.1. Examples of case architecture from the three caddisfly species used in this experiment. (A) Sericostomatidae *Sericostoma personatum* constructs a tubular, curved case from fine sand. (B) Limnephilidae *Potamophylax latipennis* uses coarser sand and fine gravel. Cases of both these species are tubular in shape and lined internally with silk. (C) Glossosomatidae *Agapetus fuscipes* dorsal view and (D) ventral view. *A. fuscipes* constructs cases from a similar sediment size range as *P. latipennis* but the case design is different. *A. fuscipes* cases weigh less and have a domed shape with a flat base of finer sediments, similar to a turtle shell. Bar indicates 5 mm.

3.2 Methods

3.2.1 Selection of species

Three caddisfly species building mobile cases were selected based on their case design (Figure 3.1), and potential zoogeomorphic importance (Chapter 2). Sericostomatidae (*Sericostoma personatum*) cases are tubular in shape, narrowing towards the posterior and slightly curved (Figure 3.1A). *S. personatum* cases are constructed from fine sand ($D_{50} = 0.27$ mm, Table 2.1). Whilst Sericostomatidae contributed a small amount to total sediment use in Chapter 2, they used a substantial proportion of fine sand and also have zoogeomorphic implications; Sericostomatidae burrow diurnally (Wagner, 1991, 1990) and bioturbate fine sediment (De Nadaï-Monoury *et al.*, 2013).

Limnephilidae probably exhibit the greatest diversity in case design of all caddisfly families (Wiggins, 2004). The Limnephilid *P. latipennis* (Figure 3.1B) builds large cases from coarse sand $(D_{50} = 1.31 \text{ mm}; \text{Table 2.1})$. *P. latipennis* construct cases from both mineral and organic sediment

(Table 2.2), but in this experiment their cases were exclusively mineral. Limnephilidae contributed 11% to total sediment use in Wood Brook (Figure 2.3) and are highly mobile so may have zoogeomorphic effects by transporting sediment horizontally (Figure 1.4C).

Glossosomatidae *Agapetus fuscipes* were responsible for 64% of sediment mass used by caddisfly larvae in Wood Brook (Figure 2.3) and primarily used coarse sand ($D_{50} = 0.94$ mm; Table 2.1). Glossosomatidae larvae usually occupy locations with high flow velocity (Morris and Hondzo, 2013b) and therefore, the entrainment threshold of their cases is geomorphologically important. The shape of *A. fuscipes* cases differ from the other two species being domed, with a flat base, rather than tubular (Figure 3.1C).

S. personatum and *P. latipennis* were collected from Black Brook (a second order gravel-bed stream; 52°46'33.4"N 1°17'57.6"W) on 17th June 2019 and preserved in 70% industrial methylated spirit (IMS). Glossosomatidae are very difficult to remove from their fragile cases when preserved, so *A. fuscipes* larvae were collected more locally from a similar stream on 18th August 2019 (Burleigh Brook, 52°45'47.5" N 1°14'32.8" W), and whilst alive, were gently removed from their cases with tweezers. *S. personatum* and *P. latipennis* larvae were removed from their cases after preservation. Cases were then soaked in deionised water to ensure that no IMS remained and to remove any air trapped within the case. Entrainment experiments were conducted between 20th August and 20th September 2019.

3.2.2 Flume setup

Experiments were conducted in a 10 m long by 0.3 m wide Armfield S6 flume, with parallel glass sides (Figure 3.2A). A raised bed of fixed gravel ($D_{50} = 15$ mm) was constructed at 0.08 m height along the entire flume length. Large cobbles at the flume inlet broke up initial flow structures and ensured well-mixed and fully turbulent flow conditions. Experiments were conducted 7 m from the flume inflow. A flat circular platform (0.065 m in diameter) was fixed so that its surface was 0.1 m from the flume base (thus approximately level with the surfaces of D_{90} particles on the bed; Figure 3.2B & Figure 3.3A). The flat platform standardised variation in hiding or protrusion of cases and sediment (compared to a gravel-bed) and improved observations of entrainment. A sediment trap was located 0.6 m downstream of this platform (depth 0.065 m, length 0.18 m) to collect entrained material (Figure 3.2B). Caddisfly cases and sediment were placed individually on the platform and hydraulic force gradually increased until entrainment occurred.



Figure 3.2. The laboratory flume setup. (A) Water recirculates over a raised gravel-bed. Entrainment was measured from a flat platform located 7 m down the flume. A sediment trap was positioned 0.6 m downstream of this platform. Discharge (pump speed), tail gate and slope were all adjusted to produce a series of 11, reproducible, flow stages with constant depth and increasing velocity and bed shear stress. (B) plan view of the measurement platform, fixed gravel bed and sediment trap.

Eleven flow stages with increasing flow velocity and shear stress were determined *a-priori* (Table 3.1). Pump discharge was increased between each flow stage, and the tail gate and slope were adjusted to maintain a constant depth of 0.08 m and uniform flow over the test area. A side facing Nortek 10 MHz acoustic Doppler velocimeter (ADV) was used to simultaneously measure the three orthogonal components of water velocity in the centre of the experimental platform (X, Figure 3.2B). These velocity measurements were used to determine near bed velocity and to calculate bed shear stress for each flow stage. Velocity measurements were recorded for 120 s at a sampling frequency of 50 Hz. The cylindrical volume over which velocity was measured between 7-13 mm above the bed ; Nortek, 2009). The horizontal length of the measured area (measurement length) was adjusted to increase data quality but for most runs was 7 mm. Six replicates of each flow stage (two measurements taken before, two during and two after the experimental period) were conducted for calculation of bed shear stress. This provided

an estimate of the variability in flow conditions within and between flow stages (Table 3.1). An additional velocity measurement taken at 60% depth was used to estimate depth averaged velocity (n = 3).

Table 3.1. Flume and hydraulic measurements for each flow stage (1-11). Discharge, velocity and bed shear stress increased with each run, whilst depth remained constant. Depth average velocity (U) was based on a single point taken at 60% depth (independent of TKE estimates) and was not collected for the last stage due to an oversight. Near bed U was centred at 10 mm above the bed. For the Reynolds calculations ⁺ indicates flow is transitional according to 500<Re<2500. All other runs were fully turbulent. Froude numbers reveal that all runs were subcritical (Fr<1). Mean bed shear stress (τ_b) was calculated using the turbulent kinetic energy approach.

Flow	Bed slope		Discharge	Depth Average	Near bed U			Mean $ au_b$
stage	(%)	Depth (m)	(m³s⁻¹)	U (ms⁻¹)	(ms⁻¹)	Reynolds	Froude	(N m ⁻²)
1	0.02	0.081	0.0005	0.02	0.02	1152 +	0.025	0.0030
2	0.02	0.081	0.0010	0.04	0.03	2199+	0.047	0.0114
3	0.02	0.081	0.0018	0.08	0.06	3976	0.085	0.0412
4	0.02	0.079	0.0025	0.11	0.08	5532	0.121	0.0668
5	0.02	0.082	0.0032	0.13	0.10	6918	0.147	0.1125
6	0.02	0.080	0.0042	0.18	0.13	9091	0.197	0.1599
7	0.02	0.082	0.0051	0.21	0.15	10900	0.229	0.2287
8	0.03	0.081	0.0060	0.27	0.18	12955	0.276	0.2830
9	0.05	0.082	0.0070	0.29	0.19	15118	0.320	0.3432
10	0.06	0.081	0.0085	0.35	0.24	18351	0.393	0.4907
11	0.08	0.080	N/A	N/A	0.28	N/A	N/A	0.7277

3.2.3 Flume procedure

A single case was added to the centre of the measurement platform (Figure 3.2B) during flow stage 1 (mean water velocity = 0.02 ms^{-1} , Table 3.1). Cases were orientated facing upstream as case positioning is important to entrainment (Waringer, 1993; Figure 3.3A). Each flow stage was maintained for 270 seconds, at which point the discharge was increased and the tail gate and slope adjusted. This changeover process was completed in 30 seconds so that each flow stage took approximately five minutes. A GoPro camera positioned above the measurement platform was used to observe entrainment.

After 8 replicates had been completed for each caddisfly species, cases were broken down into their constituent sediment grains (hereafter *sediment*). Cases were dried at 70°C for 3 hours and weighed. Subsequently, 30% hydrogen peroxide (H₂O₂) was added at 80°C and cases were stirred to facilitate the breakdown of silk (following Chapter 2). The remaining case sediment was

washed through a 0.063 mm sieve and dried. Subsequently, the sediment for each case was individually sieved into half phi intervals. Despite the low mass of sediment, small diameter sieves (38 mm diameter) resulted in a mean mass loss of only 0.7% and max 2.4% (0.0006 g) per case.

The flume experimental procedure was then repeated with the loose case sediment. After soaking in deionised water, loose sediment was added to the flume through a small funnel, which prevented entrainment before the experiment began. Sediment was spread over an approximately circular patch, 1 grain thick in the centre of the measurement platform (Figure 3.3B).



Figure 3.3. Entrainment of cases and sediment from measurement platform. (A) *Potamophylax latipennis* case position during the first flow stage. Case were orientated parallel to the flow direction and facing upstream. (B) *P. latipennis* sediment during the first flow stage. Sediment was spread over a wider area, one grain thick. For both cases and sediment, transport out of an area 36 mm² (red box) was classified as entrainment. (C) Image analysis was used to remove subjectivity from the classification of entrainment thresholds for sediment runs. The surface area of particles entrained during each flow stage was calculated as a percentage of the initial area of sediment.

3.3 Data Analysis

3.3.1 Hydraulic stages

Velocity time series data were post-processed using phase-space thresholding to remove spikes, which were replaced by linear interpolation (Biron *et al.*, 2004; Goring and Nikora, 2002) using the Velocity Signal Analyser tool for Matlab (Jesson *et al.*, 2015). Velocity measurements had to be taken close to the bed, to estimate hydraulics close to those experienced by caddisfly larvae. Velocity measurement close to the bed using ADVs is difficult due to the increased turbulence, shear and reflection near boundaries (Martin *et al.*, 2002; McLelland and Nicholas, 2000; Goring and Nikora, 2002; Voulgaris and Trowbridge, 1998). Data with correlation less than 60% and signal to noise ratio (SNR) less than 10 were removed. Whilst these thresholds are lower than recommended for turbulence calculations (recommended correlation = 70%, SNR = 15; Nortek, 2009), they are an acceptable compromise when working close to boundaries (McLelland and Nicholas, 2000). Any velocity time series with more than 20% data removed from any axis was discarded. This left 5-6 replicates of 2 minute, 50Hz, time series velocity data for each flow stage.

Bed shear stress was calculated via the turbulent kinetic energy (TKE) approach. TKE was calculated from the intensity of the velocity fluctuations in three dimensions (the second order moment statistics; Equation 1 and Equation 2). This method requires turbulence data from a single point close to the bed (e.g. 0.1 of flow depth; Biron *et al.*, 2004) and has been consistently found to give a good estimate of bed shear stress (Kim *et al.*, 2000). Although more susceptible to errors in measurement than methods relying on mean velocities (e.g. Law of the Wall; Karman, 1931) the TKE method has been found to be less susceptible to error in vertical elevation (Kim *et al.*, 2000) and sensor orientation (Biron *et al.*, 2004). TKE (*E*) was calculated according to Equation 1.

$$E = \frac{\rho(u'^2 + v'^2 + w'^2)}{2} \tag{1}$$

Where ρ was water density = 1000 kg m⁻³ and u'^2 , v'^2 and w'^2 were the second order moment statistics for velocity in the long-stream, cross-stream and vertical directions, respectively.

Subsequently, bed shear stress τ_b (N m⁻²) was estimated by multiplying TKE (*E*) by a constant $C_1 = 0.19$ (Soulsby, 1983; Stapleton and Huntley, 1995), Equation 2.

$$\tau_b = C_1 E \tag{2}$$

 τ_b calculated via the TKE approach increased exponentially with flow stage. A Tukey Honest Significance Difference test indicated that the bed shear stress of the four lowest flow stages were not significantly different from their neighbours, but later stages were different (adjusted *P* < 0.05, Figure 3.4A). All flow stages were significantly different for mean streamwise bed velocity (Figure 3.4B). For further analysis the mean τ_b and mean near bed velocity (u) for each flow stage were extracted (Table 3.1). Statistics were conducted in the *stats* package for R Studio (version 3.5.1, R Core Team, 2018).



Figure 3.4. Hydraulic conditions during successional flow stages 1-11. (A) Bed shear stress replicates at each flow stage calculated using the turbulent kinetic energy approach. Demonstrates exponentially increasing bed shear stress with flow stage. (B) Mean near bed velocity. Boxes show the median and interquartile range, whiskers show the interquartile range multiplied by 1.5 and the mean is indicated by ×. * indicates that the stage was significantly different from both neighbouring flow stages according to a Tukey Honest Significant Difference test (adjusted P<0.05).

3.3.2 Entrainment

Establishing entrainment thresholds is subjective, resulting in difficulty comparing between studies (Perret *et al.*, 2015; Garcia, 2008). To remove some of this subjectivity, cases and sediment were required to move > 18 mm (approximately the longest case length of the three species; *P. latipennis*) to classify as entrainment. For loose sediment this was complicated by many grains moving at different entrainment thresholds. Photographs were used to measure the proportion of sediment which was entrainment under each flow stage. Photographs were taken 240 seconds into each flow stage (immediately prior to adjusting flume setup for the next stage). Photos were later analysed in ImageJ (Figure 3B & 3C; Abramoff *et al.*, 2004), following this procedure:

- 1) The image was scaled and then cropped to a 36 mm diameter square centred on the sediment (18 mm x 2; Figure 3.3B).
- The image was thresholded to identify sediment particles from the white background (Figure 3.3C).
- The image was edited to remove the scale bar and then the surface area of particles remaining was calculated.

These steps were repeated for photographs of each flow stage. The surface area of sediment was used as an estimate of sediment remaining. The percentage difference between the area of sediment in the initial image (flow stage 1) and each subsequent flow stage was calculated (Figure 3.3C). This gave a quantitative estimate of sediment entrained during each flow stage.

For each experimental run the initiation of movement and cumulative sediment entrained during each flow stage were determined. To compare sediment entrainment thresholds to those of caddisfly cases, the flow stage during which 90% sediment area had been entrained was used. Because a case moves as a single particle, case movement is equivalent to 100% of sediment entrainment. However, the 90% threshold allows a few particles to remain and therefore provides a conservative measure of entrainment for sediment.

Entrainment threshold data for cases and sediment of each species was mostly non-normal (Shapiro-Wilk P<0.05 for all except A. *fuscipes* cases). Therefore, non-parametric statistics were used to test each of the research questions. Question 1 examined whether a significant difference existed between the entrainment thresholds of cases and their constituent sediment. These data were paired because the same material was used as a case and as disaggregated sediment. Therefore, Wilcoxon signed-rank tests were used with a Bonferroni adjustment. To determine if a significant difference existed between species, for either case or sediment entrainment,

57

(Question 2) a Kruskal-Wallis test was conducted followed by post-hoc Mann-Whitney (Wilcoxon rank sum) tests.

3.3.3 Caddisfly case design

To assess the role of case shape on entrainment between species with different case designs it was important to measure the case dimensions. An image taken prior to flume experiments was analysed in ImageJ to measure the length and width of each case used in the flume experiments (a and b axis respectively). After the flume experiment, it became evident that case volume was important. Therefore, a separate sample of 8 cases of *P. latipennis* and *S. personatum* were measured for all three axes, using electronic callipers (checked with a microscope eyepiece graticule and stage micrometer). Measured caddisfly were collected from the same site on the same date as those used in the flume experiments. *A. fuscipes* cases were extremely variable in shape, and therefore 16 cases were measured (Figure 3.5). A linear association between the b and c axis of each species was then used to estimate the c axis for the cases used in the flume experiment (R² values for *S. personatum* = 0.99, *P. latipennis* = 0.64, *A. fuscipes* = 0.15; Figure 3.5). The dimensions of cases of all three species were very different, so this method was a sufficiently accurate estimate of case-volume to differentiate between species.

To compare differences in shape between the cases of each species, Corey shape factor (Corey, 1949) was determined (Equation 3).

$$SF = \frac{c}{\sqrt{ab}} \tag{3}$$

Where a, b and c (mm) were the major, intermediate, and minor axis of the case, respectively. Shields criterion (Shields, 1936) is widely used to nondimensionalise entrainment thresholds (Yang *et al.*, 2019). Shields parameter (τ_{*c}) was calculated according to Equation 4, for cases and sediment of each species.

$$\tau_{*c} = \frac{\tau_c}{(\rho s - \rho)gD} \tag{4}$$

Where τ_c was bed shear stress (τ_b) during the flow stage at which the case or sediment was entrained (N m⁻²), ρ was water density = 1000 kg m⁻³, ρ s was sediment density = 2650 kg m⁻³ for both cases and sediment, g was acceleration due to gravity = 9.81 m s⁻² and D was particle diameter (D₅₀ for sediment or b axis for cases, m). This was plotted against grain Reynolds

number (Re_* Equation 5, dimensionless) to compare case and sediment entrainment thresholds to the curve developed by Shields (1936).

$$Re_* = \frac{u_* \mathrm{D}}{\mathrm{v}} \tag{5}$$

Where u_* was the shear velocity (Equation 6), D particle diameter (m), and v, kinematic viscosity = $1 \times 10^{-6} \text{ m}^2 \text{ s}^{-1}$.

$$u_* = \frac{\tau_c}{\rho} \tag{6}$$

Shields dimensionless criteria is widely used in studies of particle entrainment (Yang *et al.*, 2019). However, it is less appropriate for caddisfly cases because it uses particle size (D) multiplied by density as a proxy for particle mass (because sediment size is more commonly measured than mass in rivers; Wolman, 1954). This is an issue for caddisfly cases because their size to mass ratios are not consistent between species (compare *S. personatum* large but lightweight cases to *A. fuscipes* cases, which are small but more dense; Figure 3.1; Table 2.1). Furthermore, caddisfly cases are hollow, so that if density were calculated from their external dimensions and mass, without allowing for their hollow centre, their density is less than that of water and they should float. For this reason, case density was taken to be the same as sediment density (2650 kg m⁻³ the density of quartz sand). For particle diameter (D in Equation 4) case width (b axis) was used for cases and mean D_{50} for sediment.

As an alternative standardisation method, better suited to hollow particles, cases were standardised by mass; τ_c values were divided by mass (τ_c/m). If differences in entrainment thresholds between cases still exist, they can therefore be attributed to case shape (Question 2). As for entrainment threshold data (τ_c) Kruskal Wallis followed by Wilcoxon rank sum tests were then conducted for differences between species (for cases and sediment) for $\frac{\tau_c}{m}$ and τ_{*c} and Wilcoxon signed rank for paired case versus sediment data.



Figure 3.5. Association between width (b axis) and depth (c axis) of cases of each species. A linear trend between b and c was used to estimate the unknown c axis of caddisfly used in the flume experiments. The trend is weaker for *P. latipennis* and particularly *A. fuscipes* due to variability in case design between species. However, it provides a sufficiently accurate method to calculate case volume for the analysis in this chapter because it distinguishes between species.

3.4 Results

3.4.1 Entrainment of cases versus constituent sediment

The construction of cases resulted in composite particles which were much larger than the constituent sediment (Table 3.2). *S. personatum* used the finest sediment (mean $D_{50} = 0.27$ mm), converting this to a case of mean width (B axis) of 3.07 mm and length 14.57 mm. *P. latipennis* and *A. fuscipes* used larger coarse sand particles with similar size distributions (mean width = 1.16 and 0.98 mm respectively; Figure 3.6) but *P. latipennis* cases were considerably larger and heavier than both of the other species (Table 3.2).

Comparing entrainment of cases and sediment (Question 1) both *S. personatum* and *P. latipennis* cases moved at significantly lower critical shear stress thresholds than their loose case sediment (Figure 3.7A, Table 3.3). All cases of *S. personatum* or *P. latipennis* were entrained below the shear stress required to entrain their respective sediment grains (Figure 3.8). *A. fuscipes* cases however, moved over a similar range of τ_c as their constituent sediment (Figure 3.7A) and critical entrainment was not significantly different between cases and constituent grains (Table 3.3B). For *A. fuscipes*, the range of case movement overlaps that of sediment, although a few sediment particles moved at lower and higher shear stress than cases (Figure 3.8C). Shields criterion indicates that case critical entrainment was much lower than would be expected from particle size for all species (Figure 3.10).

Table 3.2. Size and shape characteristics for cases and sediment. Mean with standard deviation in brackets. a,
b, and c indicate major, intermediate and minor particle axes respectively. D ₅₀ reported as b axis for loose
sediment. *Minor (c axis) were not measured on these specimens but 8 cases (16 for A. fuscipes), collected
from the same sites on the same date were used to construct a regression model between b and c axis to
predict the c axis of these individuals (Figure 3.5). SF is the dimensionless Corey shape factor.

		a (mm)	b or D ₅₀ (mm)	c* (mm)	SF	Volume (mm ³)	Mass (g)
Case	S. personatum	14.57 (0.66)	3.07 (0.11)	3.01 (0.11)	0.45 (0.01)	134.57 (12.13)	0.03 (0.004)
	P. latipennis.	21.75 (0.86)	5.73 (0.42)	4.84 (0.37)	0.43 (0.01)	608.11 (110.33)	0.26 (0.04)
	A. fuscipes	6.01 (0.32)	3.63 (0.59)	2.34 (0.11)	0.51 (0.03)	51.70 (12.79)	0.03 (0.01)
Sedimen	^t S. personatum		0.27 (0.02)				0.03 (0.004)
	P. latipennis.		1.16 (0.15)				0.25 (0.04)
	A. fuscipes		0.98 (0.2)				0.03 (0.006)



Figure 3.6. Cumulative particle size distribution for case sediment of each of the three species. Mean indicated by line with markers. Particle size indicates passing sieve.

Table 3.3. Results of statistical tests performed on (A) Difference between entrainment of each species, for both cases and loose sediment and (B) Difference between case and sediment. Chi-squared values reported for Kruskal Wallis and W and V statistics for Wilcoxon rank and signed rank respectively. Significance is indicated by * 0.05 > P > 0.01; ** 0.01 > P > 0.001; *** P < 0.001. Each test was done for critical bed shear stress (τ_c), bed shear stress divided by mass ($\frac{\tau_c}{m}$) and shields dimensionless shear stress (τ_c).

A. Species								
	Case or sediment	Kruckal –	Wilcoxon rank (w)					
Parameter		Wallis (K)	A. fuscipes - S. personatum	S. personatum – P. latipennis	P. latipennis – A. fuscipes			
τ _c (N m ⁻²)	Case	18.8***	0***	0***	8*			
Tc	Sediment	16.8***	43	4**	64***			
$\frac{c_c}{m}$ (N m ⁻² kg ⁻¹)	Case	20.5***	0***	0***	64***			
	Sediment	16.6***	46	64***	0***			
τ _{*c}	Case	19.3***	0***	4**	0***			
	Sediment	17.5***	64***	64***	53*			
B. Sediment v	ersus case							
		Parameter	V	/ilcoxon signed rank (v)			
			S. personatum	P. latipennis	A. fuscipes			
		τ _c (N m⁻²)	0*	0*	3			
		$\frac{\tau_c}{m}$ (N m ⁻² kg ⁻¹)	0*	0**	3*			
		τ _{*c}	0**	0**	0**			

3.4.2 Differences in case entrainment between species

The shear stress required to entrain cases was significantly different between each species (Table 3.3A, Figure 3.7A; Question 2). *S. personatum* moved under the lowest shear stress (mean $\tau_c = 0.06$ N m⁻²). *P. latipennis* required 0.18 N m⁻² to move and *A. fuscipes* required the greatest shear stress to entrain (mean $\tau_c = 0.28$ N m⁻²; Figure 3.7A). Qualitative observation of the method of movement for cases indicated variability between species. As shear stress increased, tubular cases (*S. personatum* and *P. latipennis*), 1) rocked in situ, 2) spun so that they were transverse to the flow and, 3) rolled off the measurement area. In contrast, *A. fuscipes* initially moved by sliding followed by either continued sliding or, occasionally, by rolling.

3.4.3 Differences in constituent sediment entrainment between species

Unlike cases which moved as one particle, loose sediment particles moved over a much wider range of τ_b (Figure 3.8). For example, transport of *S. personatum* sediment was initiated as early as 0.11 N m⁻² but reached the 90% threshold at 0.34 N m⁻² (Figure 3.8A). *P. latipennis* sediment required significantly greater shear stresses to entrain than that of the sediment associated with other two species, which were not significantly different from one another (Table 3.3; Figure 3.7A).

Observing the method of movement for loose sediment runs was considerably harder than for cases and many particles were too small to identify. Larger particles of sediment of all species moved by a mixture of rolling and sliding (dependant on the sphericity of the particle). Loose sediment also showed signs of grain interactions, with sediment patches moving together and areas of smaller grains building up behind larger grains. This was particularly true for *A. fuscipes* and *P. latipennis* which had a greater range of particle sizes than *S. personatum* (Figure 3.6).


Figure 3.7. Parameters for the entrainment of cases and 90% of constituent sediment for each species. (A) Critical bed shear stress (τ_c). (B) Critical bed shear stress divided by mass, τ_c/m . (C) Shields dimensionless critical shear stress, τ_{*c} . Boxes show the median and interquartile range, whiskers show the interquartile range multiplied by 1.5 and the mean is indicated by ×.



Figure 3.8. Percent of sediment entrained by the end of each flow stage for cases (solid line) and sediment (dashed line) of each species (A-C). Percent calculated as the spatial area of grains remaining (following Figure 3.3C).

3.4.4 The role of case shape for case entrainment

The case designs of each species were very different (Figure 3.1; Table 3.2). *P. latipennis* built cases both substantially heavier and of greater volume than the other two species (Table 3.2). *S. personatum* and *A. fuscipes* built cases of similar mass (both 0.03 g; Table 3.2) but *S. personatum* cases were more than twice as long. Using the Zingg classification indicated a notable difference in shape between cases of *A. fuscipes* and the other species (Figure 3.9). *A. fuscipes* cases had lower c/b and higher b/a ratios and consequently a more spherical shape. Both *S. personatum* and *P. latipennis* have similar b and c axis dimensions which were far exceeded by their a axis, and were both therefore rod shaped (Figure 3.9). Importantly for their entrainment, *S. personatum* cases had a circular cross section while *P. latipennis* cases were roughly oval. *A. fuscipes* cases were elongated hemispheres with a flat base and sharp angles between the base and sides; they were spherical (Figure 3.9) but the edges were not rounded.

After dividing by mass $\binom{\tau_c}{m}$, the entrainment threshold of each species were still significantly different (Table 3.3; Figure 3.7B). This indicates that differences between species are at least partly due to case shape, not solely related to their relative mass. Per unit mass, *P. latipennis* cases were entrained at significantly lower shear stress than *S. personatum* (Figure 3.7B; Table 3.3). This suggests that differences in mass are largely responsible for the significantly higher entrainment threshold of *P. latipennis* than *S. personatum* (Figure 3.7; Table 3.3). After controlling for mass, *A. fuscipes* remains significantly more difficult to entrain than the other species (Figure 3.7B; Table 3.3). This suggests that shape, not mass, was largely responsible for the difference between dome shaped *A. fuscipes* and the tubular cases of the other species. Using Shields dimensionless criterion (τ_{*c}) entrainment thresholds of cases of all species were still significantly different, although dwarfed by the much higher values of τ_{*c} for loose sediment (Figure 3.7C).



Figure 3.9. Zingg classification of particle shape for cases of the three species (Mclean 1982). Distinct differences in shape are evident between *A. fuscipes* cases, which more closely resemble spheres and the rod-shaped cases of *P. latipennis* and *S. personatum*.

3.5 Discussion

This chapter has examined the bed shear stress required to entrain empty cases of three species of caddisfly and the loose sediment the cases were constructed from. Cases and sediment were entrained from a measurement plate (simulating a flat-topped cobble) within a gravel-bed. Consequently, hydraulic conditions were analogous to those of gravel-bed streams, where high densities of these species typically occur (including Black Brook and Burleigh Brook where larvae were collected, and Wood Brook studied in Chapter 2). The experiment successfully identified and replicated a series of flow stages, which provide a stepped increase in near bed velocity and bed shear stress. (Figure 3.4). This enabled precise determination of critical entrainment thresholds, evidenced by the fact that the entrainment thresholds of sediment particles match those predicted by Shields (1936; Figure 3.10).

Cases built by two species; *P. latipennis* and *S. personatum* were entrained at significantly lower shear stresses than their case sediment (Table 3.3). This might be expected from the fact that caddisfly case construction increases the size and therefore protrusion of sediment into flow (Table 3.2). This fits with previous studies on the mobility of other naturally occurring agglomerates, including Polychaete faecal pellets (Taghon *et al.*, 1984) and clay aggregates (Rust and Nanson, 1989). Therefore, caddisfly bioconstructions may increase the potential for sediment transport. In contrast, cases of *A. fuscipes* were entrained at similar shear stresses to their incorporated sediment (Figure 3.8C). This result is particularly interesting and suggests that case shape, at least partly, affects the difference in entrainment between caddisfly species. Whilst the experiment is limited in the generality of the results, which are specific to empty cases of three species, it provides an important step towards understanding the mechanisms by which case-building caddisfly may affect sediment transport in rivers.



Figure 3.10. Shields criteria for cases and sediment for each species. Loose sediment plots approximately on the Shields line (based on the empirical work of Shields (1936) and generally accepted (Paphitis, 2001)), giving confidence in the methodological approach of this chapter. However, cases were moved at τ_{*c} far below expected for the size of these particles. Arrows indicate approximate reduction in Shields criteria and increase in Reynolds which occurs when cases are constructed from loose sediment.

3.5.1 Transport of caddisfly cases versus sediment (Question 1)

Caddisfly of all three species created cases with very different size and shape characteristics to the sediment grains used in construction. All three species incorporated fine sand to fine gravel sized mineral sediment (Figure 3.6), into much larger aggregate particles. *P. latipennis*, for example, built the largest cases which were on average 22 mm long and 6 mm wide from sediment with an average diameter of 1.16 mm (Table 3.2). Hydraulic force increases rapidly with distance above the riverbed and therefore larger particles experience higher drag. For mineral sediment grains, this increased exposure is usually offset by the increase in particle mass, meaning that larger particles require higher shear stress to entrain (Shields, 1936). Caddisfly cases are substantially heavier than the individual sand grains incorporated but are also hollow, and therefore weigh considerably less than a solid mineral particle of the same size. *S. personatum* and *P. latipennis* both constructed large tubular cases which were easier to entrain than their constituent sediment (Table 3.3) suggesting that the increased exposure to hydraulic force due to protrusion, outweighed the increase in mass resulting from the cementing of many small sediment grains together.

A. fuscipes also substantially increased the effective size of sediment during case construction, building a case which was a similar mass to S. personatum (both 0.03 g; Table 3.2) from sediment of a similar size to P. latipennis (Table 3.2). However, entrainment of A. fuscipes cases did not differ from their incorporated sediment (Table 3.3). This was also true after standardising for case mass (Table 3.3; Figure 3.7B), suggesting that case shape is responsible for the difference in entrainment thresholds between A. fuscipes and the tubular case-builders.

Entrainment was measured from a flat platform (Figure 3.2) and therefore the transport of cases and sediment over a rough gravel-bed are unknown. Qualitative observations of case entrainment showed that, once entrained, cases quickly came to rest in the first sheltered pocket between gravel particles. It is probable, however, that due to their size and shape *S. personatum* and *P. latipennis* cases will be transported over gravel beds at lower shear stresses than their constituent sediment. Cases of all three species are larger than their constituent sediment grains and are therefore less likely to find sheltered pockets, resulting in overpassing. Furthermore, rod shaped particles (e.g. *S. personatum* and *P. latipennis* cases) are more mobile than platy or spherical particles, because they can twist around obstacles and are less likely to find pockets in bed topography where they are stable (Demir, 2000). Hollow rods of mineral sediment are rarely found in riverbeds, so these species of caddisfly construct a particle with novel entrainment properties.

3.5.2 Entrainment of loose sediment

Unlike caddisfly cases which moved as one item, loose sediment moved over a wide range of shear stresses (Figure 3.8). The critical bed shear stress required to move 90% of case sediment was similar for *A. fuscipes* and *S. personatum* but higher for *P. latipennis* (Figure 3.7A). This is probably a combination of grain size and the total mass of grains. *P. latipennis* used the largest sediment grains (Figure 3.6) but *A. fuscipes* used only marginally smaller grains (Figure 3.6) and was entrained at significantly lower shear stress (Table 3.3). Therefore, grain size is not the only factor affecting the entrainment of sediment. *P. latipennis* cases were considerably larger than *A. fuscipes*. The greater number of particles probably increased the importance of grain interactions and armouring was evident for *P. latipennis* sediments, with larger particles hiding smaller particles. The entrainment of smaller grains is therefore dependent on the entrainment of larger particles, causing sediment to be transported at a higher shear stress threshold.

A 90% entrainment threshold of loose sediment was used to compare to entrainment of cases (100% of sediment moved at once) because often a small number of particles persisted to much higher shear stresses, presumably largely due to chance (Figure 3.8). 90% did not relate to a step change in sediment movement (as is commonly used in entrainment studies; Buffington and Montgomery, 1998; Simões, 2014; Perret *et al.*, 2015) but is approximately equivalent to 'general movement' (Perret *et al.*, 2015; Petit 1994). As the overall effect of caddisfly on sediment stability was of primary interest, comparing general sediment movement was probably more valuable than incipient motion. In any case, both *P. latipennis* and *S. personatum* cases had been transported before the incipient movement of loose sediment (Figure 3.8A & B).

3.5.3 Movement of cases, role of mass and shape (Question 2)

The difference in entrainment thresholds between cases of different species will be a result of both difference in mass and the size and shape of cases. The difference between *P. latipennis* and *S. personatum* cases appears to be largely related to the difference in case mass (Figure 3.7B). The shape is broadly similar between the two species, both being tubular (Figure 3.1) but they differ considerably in mass (mean *P. latipennis* = 0.26 g, *S. personatum* = 0.03 g; Table 3.2). However, small differences in shape may be important to case entrainment; *S. personatum* cases are narrower and therefore protrude less into the flow, reducing their exposure to turbulent hydraulics. In addition, the slightly curved profile of *S. personatum* cases may reduce rolling as the case has to be lifted (and in loose sediment the ends would dig in). Cases of *P. latipennis* are made from much coarser sediment (Figure 3.6) and therefore have a rougher surface texture (Figure 3.1). Particles

sticking out from the case may increase case protrusion and facilitate entrainment but may also make the case less inclined to roll along the bed.

The shape of *A. fuscipes* cases was very different to cases of *S. personatum and P. latipennis* (Figure 3.1; Figure 3.9), resembling domes over a flat base instead of tubes. Consequently, whilst they had a similar mass to *S. personatum* and were considerably lighter than *P. latipennis*, they were much harder to entrain than either species (Figure 3.7A). The tubular shape and rounded profile of *S. personatum* and *P. latipennis* meant that, while they remained facing upstream until entrainment, once turned sideways they rolled easily off the platform. In contrast, the flat base of *A. fuscipes* probably increased friction, reduced tendency to roll, and therefore required much greater shear stress for entrainment.

These results suggest that different case designs might serve different purposes for the larvae. *S. personatum* burrow diurnally (Wagner, 1987) and therefore may avoid entrainment by burrowing, facilitated by an externally smooth case of small sand grains. Glossosomatidae are rheophilic species, adapted to life on the exposed upper surfaces of gravel beds (Olden *et al.*, 2004). *A. fuscipes* cases may therefore be designed to reduce entrainment. Furthermore, *A. fuscipes* appear to be flexible with the grain size used in case construction (Figure 2.5). If flow resistance is a key objective of their case design, perhaps they are more likely than other species to adapt their case to the hydraulic environment. Whilst cases may not be designed purely as anti-entrainment devices they may still play an important role by; 1) reducing energy required for active resistance and, 2) keeping entrained larvae near the bed surface where they are likely to move only short distances into more sheltered areas, unlike uncased larvae which, once active resistance has failed, are more likely to move in suspension

3.5.4 Geomorphological implications of caddisfly case construction

Therefore, empty cases of *S. personatum* and *P. latipennis* are likely to be preferentially transported over other grains on the river-bed surface. However, this experiment is limited to entrainment of empty cases off a flat platform. Entrainment of cases with live larvae over a mobile gravel-bed are likely to be different. The tightly controlled nature of this experiment prohibited the use of live larvae. For example, the baseboard was fixed and painted white in order to increase sediment contrast for photographing entrainment (Figure 3.2; Figure 3.3). Experimental trials with live larvae under low flows resulted in larvae walking downstream (probably as the quickest way to escape a highly unnatural environment) and in higher flows clinging to the flume edges where velocities were reduced.

The presence of a live larvae in a case considerably increases the force required to move the case. Otto and Johansson (1995) found that cases of *Silo spp*. (Goeridae) moved at 10.3 cm s⁻¹ but with the addition of a live larvae this increased to 64.4 cm s⁻¹. Live larvae are able to actively reduce drift via a number of mechanisms including clinging on, changing case orientation (e.g. facing into flow to reduce exposure), fixing their case or themselves with silk threads and moving to avoid high exposure areas (Olden *et al.*, 2004; Rice *et al.*, 2007; Waringer, 1989). Active resistance to entrainment varies between species; for *A. auricollis* active resistance accounted for 55% of current resistance (Waringer, 1989) compared to 90% for *Potamophylax cingulatus* (Otto, 1976). Larvae may also be able to sense increasing shear stresses and seek refuge pre-emptively to avoid entrainment (Townsend and Hildrew, 1994). Therefore, it is possible that when the larvae is alive the sediment is transported further by walking than by drift (Figure 1.4B). Furthermore, if the addition of a larvae increases the shear force required to move *A. fuscipes* cases, these would become more stable than their constituent sediment.

The results of this experiment apply only to empty cases. This scenario is most likely to occur following pupation of adult caddisfly or when cases are abandoned by the larvae. Pupal cases of Limnephilidae are typically attached to the undersides of larger particles or stable wood material. Sericostomatidae pupal cases have been found buried just below the surface in fine sediment and anchored to larger grains. However, both find their way onto the bed surface and were found scattered across the bed of Black Brook, in various states of decay.

3.6 Summary

Via the construction of cases from mineral and organic sediment caddisfly create composite particles of sediment particles bound together with silk (Hansell, 1968; Okano *et al.*, 2016). Caddisfly created an agglomerate particle with greater mass than any individual incorporated grain, but cases were also much larger and hollow and therefore effectively less dense. The effect of case construction on the entrainment threshold of incorporated sediment varied between the three species. *S. personatum* and *P. latipennis* cases were entrained at significantly lower bed shear stresses than their constituent grains. This effect is also likely to be seen in the movement of cases over gravel-beds as they can more easily pivot and roll over obstacles. In contrast, *A, fuscipes* cases were entrained over the same range of shear stress as their constituent grains. This fits with the rheophilic nature of *A. fuscipes* larvae and their preference for living on exposed surfaces in fast flowing streams.

Both weight and shape were important in explaining the difference in entrainment between case designs. The difference between *S. personatum*, which moved earliest (mean $\tau_c = 0.06 \text{ N m}^{-2}$), and *P. latipennis* (mean $\tau_c = 0.18 \text{ N m}^{-2}$) was largely a result of differences in mass between the two cases. *A. fuscipes* cases were relatively lightweight but significantly more difficult to entrain than other species (mean $\tau_c = 0.28 \text{ N m}^{-2}$). Observations of case movement revealed that once entrainment thresholds were reached, *S. personatum and P. latipennis* turned perpendicular to the flow direction and rolled. However, the flat base and sharp edges of *A. fuscipes* reduced their propensity to roll and most cases slid, increasing friction with the bed and therefore entrainment thresholds.

These are important results for understanding the effect of bioconstruction by caddisfly larvae on the mobility of sediment in rivers. More research is required to determine the role of live larvae in the mobility of its case sediment, via resisting or initiating entrainment as well as transporting sediment while crawling. All three species studied in this chapter occur in large numbers (particularly *A. fuscipes*) and use substantial quantities of sediment (Chapter 2). Consequently, increased mobility of these sediments could affect the distribution and transport of sediment in rivers.



Glossosomatidae Agapetus fuscipes in an experimental flume (Chapter 4)

Chapter 4. Vertical sand displacement by Glossosomatidae Agapetus fuscipes larvae

4.1 Introduction

4.1.1 Sand transport in armoured gravel-bed rivers

Substrate surfaces are important zones for sediment transport in rivers. Gravel-bed rivers typically develop armoured surface layers, in which coarser grains are overrepresented compared to the subsurface (Dietrich *et al.*, 1989; Pitlick *et al.*, 2008). Entrainment is size selective, so fine sediment is preferentially winnowed from gravel-beds (Shields, 1936). Fine sediment also infiltrates into the interstices between the gravel framework, resulting in a downwards movement of particles; a process which is accelerated when gravel particles are mobilised and voids become available (Mao *et al.*, 2011). The fine sediment content of gravel-beds is therefore hidden below the surface in interstices and in the lee of obstacles, where it is sheltered from hydrodynamic forces at the surface (Laronne *et al.*, 2001).

The development of armoured substrate surfaces means that only high discharge events can break up and mobilise the coarse surface layer of gravel-beds. Consequently, under most flow conditions bedload transport is dominated by finer grained sediments including sand (Church, 2010). However, although it may be hidden, fine sediment plays important roles in gravel-bed river geomorphology. The sand fraction in gravel-bed rivers can also promote or inhibit the transport of larger gravel clasts, depending on fine sediment proportion (Grams *et al.*, 2007; Venditti *et al.*, 2010; Wilcock *et al.*, 2001). Therefore, the quantity and distribution of finesediment within gravel-beds is geomorphologically important. The transport of fine sediment in armoured gravel-bed rivers is limited by supply of this sediment on the surface and, the steep gradient in hydraulic exposure between the surface and subsurface of armoured gravel-beds mean that displacement of fine sediment over small vertical distances can have substantial effects on the exposure of that sediment and its contribution to bedload transport (Garcia *et al.*, 1999; Powell, 1998).

Caddisfly of the family Glossosomatidae are characteristic of gravel surfaces in fast flowing gravel-bed streams and rivers (Figure 4.1A; Houghton and Stewart, 1998; Morris and Hondzo, 2013b; Nijboer, 2004; Wood and Armitage, 1999). Observations of Glossosomatidae species indicate a preference for positioning themselves on the exposed upper faces of gravel-bed

particles (Figure 4.1B & C; Kovalak, 1976; Olden *et al.*, 2004). These zones of high hydraulic shear and exposure to predators are avoided by many aquatic macroinvertebrates, due to the energetic cost of maintaining position in turbulent conditions (Statzner *et al.*, 1988; Statzner and Higler, 1986). However, Glossosomatidae larvae are well adapted to high velocity locations due to their case mass (relative to body weight; Becker, 2001; Waringer, 1989), case shape (Figure 3.7), and their use of silk to fix their case in position on gravel surfaces (Olden *et al.*, 2004).

Respiration is an important factor structuring the distribution of Glossosomatidae larvae in gravel-bed streams (Kovalak, 1976). Unlike many caddisfly taxa, Glossosomatidae rely on flow velocity to flush oxygenated water through gaps in the dome of their case (Zwick *et al.*, 2011). Glossosomatidae cases act to reduce predation and allow them to better exploit periphyton on exposed gravel surfaces than 'unarmoured' grazers (Kohler and McPeek, 1989).



~1 m

Figure 4.1. Glossosomatidae larvae in the River Dove, Derbyshire, U.K. (A) Sketch of *Agapetus fuscipes* case design, note the use of a relatively broad range of sand sizes, often with larger particles on the case sides and the ring of fine sediment around the case opening, characteristic of *Agapetus* species (Wallace *et al.*, 2003). (B) Dense aggregations on the exposed face of a single cobble, (C) Cases visible on the upper surface of gravel particles on the riverbed.

Glossosomatidae larvae therefore exhibit 'conflicting resource requirements' (*sensu*. Statzner, 2011; Statzner *et al.*, 2005), requiring both fine sediments for case construction (Becker, 2001; Marchant, 1988) and exposed gravel surfaces for food and respiration (Kovalak, 1976). This suggests that Glossosomatidae larvae construct cases from fine sediment available in sheltered areas (interstices and sheltered patches), and then move with their case onto the gravel surface. This flux of sediment could be geomorphologically important because;

- By transporting their case sediment from sheltered patches and interstices to the surface of gravel particles, Glossosomatidae potentially increase the exposure of fine sediment to entraining flows. Where present, Glossosomatidae cases are usually conspicuous on the surface of gravel-beds (Wiggins, 2004) and are often the only fine sediment visible in these areas of high hydraulic stress (Figure 4.1C).
- 2) Similar movements are repeated by many individuals. Glossosomatidae are a widespread caddisfly family and under the appropriate environmental conditions, can reach extremely high abundances (Hickin, 1967; McNeely and Power, 2007). In Wood Brook, Glossosomatidae *Agapetus fuscipes* larvae were found at abundances up to 6710 larvae m⁻², making it by far the dominant caddisfly taxa (Chapter 2). As a result, Glossosomatidae accounted for 64% of total sediment mass used by caddisfly (Figure 2.3). At some sites, this could represent a substantial flux of sediment in a single direction; onto the riverbed surface.
- 3) Glossosomatidae larvae use sand, with a D₅₀ of 0.94 mm (Table 2.1). Sand is often the most mobile sediment fraction, accounting for a substantial proportion of bedload in many rivers (Church, 2010). The transport of sand out of interstices may also have implications for gravel transport (Wilcock and Kenworthy, 2002).

The potential displacement of sediment by Glossosomatidae larvae is likely to vary based on larvae behaviour and the abiotic conditions, particularly the hydraulic and sediment characteristics of the environment. Whilst Glossosomatidae have a wide range of flow tolerances, their behaviour under high flow velocity is important because larvae may move their case (and sediment) back to sheltered areas, or the larvae may abandon the case and drift. Glossosomatidae larvae have been found to migrate to less exposed locations on gravel particles during floods, resulting in reduced entrainment (Brooks, 1998 In: Lake, 2000). However, other studies have documented Glossosomatidae species drifting under hydraulic stress (Houghton and Stewart, 1998), potentially leaving their case exposed and easily transported downstream.

77

The nature of the gravel-bed is also likely to affect the movement of Glossosomatidae larvae. In particular, the size and protrusion of individual surface particles into the flow, as well as having geomorphological implications (Masteller and Finnegan, 2017), will dictate the space and hydraulic landscape available to larvae. Gravel-beds with higher grain protrusion will provide more opportunity for larvae to find sites suitable for feeding and respiration but will also increase the separation distance between exposed particle surfaces and available fine sediment required for case construction and modification, potentially forcing larvae to migrate further.

4.1.2 Context of this chapter

Transport of sediment at moderate flows is dictated by the availability and character of fine sediment on the bed surface, the distribution of which is of interest to fluvial geomorphologists (Curran and Waters, 2014; Laronne *et al.*, 2001; Wilcock and Kenworthy, 2002). However, the role of biology in modifying sediment sorting processes has received far less research attention (Rice *et al.*, 2012; Statzner, 2012). Many species of case building caddisfly are very mobile, transporting their case sediments several metres per day (Rice *et al.*, 2007; Elliott, 1971; Figure 1.4B & C) but active transport of sediment by caddisfly larvae has not been studied. Glossosomatidae caddisfly have the potential to affect the distribution of fine sediment by transporting sand from sheltered locations to the surface of gravel-beds. However, the movement of Glossosomatidae larvae has only been studied from a habitat preference perspective (Kovalak, 1976; Morris *et al.*, 2011; Morris and Hondzo, 2013b). This is the first attempt to understand movement of fine sediment by a caddisfly larva and the implications for zoogeomorphology. This chapter uses a laboratory flume to compare Glossosomatidae *A. fuscipes* movement behaviour under two flow velocities and two bed roughness scenarios. This chapter addresses thesis objective 3. Specifically, the following research questions were investigated:

- 1) To establish if *A. fuscipes* larvae transport sediment vertically upwards, from interstices onto the surfaces of gravel particles and to quantify any flux of sediment.
- 2) To determine if the vertical migration of *A. fuscipes* larvae is influenced by flow velocity or the grain protrusion of the gravel-bed.

4.2 Methods

4.2.1 Flume setup

Experiments were conducted in an Armfield S6 Flume, with a working section 10 m long and 0.3 m wide (Figure 4.2A, as for Chapter 3). A bed of gravel particles $(16 - 64 \text{ mm diameter}, D_{50} = 38 \text{ mm})$ cemented to a plywood base ensured fully turbulent flow with bed roughness and turbulence comparable to the stream from which caddisfly larvae were sourced. The bed was raised from the flume base by 0.08 m to allow a tray filled with loose sediment (length = 0.21 m, width = 0.14 m, area = 0.0294 m⁻²; Figure 4.2B) to be inserted flush with the bed. The sediment tray was located 7 m from the flume inflow and 3 m from the tailgate. A net (1 mm aperture) was placed 0.7 m downstream of the tray to catch drifting larvae. Four treatments were studied with varying flow velocity and gravel protrusion. Two velocities, low (mean velocity = 0.14 m s⁻¹) and high (mean velocity = 0.6 m s⁻¹) were used, whilst gravels were selected to be either large (high protrusion; b axis = 45 - 65 mm) or small (low protrusion; b axis = 32 - 45 mm). Gravel sizes and velocities chosen were representative of the range of conditions in the shallow gravel-bed stream from which larvae were collected and the upper velocity was the maximum achievable in the flume whilst maintaining stable flow conditions.

Velocity was measured using an electromagnetic current meter located 1 m upstream of the sediment tray at 0.05 m elevation from the bed (Figure 4.2). Velocity was manipulated by adjusting the discharge, flume slope and outflow weir to maintain a water depth of 0.1 m. Oxygen availability was measured at the start and end of each run and remained above 99% saturation. Water temperature was controlled between $13 - 15.6^{\circ}$ C and light was provided over the focus area using a Fluval aquasky LED between 04:30 and 22:00 to mimic the natural daylight cycle in June - July.

A. fuscipes larvae were sourced from a local stream; Black Brook (52°46'33.4"N 1°17'57.6"W). Larvae were collected the evening before experiments and then acclimatised overnight in aerated aquaria with physiochemical conditions similar to those in the flume. Experiments were conducted between mid-June and mid-July 2018.



Figure 4.2. The laboratory flume setup. (A) Experiments were conducted in the same flume as Chapter 3. An uneven fixed gravel bed ($D_{50} = 38$ mm) was placed throughout the flume length. (B) This was raised to allow a tray (length = 0.21, width = 0.14 m) to be inserted flush with the bed. An aluminium frame could be placed over the tray area facilitating accurate measurement of *A. fuscipes* locations but was removed during the flume runs.

4.2.2 Sediment tray setup

The sediment tray (Figure 4.3) was first filled with gravel and sand, collected from local streams and then dried at 130°C, until level with the tray rim (Figure 4.3A). Sand was sieved to be between 0.355 - 2.8 mm to include the D₅-D₉₅ range used by *A. fuscipes* locally (Table 2.1), with a size distribution shown in Figure 4.3C. In addition, fresh gravel particles were collected the previous day from the surface of a local stream and added to the tray surface to simulate an armoured gravel-bed (Figure 4.3A). This gravel was placed in the sediment tray with algae covered faces uppermost (as they were positioned in the river).

To ensure differences in gravel size resulted in differing protrusions between the treatments, only gravel that was not platy, elongated or bladed was selected, with a deviation from compactness value (S = c/a) greater than 0.3 (where c and a are the minor and major particle axis respectively, Bunte and Abt, 2001). Analysis of variance followed by post-hoc Tukey Honest

Significant Difference tests conducted in R Studio using the *stats* package (version 3.5.1, R Core Team, 2018), demonstrated significant differences (P < 0.05) in particle c axis between small and large gravel treatments (Figure 4.3D), but not between runs of the same gravel treatment. As particles were placed in stable positions, with c axis orientated vertically, this indicates that the larger gravel size also resulted in a difference in bed protrusion.

Flow velocity depth profiles were collected for an example of each treatment to characterise in more detail the velocity larvae were exposed to and investigate interactive effects between gravel size and flow velocity (e.g. does larger gravel also increase flow velocity by forcing flow to accelerate over the bed surface?). Velocity profiles were measured using a Nixon low speed propeller set at five locations over the sediment tray. Velocity profiles were collected at 10 mm vertical intervals from the bed (10 mm being the area measured by the propeller). At each vertical interval six, ten second measurements were taken and averaged. Velocity profiles demonstrated that flow was different between flow treatments but not between gravel treatments (Figure 4.4).



Figure 4.3. Sediment tray setup. A tray of loose sediment was added to the flume, flush with the bed. (A) Side profile of the sediment tray, containing a framework of gravel particles infilled with fine sediment. Fresh, algae coated gravel particles were placed on the surface to simulate an armoured gravel-bed. (B) The position of each larvae following the flume experiment was recorded according to their relative exposure zones. (C). Size distribution of fine sediment in the tray. (D) Mean of D_{50} of the minor (c) axis of gravels within each run demonstrates a significant difference in bed protrusion/roughness between small gravel treatments (1 and 3) and large gravel treatments (2 and 4), indicated by * (Tukey HSD, P < 0.05).





4.2.3 Experimental procedure

An aluminium measurement frame was designed which could be placed inside the flume to facilitate accurate determination of the X, Y and Z location of any point within the sediment tray (where X is the streamwise, Y, cross stream and Z vertical position). X and Y location were determined using rulers on the frame, while Z locations were measured using a digital calliper. Comparing the mean of measurements of six known locations before and after each experimental run gave a measurement area for Z of mean = 0.3 mm (max = 1.25 mm, n = 204). this accounts for error in the placement of the measurement frame and the precision of the digital callipers.

Prior to each experiment the sediment tray was installed, and gravel particles placed on top (Figure 4.3A). The measurement frame was used to measure the height of fine sediment within the tray at 10 locations. The measurement frame was then removed during the experiment so that it did not affect flow conditions. 50 *A. fuscipes* larvae were added in each flume run in order

to reproduce a population density in the sediment tray equivalent to that in the field (Chapter 2). *A. fuscipes* larvae were gently removed from their cases using blunt forceps. Larvae were added to the sediment tray under low flow velocity conditions (velocity = 0.111 m s^{-1} , depth = 82 mm) which were maintained for 2 hours, with larvae contained within the sediment tray by a plastic surround. This two hour period was identified in trials to allow larvae to settle and begin case construction (*sensu*. Houghton and Stewart, 1998). Subsequently, the plastic surround was removed, and discharge was gradually increased, and the tail gate lowered to maintain a constant depth of 100 mm until the desired velocity was reached. This velocity was maintained until the following morning (11 am), a total of 21 hrs. At this point discharge was reduced and a series of measurements were taken:

- Larvae location were obtained using the measurement frame and callipers. For larvae which had migrated outside of the sediment tray only X and Y were recorded using a tape measure.
- The position of each larvae with respect to the gravel particle they were located on was recorded according to Figure 4.3B, allowing flow exposure to be inferred (*sensu*. Kovalak, 1976).
- 3) Following the measurement of the visible larvae, the entire base board and measuring frame were removed from the flume. Each gravel particle was lifted to check for larvae on the fine sediment surface or within the substrate, the locations of which were recorded.

A total of 13 flume runs were conducted (using 650 individual *A. fuscipes* larvae). Initially three replicates were intended for each treatment (1-4) however, an extra replicate of treatment 4 was added because of low numbers of larvae remaining to be measured (Table 4.1), and one run of treatment 2 had to be discounted because flow conditions changed overnight.

4.2.4 Caddisfly and case analysis

Cases removed from the larvae prior to flume experiments were dried and stored. Caddisflies and cases built during flume runs were stored in 70% Industrial Methylated Spirits. Larvae were then removed from their cases, identified (all larvae were identified as *A. fuscipes*) and their length and head width measured using a microscope eyepiece graticule calibrated using a stage micrometer. The length and width of cases was measured as for larvae. All cases from each flume run were amalgamated to determine their mass and grain-size characteristics to determine the mass and size fractions of sediment transported by *A. fuscipes*. Cases were included in sediment analysis if they were considered complete (not missing any obvious sections or large

83

grains) and cases that were attached to the gravel particles were not included as they tended to break up when removed from the gravel they were attached to. The results from analysed cases were later extrapolated to the count of all cases found within the sediment tray to calculate the total quantity of sediment used. Cases were thoroughly rinsed in deionised water and then dried at 105°C overnight to obtain mass. Subsequently, case sediment was sieved through 38 mm diameter sieves at half phi intervals down to 0.063 mm, giving a sediment size distribution for cases built in the field and in the flume, for each run.

4.2.5 Data analysis

The X Y and Z of each larvae location were used to ascertain whether the larvae had stayed within the tray or moved downstream or upstream from the tray (as all larvae started in the sediment tray). For longitudinal movement, distance travelled was calculated for each larva from the respective upstream or downstream edge of the sediment tray. Distance travelled in a vertical direction was of primary interest for this experiment. This was calculated by comparing measured height of each larvae at the end of the flume run to the mean level of fine sediment in the tray measured prior to the flume run (Figure 4.3A), according to Equation 1.

$$Z_{dist} = Z_{larvae} - Z_{fine \ sediment} \tag{5}$$

Where Z_{dist} was the vertical distance moved, Z_{larvae} the height of larvae at the end of the run and $Z_{fine \ sediment}$ was the height of the fine sediment surface prior to the flume run.

Linear mixed effects models allowed the best statistical use of this data, considering the full number of larvae in each treatment whilst controlling for any variability resulting from the specific run. Linear mixed effects models were used to investigate the association between velocity and gravel size (as fixed effects) on vertical distance travelled by *A. fuscipes* larvae (response variable), using the lme4 package in R Studio (Bates *et al.*, 2014). Run number was controlled for as a random effect. Visual inspection of the residual plots showed little deviation from homoscedasticity or normality. Significance was determined using likelihood ratio tests (using analysis of variance) between the full model and models with the fixed effect in question removed (with restricted maximum likelihood true for the final model).

4.3 Results

The experiment studied larvae position and movement under four treatments, differing in gravel size /protrusion (Figure 4.3) and flow velocity (Figure 4.4), whilst controlling other physiochemical conditions (Table 4.1). Temperature was slightly higher in the high flow velocity treatments due to the difficulty the coolers had in maintaining a low water temperature when pumps were operating at higher output, but this was not reflected in oxygen saturation, which remained high (Table 4.1). The mean head width of *A. fuscipes* in each run ranged between 0.47 and 0.51 mm, indicating that larvae were final instar (Wallace *et al.*, 2003). Once added to the sediment tray, larvae crawled over the bed surface, sometimes for several tens of minutes before beginning case construction. Several abandoned "emergency" cases were found. These were built from finer sediment than the final cases and had little structural integrity. Final cases were morphologically similar to those built in the field. The number of larvae remaining in the sediment tray (from the original 50), ranged between 13 and 39, averaging 66% although this figure was lower in treatment 4 (Table 4.1). Upstream migration was greater in low flow runs and downstream movement was more common at the high flow velocity (Figure 4.5).

Table 4.1. Details of each treatment. (A) Four treatments were conducted; 1) small gravel, low velocity, 2) large gravel, low velocity, 3) small gravel, high velocity and 4) large gravel, high velocity. For each treatment (B) physiochemical conditions and, (C) details of larvae which stayed within the sediment tray are given. Only those larvae which remained in the sediment tray had their vertical (Z) movement measured. Numbers of larvae remaining are the sum of all runs. Percent remaining is the number of larvae remaining compared to the number added (Number of runs x 50 larvae) and gives an indication of *A. fuscipes* migration out of the sediment tray.

								C. Larvae ii	n sediment
A. Treatment details				B. Physiochemical conditions				tray	
Treatment	Gravel b axis (mm)	Velocity (m s ⁻¹)	Number of runs	Temp (°C)	рН	$\begin{array}{c} \text{Dissolved} \\ O_2(\text{mg L}^{\text{-1}}) \end{array}$	EC (µS cm ⁻¹)	Larvae remaining	Percent remaining
1	32 - 45	0.14	4	13.80	7.39	10.71	324.63	121	69.14
2	45 - 65	0.14	2	13.88	7.48	10.63	314.75	67	71.28
3	32 - 45	0.60	3	15.25	7.48	10.22	314.00	96	69.57
4	45 - 65	0.60	4	14.60	7.36	10.78	315.00	68	53.13
All			13					352	65.79



Figure 4.5. Longitudinal movement of *A. fuscipes* larvae within each treatment. (A) Upstream, and (B) downstream. Only larvae which moved either upstream or downstream of the sediment tray are included. Boxes indicate interquartile range, lines extend within 1.5 times IQR beyond box. Median indicated by a line, mean by x.



Figure 4.6. Vertical sediment transport by Glossosomatidae *A. fuscipes* larvae. (A) Larvae transport sediment upwards, from interstices onto the surfaces of gravel particles. This was true for all treatments and larvae elevation increased with gravel size but not flow velocity. (B) The amount of sediment displaced was largely dependent on the percentage of larvae remaining within the sediment tray and was similar during treatments 1-3 but lower in treatment 4. Boxes indicate interquartile range, lines extend within 1.5 times IQR beyond box. Median indicated by a line, mean by x.

Table 4.2. Characteristics of *A. fuscipes* cases and sediment use within each treatment. (A) Case characteristics; Mean case mass and D_{50} , (B) Sediment displacement; Mean vertical distance moved (Z distance) and sediment flux. Sediment flux is calculated by multiplying the number of larvae remaining in the sediment tray (Table 4.1C) by the mass of their cases (Table 4.2A).

	A. Case c	haracteristics	B. Sediment displacement		
Treatment	Weight (g)	D ₅₀ (mm)	Z distance (mm)	Sediment flux (g m ⁻²)	
1	0.055	1.33	21.83	56.88	
2	0.053	1.29	30.95	59.88	
3	0.054	1.43	24.43	58.19	
4	0.057	1.40	27.05	32.95	
All	0.055	1.37	25.40	50.28	

When all experimental runs and treatments were considered, the vertical elevation of 352 larvae was measured (from an original 650 larvae added to the flume; Table 4.1). The mean vertical migration was 25.40 mm upwards from the level of fine sediment (Table 4.2) and only 6 larvae travelled less than 5 mm vertically (Figure 4.6). Some larvae moved up to 50 mm upwards (Figure 4.6). Consequently, *A. fuscipes* transport sediment upwards by incorporating it into their cases and crawling upwards (Question 1). The quantity of sediment used by larvae in the sediment tray was lower in high flow velocity runs, particularly run 4, as a result of increased larvae migration out of the tray (Table 4.2).

Mixed effects models indicated that bed protrusion affected the vertical distance travelled by A. fuscipes larvae (p = 0.01, Chi² = 6.56), with larvae moving on average 5.49 +/- 2.14 mm (standard error, SE) higher when larger gravel particles were present (Question 2; Figure 4.6). Flow velocity did not significantly affect vertical distance travelled (p = 0.93, Chi² = 0.007), during the low flow velocity runs vertical distance travelled was increased by only 0.18 mm +/- 2.13 (SE). The interaction between gravel size and flow velocity was not significant.

Case characteristics varied between treatments (Figure 4.7; Table 4.2). The size of mineral particles used by larvae during high flow velocity experiments (Treatments 3 and 4) were larger, reflected in the particle size distribution (Figure 4.7A) and the case D_{50} (Figure 4.7B). However, case mass showed no pattern among treatments (Figure 4.7C). The number of larvae occupying sheltered and exposed locations was broadly similar between treatments 1-3, with slightly more larvae in exposed locations than sheltered in each treatment (Figure 4.8). However, during treatment 4 (high flow, large particles) the number of larvae in sheltered locations remains similar but there are far fewer in exposed locations (Figure 4.8). Furthermore, in treatment 4, larvae

were largely recorded in the downstream half of the sediment tray, unlike other treatments where they were more evenly spread (Figure 4.9).



Figure 4.7. Case architecture of *A. fuscipes* larvae from each treatment. (A) particle size distribution for *A. fuscipes* cases from each run with legend indicating treatment. Particle size indicates passing sieve. (B) Case D₅₀ for each treatment. (C) Case mass for each treatment. For boxplots, boxes indicate interquartile range, lines range within 1.5 times IQ beyond box. Median indicated by a line, mean by x.



Figure 4.8. The relative hydraulic exposure of *A. fuscipes* larvae, inferred from the position of larvae (Figure 3B). Larvae on the top or stoss of particles were classified as exposed while larvae on the bottom, side or in the lee of gravels were sheltered. Error bars indicate standard deviation between replicate runs.



Figure 4.9. The location of each larvae (which remained within the sediment tray) for each run within each treatment. Each run had a different gravel particle configuration so precise positions of larvae are not useful. In treatment 4 (and possibly 3) there are more larvae in the downstream end of the sediment tray compared to a more even spread in treatments 1 and 2. Legends show the symbol for larvae of each run.

4.4 Discussion

4.4.1 Vertical displacement of sand by Agapetus fuscipes larvae

The positioning of *A. fuscipes* larvae on the gravel particles was nearly universal (99% of larvae), with only a few individuals found on the fine sediment. During all treatments there was a wide range of vertical elevations of larvae (Figure 4.6A), but larvae predominantly occupied the upper half of gravel particle surfaces (compare Figure 4.3D and Figure 4.6A). Each larvae carried with them a case constructed from sand (mean $D_{50} = 1.37$ mm), resulting in the mean displacement of 0.055 g, 25.40 mm upwards for each *A. fuscipes* individual (Figure 4.7; Table 4.2). Cumulatively, *A. fuscipes* larvae resulted in a substantial flux of sediment from sheltered interstices and onto the gravel surface (Figure 4.6B; 25 – 71 g m⁻²). The abundance of *A. fuscipes* larvae in this study and the quantity of sediment used by these larvae were both representative of the field (sediment use by *A. fuscipes* in Wood Brook Chapter 2, mean = 25.95 g m⁻² max = 88.92 g m⁻²).

The biotic sediment flux is dependent on the mass of each A. fuscipes case and the number of individual larvae exhibiting this behaviour. The mean case mass was greater in this study than Chapter 2 (this chapter = 0.055 g; Table 4.2, Chapter 2 = 0.020 g; Table 2.1), due to the selection of final instar larvae for the experiment. Case mass was broadly similar between treatments, but fewer larvae stayed in the sediment tray during treatment 4, resulting in less sediment displacement within the sediment tray (Table 4.2) and more displaced downstream (Figure 4.5B). Interestingly, whilst there was no difference in case mass between treatments, cases were built from coarser particles in the high flow velocity treatments (Figure 4.7; Table 4.2). Larvae may have used coarser particles to increase their resistance to entrainment, but as case mass was not increased, this would have little effect on case stability. Alternatively, higher flow velocities may have preferentially transported finer sediment particles downstream, effectively armouring the fine sediment surface and leaving only coarser particles available to the larvae. A. fuscipes were found to vary the sediment size used in case construction in the field (Figure 2.5) and are believed to adapt their cases to the sediment sources available. Therefore, this study and Chapter 2 suggest that the size range of sediment displaced by Glossosomatidae may be related to the sand size fractions most readily available to larvae.

This study demonstrates that Glossosomatidae larvae move sediment from where it is available (i.e. sheltered interstices) to more exposed gravel surfaces. The positioning of *A. fuscipes* on the upper surfaces of gravel beds has been widely observed in the field (e.g. Figure 4.1). Furthermore, this behaviour has been documented in the field for a number of other Glossosomatidae species including *Glossosoma nigrior* (Kovalak, 1976), *Glossosoma boltoni* (Scott,

91

1958) and *Agapetus boulderensis* (Olden *et al.*, 2004; Wellnitz *et al.*, 2001). Therefore, this is a widespread behaviour among Glossosomatidae taxa and suggests that the findings of this experiment using *A. fuscipes* may be applicable to other species and beyond laboratory conditions. This also suggests that the behaviour observed in this flume experiment is analogous to that in the field, an important consideration for mesocosm experiments using live organisms (Skelly, 2002; Thomas *et al.*, 2014). This experiment was carefully designed to represent as closely as possible the conditions in Black Brook from which larvae were collected. The use of field conditioned gravels means that food availability was comparable to the field and physiochemical conditions were carefully monitored (Table 4.1). It was important to conduct experiments with large numbers of larvae (650 total) to reduce individual behavioural differences. Furthermore, hydraulics were representative of small streams with low depth in which *A. fuscipes* are commonly abundant (Morris *et al.*, 2015, 2011; Morris and Hondzo, 2013b).

4.4.2 Influence of grain protrusion on sand displacement

Increased bed protrusion resulted in larvae moving to relatively higher elevations (Figure 4.6A). During the low gravel protrusion treatments, the mean elevation of *A fuscipes* larvae was 22 and 24 mm (Table 4.2), compared to a mean particle c axis of 29 mm (Figure 4.3D). Mean larvae elevations during the high protrusion run were 31 and 27 mm compared to mean gravel c axis of 40 and 44 mm (Treatments 2 and 3 respectively; compare Figure 4.6A and Figure 4.3D). Therefore, larvae moved higher up gravel particles but were not restricted to the highest surfaces and occupied a wide range of elevations at all bed protrusions (Figure 4.6A).

Conditions on the top of particles might be preferable to *A. fuscipes* for several reasons, especially respiration efficiency and food availability (Wellnitz *et al.*, 2001). Initial trials for this experiment used clean gravels without an algae coating but most *A. fuscipes* larvae migrated out of the sediment tray. Using fresh algae coated gravels greatly increased larvae retention, suggesting that food is an important control on *A. fuscipes* horizontal movement. A second trial run was conducted with gravel particles placed upside down (with most algae now on the base). This run resulted in substantially reduced vertical movement of *A. fuscipes*. Therefore, although this experiment did not explicitly consider food availability, this finding does support Katano and Doi (2014) and McNeely and Power (2007) who found algal communities to be important for Glossosomatidae larvae distributions.

A. *fuscipes* positions, relative to flow exposure, are a compromise between oxygen availability and flow velocity. Kovalak, (1976) recorded that *G. nigrior* occupied more exposed locations as temperature increased (and therefore dissolved oxygen content decreased) and as flow velocity

decreased. In the current experiment, the action of the pumps and tail gate, and oxygenation in the tanks, maintained high oxygen levels even during low flow runs (Table 4.1). Whilst marginally elevated temperatures during high flow runs may have required larvae to occupy more exposed locations to satisfy oxygen demand, this is unlikely as oxygen concentrations were not reduced during these treatments (Table 4.1). Consequently, oxygen availability was high and therefore is unlikely to be the reason for *A. fuscipes* larvae occupying exposed locations.

4.4.3 Influence of flow velocity on sand displacement

Flow velocity did not affect the vertical elevation of *A. fuscipes*; larvae moved to exposed locations regardless of flow velocity (Figure 4.6A). Whilst vertical distance travelled was not affected by flow velocity, larvae occupied more sheltered positions when flow velocity was high and when the gravel-bed had high protrusion (treatment 4; Figure 4.8). Some previous research has documented Glossosomatidae larvae occupying less exposed locations as flow velocity increases (Brooks, 1998 In: Lake, 2000; Kovalak, 1976; Houghton and Stewart, 1998). For example, flume experiments conducted by Brooks (1998 In: Lake, 2000) found that high flow velocity or turbidity, prompted *A. kimminsi* to migrate from the exposed tops to the sides of stable particles, presumably seeking refugia. However, Glossosomatidae larvae are well adapted to maintain their positions in areas of high hydraulic stress by virtue of their case design and, under hydraulic stress are able to attach directly to gravel surfaces and can move and change direction whilst remaining attached to stable particles (Olden *et al.*, 2004). Consequently, it is probable that the high flow treatment used in this experiment (0.6 m s⁻¹; Figure 4.4) was not enough to restrict larvae movements. Kovalak, (1976) found that Glossosomatidae larvae sought less exposed locations at velocities over 0.7 m s⁻¹.

4.4.4 Combined influence of gravel protrusion and high flow velocity on larvae positioning

A. *fuscipes* larvae preferred exposed locations on gravels (top and stoss) rather than sheltered locations (bottom, side and lee) during treatments 1-3 (Figure 4.8). However, in treatment 4 (high flow, large particles) most larvae occupied sheltered positions, although the elevation of these positions was unchanged (Figure 4.6A). Treatment 4 also differed from other treatments by increased *A fuscipes* larvae migration outside of the sediment tray (largely downstream; Figure 4.5) and positioning of larvae in the downstream half of the sediment tray (Figure 4.9).

It is interesting that treatment 4 was different to the other treatments, which had either particle size or flow velocity in common. This suggests that an interaction effect between high flows and

high particle protrusion affects the exposure of larvae positions, although not their vertical elevation. During high flow runs some transport of fine sediment occurred from between gravel particles and this was greater during treatment 4. The increased gravel-bed protrusion may have caused flow to accelerate over the sediment tray, resulting in increased near-bed velocities. However, this was not reflected in the velocity profiles (Figure 4.4) which show similar near bed velocities and velocity slopes under both high flow treatments, regardless of particle size. Further research is therefore required to characterise *A. fuscipes* case building behaviour (do they use larger sediment?) and positioning (do they seek shelter?) under flow velocities at the upper limit of their tolerance.

4.4.5 Zoogeomorphic effects of sand displacement

In armoured gravel-bed rivers, fine sediment is typically located in sheltered locations; filling interstices, voids and sheltered patches in the lee of larger particles (Dietrich *et al.*, 1989; Parker and Sutherland, 1990). This results in a coarse riverbed surface where transport of finer sediment is limited by its availability on this surface (Church, 2010). Glossosomatidae larvae may affect bed armouring by transporting previously sheltered sand particles out of voids and onto the surface of the gravel-bed, thereby increasing the flow exposure of this sediment (Figure 4.10). Furthermore, Glossosomatidae larvae may move fine sediments in low flow conditions, conditioning the bed and potentially increasing the transport of this sediment during subsequent high flows.



Figure 4.10. Physical bed armouring processes (blue) and Glossosomatidae movement of sediment (green) occurring in an armoured gravel-bed. Movement of sediment, even small distances can substantially alter the exposure of this sediment to entraining flows and may affect downstream sediment flux in rivers. Glossosomatidae transport opposes vertical winnowing and increases the hydraulic exposure of case sediment.

In this study, the cumulative impacts of many individuals resulted in the displacement and increased flow exposure of 25 to 71 g m⁻² (Figure 4.6B) by 25.40 mm. The zoogeomorphic significance of this sediment flux will depend on the stability of sand in its exposed location. The stability of Glossosomatidae sediment in these exposed locations is controlled by 1) the hydraulic and sedimentological conditions in the river and 2) the exposure of the case and 3) the passive and active resistance of the case and larvae to entrainment (Figure 4.11). Passive resistance of *A*. *fuscipes* cases is much higher than that of tubular case species (Figure 3.7A). *A. fuscipes* larvae can also actively resist entrainment by clinging on, by fixing their case to larger particles with silk (Olden *et al.*, 2004), or by moving to a less exposed location (Brooks, 1998; Treatment 4 - Figure 4.9).



Figure 4.11. The stability of an *A. fuscipes* case (C) is dependent upon anchoring of the case by the larvae, (A) and hydraulic force (B). *A. fuscipes* cases may be anchored passively due to case mass and friction (Chapter 3) and actively, due to the larvae clinging on or attaching the case with silk. During pupation cases are attached securely with silk which is likely to substantially increase the stability of the sediment.

The resistance of *A. fuscipes* to drift increases with instar (Van Der Lee *et al.*, 2020), and therefore, cases of early instar Glossosomatidae are more easily eroded whilst, late instar larvae are well adapted to mitigating entrainment (Van Der Lee *et al.*, 2020). Glossosomatidae may fix their cases under hydraulic stress (Olden *et al.*, 2004) and also fix their pupal cases on cobbles in flowing water, often on the upstream face (Anderson and Bourne, 1974). Fixing their cases will considerably increase the hydraulic force required to entrain the constituent sediment (Figure 4.11A). Consequently, sand displaced by Glossosomatidae larvae may be resistant to entrainment and downstream transport, despite the increase in its hydraulic exposure. However, Glossosomatidae are unusual among caddisfly because they build a new case at each instar (Becker, 2005) and under adverse conditions, including low oxygen availability (Morris and Hondzo, 2013a) and fine sediment deposition (Wagner, 1987), Glossosomatidae larvae have

been observed to abandon their case to drift (far more readily than most other caddisfly species; Merrill, 1969; Wagner, 1987). Abandoned Glossosomatidae cases are likely to be more susceptible to entrainment. The stability of the sand displaced by Glossosomatidae larvae is therefore dependent on the behaviour and life stage of the larvae.

4.5 Summary

This experiment demonstrated that Glossosomatidae *A. fuscipes* larvae transport sediment vertically upwards through armoured gravel-bed surfaces. An experimental flume was used to study the movement of Glossosomatidae *A. fuscipes* larvae and the sediment incorporated in their cases. When added to the flume, caseless *A. fuscipes* larvae began to construct cases from the fine sediment available to them below the gravel surface. Larvae transported these cases from sheltered interstices onto surface grains, resulting in a flux of sand (Mean $D_{50} = 1.37$ mm; Table 4.2) vertically upwards by a mean distance of 25.40 mm (Table 4.2; Figure 4.6A). The vertical displacement was greater in treatments with greater bed protrusion, but flow velocity did not affect the vertical distances moved (Figure 4.6A). This behaviour was exhibited by 99% of larvae, resulting in an upwards flux of 25 – 71 g m⁻² (Figure 4.6B) This is the first study to quantify the direction and magnitude of sediment transport caused by active movement of a caddisfly larvae.

The experimental results are comparable to field observations and previous research on positioning of Glossosomatidae spp. on gravel-bed surfaces. Consequently, this displacement of sediment is expected to occur at most sites where *A. fuscipes* and other Glossosomatidae species are abundant, but this will require field validation. Glossosomatidae sediment transport may have zoogeomorphic importance because 1) sediment transport is repeated by 99% of individuals and Glossosomatidae are often abundant, 2) while distances moved are small (25.40 mm) larvae increase the exposure of the sediment they transport, and 3) Glossosomatidae use sand, an important size fraction in gravel-bed rivers. Sediment transport by Glossosomatidae larvae could have consequences for the mobility of the sand involved, but more research is required to consider the entrainment of displaced sediment and how this varies with larval behaviour, including positioning on particles and fixing of larval or pupal cases with silk. This will be required to understand whether Glossosomatidae have a stabilising or destabilising effect on sand sediments in gravel-bed rivers.



Glossosomatidae caddisfly on the River Dove, Peak District, U.K. Photo: S. Rice & M. Johnson.

Chapter 5. The relative contributions of hydraulics and casebuilding caddisfly to bedload transport in a gravel-bed stream

5.1 Introduction

Measuring and predicting the sediment load of rivers is a fundamental objective of fluvial geomorphology. River sediment loads ultimately determine channel morphology, habitat provision and fluvial landscape denudation (Church, 2006; Schumm, 1977). Sediment load is a combination of wash load and bedload (Gordon *et al.* 2004). Bedload maintains frequent contact with the river bed, moving either as contact load (rolling or sliding) or by saltation (hopping; Parker, 2008). The transport of bedload sediment is a function of the force provided by the water and the availability of sediment to be transported. Sediment transport is also controlled by animals and plants (Chapter 1). Many studies have demonstrated the small scale effects of a wide range of animals on sediment transport in rivers (Chapter 1; Rice *et al.*, 2012; Statzner, 2012) but the larger-scale effects on sediment loads in streams are rarely considered.

Case-building caddisfly may affect bedload sediment transport by several mechanisms outlined in Chapter 1 (Figure 1.4). Many species of case-building caddisfly taxa are extremely mobile and directly transport the sand incorporated into their cases. For example, during a flume experiment, *Potamophylax latipennis* (Limnephilidae) crawled 93% of the time (reducing to 78% at high discharge), equivalent to several metres per hour (Lancaster *et al.*, 2006). Furthermore, Limnephilidae larvae may move sediment in their cases large distances. Erman (1986) found that *Chyrnda centralis* migrated first upstream (up to 56.9 m), possibly in response to environmental conditions, and later downstream, prior to pupation. Individual *C. centralis* larvae crawled at up to 1.4 m per hour (Erman, 1986). However, the movement dynamics of aquatic invertebrates are much less studied than their spatial distribution (Lancaster *et al.*, 2008; 2006). Due to their high abundance and relatively high mobility, case-building caddisfly taxa may account for a substantial flux of sediment.

In addition to directly transporting sediment themselves, case-building caddisfly may modify the exposure or stability of sediment, affecting entrainment thresholds and transport by hydraulic processes. Case construction by *P. latipennis* and *Sericostoma personatum*, reduces the entrainment thresholds of incorporated sediment when empty (Figure 3.7A), potentially resulting in preferential entrainment of cases over other river bed sediment. Furthermore, even relatively small movements of sediment by caddisfly larvae may be geomorphologically important if they

98

alter the exposure of sediment grains to entrainment by hydraulic processes. Glossosomatidae move sand from sheltered interstices upwards onto the exposed surface of gravel-beds (Figure 4.6). Therefore, caddisfly larvae may both directly transport sediment and modify its exposure to hydraulic processes. This thesis has identified and examined several mechanisms by which caddisfly may affect sediment dynamics in rivers, using small scale flume experiments (Chapters 3 & 4). However, the wider scale effect of these processes on river sediment load has not been considered.

Biogeomorphological research is often limited by its narrow spatial and temporal scale (Naylor *et al.*, 2002). This is particularly true of zoogeomorphology; most studies are conducted at the scale of organisms (or small communities) over short time periods (hours – weeks; Albertson and Allen, 2015; Rice *et al.*, 2012; Statzner, 2012). As a consequence, the geomorphic implications are often hard to interpret and communicate to geomorphologists who typically work at the scale of river reaches or catchments. Relatively few attempts have been made to measure the zoogeomorphic effects at larger scales in rivers. Rice *et al.* (2016) measured the contribution of signal crayfish to suspended sediment load for a tributary of the River Nene, UK, over a 13-month period and compared this to sediment during base flow (Rice *et al.*, 2016). Furthermore, Hassan *et al.*, (2008) determined that salmonids were responsible for 35% of the bedload transport over a 3 year period at spawning locations on four Creeks draining into Takla Lake, British Columbia, Canada.

A handful of studies have used statistical methods to extrapolate the zoogeomorphic consequences of fish. Rice *et al.* (2019) scored benthivorous fish species according to their zoogeomorphic effectiveness, based on field and flume experiments. They then applied these scores to fish population data to assess the cumulative effects of benthivorous fish at the catchment scale (Rice *et al.*, 2019). Fremier *et al.*, (2017) considered salmonid impacts on river long profiles over geological timescales. However, extrapolating from small scale experiments is subject to considerable error and both studies serve to give only a general indication of the role of the zoogeomorphic effects of fish at larger scales.

Estimates of the zoogeomorphic effects of animals at scales beyond the microhabitat are rare, particularly for smaller organisms such as invertebrates (Rice *et al.*, 2012; Statzner, 2012). Furthermore, whilst they allow for tightly controlled analysis, mesocosm and flume experiments cannot fully replicate the river environment (Kirkegaard *et al.*, 2011; Thomas *et al.*, 2014), thus measuring the impact of caddisfly in the field is vital to understand their zoogeomorphic effects.

99
This chapter presents a reach scale investigation of the proportion of bedload sediment transport that is directly attributable to case-building caddisfly for a small lowland stream (Thesis objective 3). The relative importance of hydraulic and biotic (caddisfly) controls on bedload transport are evaluated and compared to seasonal changes in discharge and caddisfly larvae development over an annual cycle. The chapter also considers the whole caddisfly community, rather than being limited to a single (Chapter 4) or small selection of species (Chapter 3). Consequently, this chapter represents an attempt to increase the spatial, temporal and taxonomic scope of the thesis. This chapter addresses thesis objective 3 via three specific research questions:

- What is the mass and grain size distribution of sediment transport in caddisfly cases, and how does this vary seasonally?
- 2) What is the contribution of individual caddisfly taxa to sediment transport, and how does this vary seasonally?
- 3) How does the magnitude of caddisfly sediment transport compare to mineral and organic sediment transported by hydraulic processes, and how does this vary seasonally?

5.2 Methods

5.2.1 Field site

Sampling was conducted on a 2nd order gravel-bed stream (Black Brook, near Loughborough, U.K. 52°46'33.4"N 1°17'57.6"W; Figure 5.1). Black Brook is separated from arable fields by a wide margin of vegetation including mature trees which shade the site (Figure 5.1A). Discharge in Black Brook is primarily controlled by Black Brook reservoir, 3 km upstream. The study site consisted of two riffles separated by a pool (Figure 5.1B & C).

A detailed reconnaissance of the site was conducted in Summer 2017 during low discharge conditions. Depth and flow velocity were measured at two cross sections where traps were to be located, prior to deployment (Figure 5.2). Velocity was measured using an electromagnetic flow meter at 0.05 m above the bed, averaged over 30 seconds. The bed surface particle size distribution was determined according to Wolman (1954). This demonstrated similar particle size distributions in each riffle (Upper $D_{50} = 31.50 \text{ mm}$, n = 112, Lower $D_{50} = 35.60 \text{ mm}$, n = 120).

5.2.2 Pit trap design and deployment

Pit traps were used to collect sediment transported as bedload, and to isolate the contributions of: 1) hydraulically transported mineral and organic sediment and, 2) sediment transported in caddisfly cases. Pit traps have been used extensively as measures of bedload transport (Einstein,

1944; Poreh *et al.* 1970; Reid, *et al.*, 1980) and allow the collection of long term data (unlike temporary sediment samplers like the Helley-Smith). Pit traps are also used in terrestrial habitats to sample mobile invertebrates (Drake *et al.*, 2007). However, the use of pit traps in rivers has been limited to ephemeral rivers during their dry phases (Corti *et al.*, 2013) because many invertebrates can swim or drift. Case-building caddisfly (particularly those using mineral sediment) travel rarely and only short distances by drift (De Brouwer *et al.*, 2017; Lancaster *et al.*, 2006) and therefore pit traps can be used to sample mobile case-building caddisfly. Pit traps allowed simultaneous sampling of hydraulically transported bedload sediment, mobile caddisfly and caddisfly sediment transported in caddisfly cases, providing a measure of the mass and proportion of bedload flux which is directly attributable to caddisfly.

Circular sediment traps were used so that the direction of travel of sediment and caddisfly did not affect collection efficiency (Sterling and Church, 2002). Traps were formed of an outer tube buried flush with the bed surface, with an inner plastic bucket which could be removed with minimal disturbance to the surrounding bed. The width of each trap at the bed surface was 0.19 m and the depth was 0.22 m which was infilled by 0.04 m of concrete to weigh down empty traps. Traps for the estimation of bedload in rivers are typically larger to catch more saltating grains which may pass over smaller trap diameters (Sterling and Church, 2002). However, traps used for invertebrate sampling are typically much smaller (cup sized; Corti *et al.*, 2013). The size of traps used here was a compromise to ensure trap efficiency, whilst keeping the volume of material collected low enough for subsequent laboratory analysis. Initial burial of the traps caused disturbance to the bed of the river and therefore traps were buried in June 2017 and sampled between March and December 2018.



Figure 5.1. Sampling location on Black Brook near Loughborough. (A) Map showing the sampling site. The upper and lower riffles were separated by a pool. The upper traps were situated midway down the upper riffle in a row of 7. The downstream traps were placed at the head of the lower riffle in a row of 5. An impact plate was placed in the thalweg, upstream of the upper traps. Figure drawn from EDINA 2018 and field reconnaissance. (B) Photograph looking upstream towards the upper riffle traps and impact plate, October 2018. Note that only 4 of the 7 traps are in the active channel at this discharge. (C) Lower traps, looking upstream into pool, December 2018.





Figure 5.2. Depth and velocity profiles for sediment trap cross sections, looking downstream. (A) Upper traps and (B) Lower traps. The upper riffle was wider, with thalweg close to the river left bank and a gravel bar to the right which periodically dried under low discharge. The lower riffle was narrower and flow was funnelled out of the pool resulting in higher velocities. The lower riffle had two thalwegs with a slightly raised bar in the centre which never dried out. Velocity measurements could not be taken when water depth was less than the depth of the flow meter (0.05 m).

Two rows of pit traps were used, with spacing between traps within each row approximately 0.4 m. A row of 7 traps was located mid-way down the upper riffle. At these upper traps, Black Brook had an asymmetric cross-section with a steep left-hand bank and thalweg close to that bank (depth = 0.14 m, Figure 5.2A). Depth gradually decreased towards the right bank, with a gravel bar fully submerged during high discharge but, exposed under low flow conditions (Figure 5.1B). The lower riffle was narrower than the upper (upper width = 4.6 m, lower = 2.9 m) so 5 traps were used. Traps were located at the riffle head, where flow was funnelled out of the pool (Figure 5.1C). Depth was similar and flow velocities slightly higher than at the upper traps (Figure 5.2B). The two sets of traps were separated by a long riffle and deep pool (Figure 5.1A) and the presence of both erosional and depositional environments between the traps meant that sediment trapping in the upper traps was unlikely to have any effect on sediment transport at the lower traps. Traps were thus located in different geomorphological locations, with the upper traps providing an estimate of sediment transport over a riffle and the lower traps situated immediately downstream of a pool, together giving a reach scale estimate of sediment transport

5.2.3 Pit trap sampling

Pit traps were deployed on 6 occasions between February and December 2018 to collect bedload transport and isolate the caddisfly and hydraulic constituents (Table 5.1). Evenly spaced sampling intervals were used to capture seasonal variation in caddisfly sediment use and the abiotic environment, including discharge and hydraulic bedload transport. Traps were usually deployed for 14 days (exceptions are 19 days in August samples and 13 in December; Table 5.1). The contents of all 5 lower traps were aggregated in the field. For the 7 traps at the upper site, only traps within the active channel at the time of collection were sampled (this varied between 4 and 7; Figure 5.1B shows dry gravel bar) and these were also aggregated. Trap samples were sieved using a 0.063 mm mesh in the field and preserved in 4% formaldehyde solution. This resulted in 12 detailed bedload samples; two sets of amalgamated traps (upper and lower) for 6 temporal sampling periods.

Between these detailed sampling periods, trap contents were collected to allow estimation of the annual sediment flux of the stream. Traps were usually emptied before they were 2/3 full because once material builds up inside a pit trap, flow recirculation can cause sediment to be lost (Laronne *et al.*, 2002). This was not possible on one occasion in mid-March because the traps filled within a single flood event. These samples were dried at 105° C overnight and weighed.

Table 5.1. Details of the six sampling occasions. (A) Detailed bedload trap samples were collected on six occasions over the year (date emptied), having been in place for approximately 14 days (19 in August, 13 in December). All lower traps were sampled on each occasion but some upper traps were not in the channel under low discharge (Figure 5.1B) so only 4 were sampled in August and October. Daily discharge indicates the mean for the sampling period, with standard deviation in brackets. (B) Caddisfly species richness and abundance for each sample for upper and lower traps.

A. Sample	characteristi	CS	B. Caddisfly			
Sample Number	Date emptied	Days sampled	Number upper traps	Daily discharge (m ³ s ⁻¹)	Species richness upper/lower	Abundance upper/lower (larvae m ⁻¹ day ⁻¹)
Feb	1.3.18	14	7	0.70 (0.06)	7/6	11.98/17.67
April	3.5.18	14	7	0.75 (0.07)	10/6	134.65/15.71
June	28.6.18	14	7	0.67 (0.004)	9/4	16.81/1.79
August	29.8.18	19	4	0.64 (0.05)	7/6	48.89/122.77
October	22.10.18	14	4	0.66 (0.07)	7/10	185.34/110.30
December	20.12.18	13	7	0.71 (0.08)	10/9	165.82/146.23

5.2.4 Environmental measurements

To provide environmental context, continuous measurement of discharge and bedload transport were required. In addition to the pit trap samples of bedload transport, an impact plate sensor was positioned in the channel thalweg upstream of the upper traps (Figure 5.1A). Impact sensors are steel plates placed flush with the riverbed and equipped with an accelerometer which can measure the impact of moving particles greater than 3 mm diameter (Richardson *et al.*, 2003). They provide good estimates of the timings of bed movement and a less accurate measure of intensity as they can only record three hits per second (Reid *et al.*, 2006). The impact sensor was set to record the number of hits in 5-minute intervals.

Discharge was estimated from an Environment Agency gauge, 1.79 km downstream of the study site. A rating curve was generated from 37 discharge measurements, also collected by the Environment Agency. This was best characterised by two separate linear trends for values above and below 0.5 m depth. Some negative discharge values were predicted by this rating and were assigned a low discharge value of 0.05 m³ s⁻¹.

5.2.5 Laboratory analysis of detailed samples

When the wet mass of the 12 detailed samples exceeded 1 kg, sub sampling was necessary due to the complexity of the laboratory analysis. To do this, the sample was mixed well, and then 1 kg of wet mass was randomly extracted. Samples were subsequently washed and all material > 0.063 mm retained. Material > 0.25 mm was spread out in trays and caddisfly larvae, cases and case fragments removed manually. Many samples contained extremely high abundances of Glossosomatidae larvae. For these samples, this taxon was subsampled by picking ¹/₄ of each tray

to estimate the abundance of larvae and sediment in their cases. The remaining sediment, with caddisfly removed (hereafter, *hydraulic* sediment), was dried at 105°C and sieved through 42 mm -0.063 mm sieves in half phi intervals. The proportion of mineral and organic sediment was measured for gravel (> 2mm) by manually separating organic and mineral sediment. Finer fractions underwent loss on ignition at 550°C for 3 hrs to determine the mass of organic material.

The removed caddisfly and cases (hereafter *caddisfly* sediment) were stored in 70% industrial methylated spirits until they could be identified to the lowest practical level (usually species unless early instar larvae, as for Chapter 2). Caddisfly cases of each species, within each sample, were combined for further analysis. Case samples were rinsed in deionised water and added to test tubes with 30% hydrogen peroxide H_2O_2 at 70°C and stirred periodically. In contrast to Chapter 2, this was also conducted for cases which were partially or wholly organic. Therefore, as soon as silk had broken down, samples were removed from H_2O_2 to prevent further degradation of organic material (OM). While some mass of OM is likely to have been removed by this process, observations suggest this was minimal. Case samples were then rinsed through a 0.063 mm aperture, 38 mm diameter sieve and dried at 70°C for two days. Samples were sieved through 5.6 – 0.063 mm (38 mm diameter) sieves to ascertain grain size distribution (GSD). Subsequently, loss on ignition was conducted for each caddisfly species sample (there was insufficient material to do each size fraction individually).

Therefore, for each of the 12 detailed samples (6 upper and 6 lower) this data analysis established: 1) the mass and GSD used by each caddisfly species and the organic content of the whole sample for each species, and 2) the mass and GSD of hydraulic sediment in the traps, with organic content known for each size fraction. In addition, the mass of sediment in upper and lower traps is known for approximate two-week intervals (less frequently during low discharge periods) between 2nd January and 20th December 2018.

5.3 Data Analysis

5.3.1 Calculating caddisfly sediment flux

First, caddisfly sediment in the traps was standardised to allow comparison between upper and lower trap sites. Within each sample, for each grain size fraction (*i*), of each species, the mass of caddisfly sediment found in traps was standardised to a flux, g_{ci} (g m⁻¹ day⁻¹; Equation 1).

$$g_{ci} = \frac{G_{ci}}{W \times t} \tag{6}$$

Where, G_{ci} = the mass of caddisfly sediment of each grain size fraction (*i*) collected from traps in grams, W = width of traps sampled in m (Equation 2) and t = time traps deployed for (days; usually 14 days; Table 5.1A).

$$W = d \times n \tag{7}$$

Where d = trap diameter (0.19 m), n = number of traps sampled (5 at the lower site and between 4 and 7 at the upper; Table 5.1A).

The mass of sediment used in the cases of each caddisfly species was then determined as the sum of the individual grain size fractions (g_{ci}) for that species. For subsequent analysis, species were also summed into respective caddisfly families (following Chapter 2). The total mass of sediment used by caddisfly (g_c) was calculated by summing the mass of each caddisfly family. To evaluate the contribution of each caddisfly family to the total caddisfly sediment mass (Question 2), the percentage of total caddisfly sediment (g_c) which was attributable to each caddisfly family was determined. The mass of caddisfly sediment which was organic was determined for each caddisfly family, by summing the mass of organic material used by each caddisfly species. This was then converted to a percentage of total caddisfly sediment for each family and subsequently the percentage of caddisfly sediment which was organic across all caddisfly families was calculated.

5.3.2 Calculating hydraulic sediment flux

As for caddisfly, the mass of total hydraulic sediment, was standardised by the width of traps sampled (W, metres) and sampling time (t, days), for each size fraction (i), to give a flux g_{hi} (g m⁻¹ day⁻¹; Equation 3).

$$g_{hi} = \frac{G_{hi}}{W \times t} \tag{8}$$

Where, G_{hi} = the mass of hydraulic sediment of each size fraction (*i*) collected from traps in grams. Total hydraulic sediment flux (g_h) was calculated by summing the individual size

fractions. The percentage of hydraulic bedload which was organic was determined for each size fraction.

5.3.3 Calculating total sediment flux

Bedload flux (g_{bi}) was then calculated for each size fraction (i) by summing caddisfly sediment (g_{ci}) and hydraulic sediment (g_{hi}) (Equation 4).

$$g_{bi} = g_{hi} + g_{ci} \tag{9}$$

In addition, the percentage of the total flux in each individual grain size fraction (*i*) that was recovered from caddisfly cases ($c_{i\%}$) was calculated as the bedload flux in that fraction (g_{bi}) divided by the sum of the corresponding g_{ci} across all species in the sample, multiplied by 100.

5.4 Results

5.4.1 Seasonal discharge and bedload transport in Black Brook

Discharge in Black Brook during the study period was flashy with the highest flows occurring in winter and spring (December 2017 – April 2018; Figure 5.3A). Discharge between May and December was generally low and stable, with a few minor peaks (Figure 5.3A). The mean daily discharge (calculated from the sum of the 15 minute interval data) averaged between December 2017 and December 2018 was $0.72 \text{ m}^3 \text{ s}^{-1}$. The average daily discharge during detailed sampling periods varied from 0.64 to $0.75 \text{ m}^3 \text{ s}^{-1}$. The average of all sampling periods (grey shading; Figure 5.3C) was $0.68 \text{ m}^3 \text{ s}^{-1}$. The max mean daily discharge during a detailed sampling period was $0.93 \text{ m}^3 \text{ s}^{-1}$ (April) which was approximately the Q10 (10 % equalled or exceeded) of the annual mean daily discharge. Therefore, the detailed sampling periods represented 90% of the hydrograph range experienced over the full year. However, the highest discharges of the year (max mean daily discharge = $1.50 \text{ m}^3 \text{ s}^{-1}$) were not sampled and it was during these periods that the majority of bedload transport occurred (Figure 5.3).

There was good corroboration between bedload transport as measured by the impact plate (Figure 5.3B) and traps (Figure 5.3C). Most annual bedload sediment transport in this stream occurred in March and April, with minimal bedload movement after April (Figure 5.3B &C). Bedload transport displays a strong association with discharge (Figure 5.3). The total annual sediment load was 488.64 kg yr⁻¹ for the upper traps and 333.40 kg yr⁻¹ for the lower traps. This

represents a minimum budget because sampling did not begin until 2nd January 2018 (missing 20th Dec to 2nd Jan) and traps were full during the bedload transport peak on 15th March, so some transported sediment was almost certainly not captured by the traps.



Figure 5.3. Discharge and bedload transport estimations for the study period. The six periods of detailed sampling are indicated by shading. (A) Discharge from Environment Agency gauge 1.79 km downstream, (B) bedload estimate from impact plate deployed at the site. (C) Bedload transport estimate from bedload traps in the upper and lower riffles. Missing data indicated by stripes.

5.4.2 What is the mass and GSD of sediment transport in caddisfly cases, and how does this vary seasonally?

14 species of caddisfly were found in the bedload traps; almost exclusively case-building taxa. Three caddisfly species that do not build cases during the larval stage were found; *Polycentropus flavomaculatus* (net builder), *Lype reducta* (fixed retreats/galleries) and a single *Rhyacophila dorsalis* pupal case (architecture groups in Figure 1.4). Species richness in the traps varied between 4 and 10 (mean = 7.25) and abundance in samples from 19 – 2867 (mean = 1139 larvae sample⁻¹; Table 5.1). This was equivalent to an average flux of 81.50 larvae m⁻¹ day⁻¹ (range = 1.79 – 185.34 larvae m⁻¹ day⁻¹) Caddisfly larvae abundance was generally lower in February to June and much higher in August, October and December (Table 5.1). However, April also had a high abundance of larvae in the upper traps (Table 5.1).



Figure 5.4. Characteristics of sediment used by caddisfly (g_c) in each sampling period. (A) Mass, (B) Size distribution and (C) Skewness. In general, there is a strong association between upper and lower traps. There is also a change in the size and skew of sediment between June and August.

The mass of caddisfly sediment was always higher in the upper traps than the lower (Figure 5.4A). The greatest mass of caddisfly sediment at both sites occurred in April (Figure 5.4A), despite low abundance compared to winter months (Table 5.1). The upper traps had a particularly high caddisfly mass at 8.85 g m⁻¹ day⁻¹, compared to a mean annual caddisfly sediment flux 1.27 g m⁻¹ day⁻¹.

Caddisfly predominantly used sediment in the medium to coarse sand size range (mean $D_{50} = 0.91$ mm). Both unimodal and bimodal distributions were present in the caddisfly sediment (Figure 5.5). There is also a notable fining of caddisfly sediment between June and August when the D_{50} dropped substantially from approximately 1 mm to approximately 0.5 mm (Figure 5.4B). This is accompanied by an abrupt change in the skew of the GSD of caddisfly sediment from negative to positive, reflecting the transition from predominantly coarse to fine sediment (Figure 5.4C). In all samples, caddisfly sediment was predominantly mineral (Figure 5.6). At the upper site, the lowest mineral content was 80.07% in February (Figure 5.6A). At the lower traps the lowest mineral content in caddisfly sediment was 96.91% (Figure 5.6B). As an average across all samples, 93.78% of caddisfly sediment was mineral.

5.4.3 What is the contribution of individual caddisfly taxa to sediment transport?

Glossosomatidae were the most abundant caddisfly taxa and dominated caddisfly sediment, accounting on average for 56.11% of sediment mass in each sample (Figure 5.7). The mass of Limnephilidae sediment exceeded Glossosomatidae in some spring samples but was much reduced after June, resulting in a mean contribution to caddisfly sediment of 35% (Figure 5.7). Glossosomatidae dominated the finer sand fractions (up to 1.4 mm) and Limnephilidae the fine gravel fractions (1.4 – 5.6 mm; Figure 5.7). Goeridae used a bimodal range of sediment sizes, but only accounted for 4.67% of caddisfly sediment (Figure 5.7). Sericostomatidae contributed up to 20% of fine sand caddisfly sediment, but overall contributed only 2.67% to total caddisfly sediment transport (Figure 5.7).

The contribution of each caddisfly family to total sediment use varied seasonally with similar trends at both sites (Figure 5.8B). In winter and early spring (December and February) Glossosomatidae and Limnephilidae contributed similar amounts to the total sediment used by caddisfly. The contribution of Limnephilidae peaked in April at both sites but then decreased relative to Glossosomatidae between June and October (Figure 5.8B). This swap from Limnephilidae to Glossosomatidae sediment therefore matches the notable change in larvae abundance (Figure 5.8A) and D₅₀ sediment size and skew occurring between June and August (Figure 5.4B &C).

Glossosomatidae, Sericostomatidae and Goeridae used entirely mineral sediment in their cases (Figure 5.6). Limnephilidae used a mixture of mineral and organic sediment. As a family, the organic proportion of Limnephilidae cases varied from 64% - 99% by mass (Figure 5.6). Limnephilidae in the upper traps used a higher proportion of organic material than those in the lower (Figure 5.6).



Figure 5.5. Grain size distribution of sediment use by caddisfly (g_{ci}) in each sampling period. (A) Upper traps; April has by far the most caddisfly sediment. (B) In the upper traps with April excluded, caddisfly sediment transport in other months was comparable to the lower traps. (C) Lower traps; April again has the greatest mass of sediment transported by caddisfly. Particle size indicates passing sieve.



Figure 5.6. The mineral content of caddisfly sediment for upper (A) and lower traps (B) and split into each caddisfly family. For most species and all families combined, caddisfly sediment transport was predominantly mineral. This was measured by loss on ignition.



Figure 5.7. Percentage of caddisfly sediment use in each size fraction contributed by each caddisfly family (Mean across all samples). The average contribution of each family to total caddisfly transport in each sample is shown in brackets on the legend. Particle size indicates passing sieve.



Figure 5.8. (A) The abundance of caddisfly larvae of each family within each sample for upper and lower traps. (B) the contribution of each caddisfly family to total caddisfly sediment mass in each sample. Taxa accounting for less than 1% of caddisfly sediment (Figure 5.7) were categorised as other.

5.4.4 Hydraulic mineral and organic sediment

Hydraulically transported sediment was divided into mineral and organic components (Figure 5.9). Most samples showed a bimodal size distribution by mass of hydraulic transported sediment, with a mode at medium sand (varying between 0.18 - 0.5 mm) and a mode at gravel (>4 mm; Figure 5.9). Therefore, in most samples there was a lower amount of coarse sand (~0.5 - 2 mm) transported by the flow (Figure 5.9). Of the detailed samples, the most sediment was transported in April in the upper traps and February and April in the lower (Figure 4A). April corresponds to increased discharge (Figure 5.3A), but bedload transport during these samples was still low compared to bedload transport during the interval between detailed sampling periods (Figure 5.3B &C).

Most hydraulically transported sediment < 0.5 mm was mineral as opposed to organic (Figure 5.10). Therefore, the peak in bedload transport of medium sand was predominantly mineral (Figure 5.9). This mode was particularly distinct in the lower traps (Figure 5.9B). The organic

content of sediment increased with grain size up to 2mm; approximately 50% > 2 mm (by mass) was organic (Figure 5.10). The upper traps had larger quantities of coarse organic sediment than the lower (Figure 5.9). This was seasonally variable with greatest percentage of organic sediment transported in late August – December (Figure 5.9A).

5.4.5 How does the magnitude of caddisfly sediment transport compare to mineral and organic sediment transported by hydraulic processes?

Caddisfly transported primarily coarse sand which, in many samples, corresponds with the size range under-represented in hydraulic sediment transport (Figure 5.9). Consequently, sediment in caddisfly cases contributed a substantial proportion of bedload transport (Figure 5.10). On average across all samples at both trap sites, the proportion of sediment used by caddisfly was also unimodal, peaking at coarse sand (36.5 %) and skewed towards smaller particle sizes (Figure 5.10).

The percentage of bedload associated with caddisfly cases varied seasonally (Figure 5.11). At the upper site, caddisfly were responsible for the greatest proportion of bedload transport later in the year (particularly October and December). In October, caddisfly transported 100% of the 0.355 - 0.71 mm bedload (Figure 5.11A). At both sites the percentage of sediment transport which was caddisfly peaked at smaller grain sizes as the year progressed (Figure 5.11). The caddisfly contribution to bedload also varied spatially. In general, the proportion of caddisfly sediment was lower at the lower traps than upper (Figure 5.11). The percentage of caddisfly sediment peaked in the lower traps during April with 72% of 2 mm sediment present in cases.



Particle size (mm)



Figure 5.9. Hydraulic and caddisfly sediment content of traps at each detailed sampling occasion for upper traps (A) and lower traps (B). *Note differing y axis scale. Particle size indicates passing sieve.



Figure 5.10. Contribution of case-building caddisfly ($c_{i\%}$) and hydraulic (mineral and organic) sediment to bedload sediment flux as a mean across both sets of traps and all sampling occasions. Split into particle size categories for simplicity.



Figure 5.11.Percentage of total bedload transport that was caddisfly sediment, for each size fraction in each sample ($c_{i\%}$). Particle size indicates passing sieve.

5.5 Discussion

The effects of aquatic fauna on sediment transport in rivers are poorly understood, particularly for invertebrate taxa (Rice *et al.*, 2012; Statzner, 2012). Case-building caddisfly have important effects on the distribution and entrainment of sand (Chapters 3 and 4), but the implications for sand transport in rivers were unknown. This chapter measured the contribution of case-building caddisfly larvae (caddisfly sediment) and geophysical processes (hydraulic sediment) to bedload transport, during low to medium discharge conditions for a small UK stream. Case-building caddisfly contributed, on average, 1.27 g m⁻¹ day⁻¹, which equates to 464.72 g m⁻¹ a⁻¹. This accounted for 30% of the bedload transport of medium-coarse sand in bedload samples. The importance of caddisfly sediment transport under high discharge and high hydraulic sediment transport conditions is unknown. Nevertheless, under low to medium discharge conditions case-building caddisfly transport substantial proportions of sand in this stream, with potential implications for instream habitat provision.

5.5.1 What is the mass and GSD of sediment transport in caddisfly cases, and how does this vary seasonally?

The results of this study demonstrate that caddisfly larvae contribute to the downstream displacement of bed material; that is, they contribute to bedload sediment flux. During the yearlong study period, case-building caddisfly accounted for an average of 1.27 g m⁻¹ day⁻¹. The maximum mass of caddisfly sediment occurred in April at 8.85 g m⁻¹ day⁻¹ for the upper traps, which equates to 42 g day⁻¹ transported at this cross section, over this two-week period. Caddisfly sediment was largely medium and coarse sand (Figure 5.5; D₅₀ = 0.91 mm). In most caddisfly samples, sediment use was unimodal, although some samples had a bimodal distribution with a peak at fine sediment (about 0.5 mm) and coarser sand (about 2 mm). The size of sediment transported by caddisfly was similar to the sizes used by caddisfly in Wood Brook (D₅₀ = 1 mm; Table 2.1).

Two mechanisms could be responsible for the caddisfly sediment transport recorded in this chapter. Caddisfly may have crawled into the traps or they may have been entrained by the flow and deposited in the traps (Figure 5.12). Active crawling by caddisfly is likely to be an important factor because caddisfly accounted for substantial amounts of bedload sediment transport even when discharge was very low and there was little hydraulic bedload (*cf.* Figure 5.11 & Table 5.1). Limnephilidae are mobile crawlers which follow hydraulically sheltered pathways, related to the

micro-topography of the river bed (Lancaster *et al.*, 2006; Erman, 1986). This mobility could explain their high abundance in traps, which far exceeded their population density in the nearby and similar Wood Brook (Chapter 2). Glossosomatidae, the caddisfly family accounting for the largest proportion of caddisfly sediment, are not known to move large distances horizontally. However, in Chapter 4, *A. fuscipes* larvae moved several metres around the flume in a 24hr period, despite their algae food source only being available in a small test area (Figure 4.5). Large Glossosomatidae populations migrating short distances for food and case building (Chapter 4) could, therefore, account for their high contribution to bedload flux.

Caddisfly larvae and cases may also have been transported into the traps by the flow (Figure 5.12). Whilst this study has differentiated between hydraulic and caddisfly transported bedload, some caddisfly sediment is also likely to have been transported by hydraulic processes. Empty cases of *P. latipennis* were found to be easily entrained (more so than their constituent sand grains; Figure 3.7A). Furthermore, Lancaster *et al.* (2006) found that *P. latipennis* larvae crawled more slowly and shorter distances under high discharge but did not attempt to shelter from increased flow velocity and consequently were moved via entrainment. Without larvae taking active measures to avoid drift, the increase in weight due to a larval inhabitant would probably have minimal effect in reducing entrainment. Therefore, it is likely that empty tubular cases, and those occupied by larvae that do not seek shelter, are preferentially transported over bed sediment (Chapter 3). In Chapter 4, downstream movement of Glossosomatidae increased at high flow velocity (Figure 4.5B; Figure 4.9D), and larvae were observed to be entrained off cobbles and settle in the next interstice. Entrainment of mineral caddisfly cases usually occurs over short distances (Elliott, 1971; Lancaster *et al.*, 2006; Otto, 1976) and whilst probably not detrimental to the larvae it could carry them into the traps in this study.



Figure 5.12. Conceptual model of sediment transport processes applying to caddisfly sediment. Caddisfly and cases may have been transported into the pit traps either by crawling (vertical and horizontal locomotion; blue arrows) or by entrainment (black arrows). The pit traps do not cover the full spectrum of cased caddisfly zoogeomorphic effects because they do not measure stabilisation of sediment by larvae which fix their cases (e.g. for pupation; white arrows).

Bedload traps do not capture the full zoogeomorphic effects of case-building caddisfly. Many caddisfly secure sediment, reducing the potential for entrainment and transport into traps (Figure 12). Rhyacophilidae and Hydropsychidae pupal cases are fixed to large stable particles, increasing the entrainment threshold of incorporated fine sediment (Statzner, 2012). These two families accounted for 15.20% of caddisfly sediment mass in Chapter 2 (Figure 2.3B). Most tubular casebuilders (e.g. Limnephilidae and Sericostomatidae) also fix their cases for pupation and Glossosomatidae may fix their case carlier in their life cycle (Olden *et al.*, 2004). For example, for pupation, *P. latipennis* fix their case to the underside of large particles (often aggregating in groups of 40 or more; Otto and Svensson, 1981). *P. latipennis* cases are therefore unlikely to be mobilised again until flow is sufficient to move these large particles (Figure 5.12). Therefore, in addition to the sediment transport by caddisfly larvae identified in this chapter, a substantial amount of sediment may also be stabilised, particularly during late spring-autumn when most taxa pupate.

The low contribution of caddisfly families to sediment transport (Figure 5.7) does not suggest their zoogeomorphological contribution is less, rather that they are not responsible for transporting sediment horizontally. Net, retreat and pupal case building activities stabilise gravel and fine sediment. Their absence from these traps suggests they do so effectively, as personal observation revealed that Hydropsychidae, at least, were abundant in this stream. Burrowing taxa such as Sericostomatidae may still bioturbate bed sediments (e.g. De Nadaï-Monoury *et al.*, 2013)

but their vertical rather than lateral movements (Wagner, 1991) mean that they are less likely to end up in traps.

5.5.2 Contribution of caddisfly families

This is, to my knowledge, the first use of pit traps to sample macroinvertebrates in active river channels. The abundance of caddisfly and the associated mass of sediment in their cases was unexpectedly high (mean flux of larvae = 81.50 larvae m⁻¹ day⁻¹). However, functional diversity was low; the vast majority of caddisfly found were case-builders. Presumably, caseless taxa are either less abundant, less mobile (e.g. fix themselves to the bed with silk) or were able to swim or drift out of traps. The energy invested in case construction, coupled with the functional benefits of the case (e.g. protection from predation), mean that most caddisfly species are reluctant to abandon their case (Dobson et al., 2000), and are therefore unlikely to escape from traps. Furthermore, the presence of a heavy, mineral case, reduces the tendency to drift and the length of drift events. For example, when entrained in their cases, P. latipennis larvae travel only short distances (about 13 cm), saltating over the river bed (Elliott, 1971; Lancaster et al., 2006; Otto, 1976), and are therefore unlikely to drift out of sediment traps. Therefore, whilst pit traps are not suitable for most aquatic invertebrate taxa, they proved to be very effective for case-building caddisfly. Pit trap contents are dependent on a combination of population density, the movement dynamics of individuals, and the characteristics of the trap and surrounding landscape (Jansen and Metz, 1979). Consequently, using these results to infer population density would be challenging, and require a detailed understanding of the mobility of each taxon. However, when the mobility of larvae is of interest, pit traps provide an excellent sampling method (Corti et al., 2013). In this study, traps were particularly useful as they allowed a direct comparison with hydraulic sediment transport.

Glossosomatidae were the most abundant caddisfly in every sample, however due to the smaller mass of their cases (Table 2.1) the total sediment mass contributed by the much larger Limnephilidae larvae was often greater (Figure 5.7A & B). Glossosomatidae larvae more readily leave their cases, particularly under adverse environmental conditions, such as fine sediment deposition (Nijboer, 2004), which may allow them to drift out of traps. This might account for the number of cases (estimated from the mass of case sediment) outweighing the number of larvae, in some samples. The opposite was found in Chapter 2 (Wood Brook Surber samples) where the abundance of larvae exceeded cases, presumably due to the relative fragility of Glossosomatidae cases. It is therefore possible that the mass of Glossosomatidae sediment

121

found in traps in the present study is an underestimate because larvae abandoned their cases and drifted out of the traps and the cases subsequently disintegrated. Glossosomatidae sediment transport may also be underestimated because they were also the only taxa found attached to the vertical trap sides. Presumably Glossosomatidae were able to escape from the trap by using silk to attach their case to the smooth plastic sides (*sensu*. Olden *et al.*, 2004).

Limnephilidae, whilst less abundant than Glossosomatidae, dominated caddisfly sediment in February and April (Figure 5.8B). There was a drop in the median grain size of caddisfly sediment between June and August (Figure 5.4B), associated with a transition from Limnephilidae to Glossosomatidae dominating caddisfly sediment (Figure 5.8B). This transition between April and August is probably related to the pupation of Limnephilidae taxa at this time. In a Swedish stream *P. latipennis* mostly hatched in October, reached instar IV by February, V in May and pupated between July and August (Otto, 1971). Otto (1971) also documented an increase in movement, both upstream and downstream of *P. latipennis* prior to pupation, hypothesised to be a result of increased food requirements at this time. Therefore, the peak in Limnephilidae sediment may be a combination of increased crawling prior to pupation and the high mass of sediment contained within each final instar Limnephilidae case. Glossosomatidae used a substantial amount of sediment all year round, without the clear seasonal pattern found for Limnephilidae (Figure 5.8B). This is likely to be largely because of overlapping cohorts of larvae of different instars. Figure 5.13 shows four *A. fuscipes* cases from the same sample demonstrating that several instars of this species are present at the same time.

By virtue of their case construction, the caddisfly element of sediment transport could be measured. Isolating biotic turbation activities has only been done a handful of times, but via a number of methods. Rice *et al.* (2016) were able to isolate the fine sediment bioturbation of crayfish temporally, as crayfish were largely active at night and there were few other reasons for a nocturnal peak in sediment. Hassan *et al.* (2008) separated bedload transport due to Salmonidae because this occurred at discharges well below those required for fluvial entrainment of sediment. The disturbance of markers (Rice *et al.*, 2019), tagged particles (Gottesfeld *et al.*, 2004) and characteristic sediment (De Nadaï-Monoury *et al.*, 2013) have also been used to measure bioturbation. Thus, case bioconstruction provided a unique opportunity to study bioturbation caused by caddisfly larvae.



Figure 5.13. Glossosomatidae A. fuscipes cases of different sizes from the same sample. Scale in mm.

5.5.3 How does the magnitude of caddisfly sediment transport compare to hydraulically transported mineral and organic sediment?

During periods of low to medium discharge, caddisfly were responsible for the transport of 30% of the medium to coarse sand (0.25 – 2 mm; Figure 5.10). Caddisfly transported sediment peaked at coarse sand (Figure 5.5). There was no seasonal pattern in either the mass of caddisfly sediment or the proportion of bedload sediment transport that could be attributed to caddisfly, other than a peak in late spring (April).

The sampling periods were determined *a-priori*, to ensure an even spread across the year. Unfortunately, it is not known what the proportion of caddisfly bedload transport was during the large discharge / sediment transport events in spring which were responsible for most bedload transport (Figure 5.3). It is likely that the percentage of caddisfly sediment in traps was reduced because, once the armour layer is disturbed, mass sediment transport often occurs (Parker, 2008). Furthermore, high discharge and bedload movement may trigger caddisfly larvae to seek refugia, reducing their transport of sediment. Consequently, caddisfly bedload transport probably represents a small fraction of total annual bedload transport by mass. However, discharge was highest in April (Figure 5.3) and this sample also had the largest amount of caddisfly sediment. This suggests that increasing discharge increases the number of caddisfly falling into the trap, presumably via entrainment, although caddisfly activity may also be highest at this time for Limnephilidae (Otto, 1971). The two are not mutually exclusive, an increase in time spent crawling will likely also increase the potential for entrainment.

The next step is to understand the contribution of caddisfly to bedload transport during high sediment transport events. Repeating this research with detailed sampling periods stratified to cover a range of discharges (rather than evenly spread across the year) would address this. However, this would be complicated by the difficulty of predicting sediment transport events *a*-*priori* and an uneven spread of samples might complicate understanding of seasonal changes in caddisfly communities and sediment use.

5.5.4 Consequences for fluvial zoogeomorphology

Sediment transport by caddisfly larvae is important because it does not conform to traditional understanding of how sediment moves in rivers. Bedload transport begins when shear forces provided by the flow reach a critical value sufficient to transport surface particles (Shields, 1936; Chapter 3) and transport rate is a function of excess shear stress. In contrast, sand transport by caddisfly larvae occurs even under low discharge, far below the critical threshold for bed mobilisation (compare Figure 5.4 and Figure 5.3A). Therefore, even though caddisfly sediment transport is probably only a small fraction of total bedload transport over the full year, transport of sand during low-medium flow conditions could be geomorphologically important.

Unlike hydraulic transport, caddisfly transport cannot be assumed to be in a downstream direction. Caddisfly movements are a combination of entrainment (downstream) and locomotion, which may be in any direction, but is often upstream (Elliott, 2003; Erman, 1986). Caddisfly movements may not be random, and consequently could result in a systematic transport of sediment in specific directions, related to their biological needs and season. Some evidence exists for mass migrations, rather than just random walks. For example, (Erman, 1986) found that C. *centralis* migrated downstream prior to pupation. Similarly, *Brachycentrus subnubilis* (Brachycentridae) spends most of the year sheltered underneath beds of the macrophyte *Ranunculus*, emerging in spring to more exposed positions for pupation (Gunn, 1985). Furthermore, many caddisfly, including *P. latipennis*, also aggregate for pupation (Martin and Barton, 1987; Otto and Svensson, 1981). Such migrations could result in sand fluxes, with more consistent impacts than if sand is moved randomly over the riverbed.

Black Brook is a small lowland stream with a high abundance and diversity of caddisfly fauna, typical of many European lowland streams. Whilst discharge in Black Brook is to some extent controlled by a reservoir, the stream still experienced flood peaks during the sampling period (Figure 5.3A). Both the mass of sediment used and transported by caddisfly will vary between rivers. In streams with more fluvial disturbance, the importance of sediment movement by caddisfly may be less. Nevertheless, it is likely that similar magnitudes of sediment transport by caddisfly larvae occur in many streams with comparable caddisfly communities.

5.6 Summary

A key challenge for research in biogeomorphology is to measure the effects of organisms at spatial and temporal scales sufficiently large to provide comparison with geophysical processes (Rice *et al.*, 2019). The results of this chapter provide evidence that case-building caddisfly contribute to reach-scale bedload transport during all seasons. Pit traps, commonly used to estimate both bedload transport in rivers and invertebrate communities, provided a novel method to sample and isolate biotic (*caddisfly sediment*) and abiotic (*hydraulic sediment*) components of bedload in a comparable manner. In result, the chapter provides the first reach scale seasonal assessment of zoogeomorphic effects of aquatic insects as well as the first use of pit traps to sample macroinvertebrates in an active river environment.

Case-building caddisfly contributed to the bedload flux of Black Brook during the low to medium flow conditions sampled, an average of 1.27 g m⁻¹ day⁻¹ which equates to 464.72 g m⁻¹ a⁻¹. For Black Brook, over an annual period, this was equivalent to 3.5 kg a⁻¹ at the upper traps and 0.7 kg a⁻¹ at the lower. Many caddisfly species are active crawlers (Lancaster, 2008; Rice *et al.,* 2007) and consequently, caddisfly transported sediment even during the lowest discharge periods. The mass of caddisfly sediment peaked in spring. Caddisfly and empty cases may also have been preferentially entrained into traps. Caddisfly sediment was predominantly coarse sand (D₅₀ = 0.91 mm) and over 90% was mineral sediment (by mass).

During base flow periods case-building caddisfly accounted for 1/3 of coarse sand transport in this stream on average, although they were responsible for over 60% of some size fractions in several samples. However, the majority of bedload transport in this stream occurred in spring between sampling periods and so the importance of case-building caddisfly during periods with high sediment mobility is unknown.

Pit traps capture a combination of taxa abundance and mobility. Glossosomatidae (*A. fuscipes*) accounted for over half of sediment transport and Limnephilidae (predominantly *P. latipennis*) a further $1/3^{rd}$. Limnephilidae show a strong seasonal pattern, ramping up their sediment contribution until pupation in late spring. Taxa which build static cases and nets were notably rare in the traps, suggesting that their stabilisation of sediment may prevent it from moving as bedload, although the abundance of these taxa in this stream is unknown.

Sediment transport in rivers is typically viewed as a physical phenomenon. The results of this chapter add to increasing literature on the importance of biology in sediment processes and provide the first reach-scale quantification of bedload movement by an insect engineer. Furthermore, the movement of sediment by caddisfly does not conform to traditional understanding of bedload dynamics. Sediment may move upstream as well as downstream and although likely influenced by hydraulic processes, high discharge events may reduce caddisfly sediment transport. It is therefore important to better understand sediment engineering roles by invertebrates in gravel-bed streams.



Black Brook, location of Chapter 5, March 2018.

Chapter 6. Discussion

6.1 Introduction

Whilst sediment dynamics in rivers have traditionally been considered a largely physical phenomenon (Reinhardt *et al.*, 2010), it is increasingly recognised that plants and animals can have substantial effects on sediment transport processes (Rice *et al.*, 2012; Statzner, 2012; Wilkes *et al.*, 2018). Most zoogeomorphic research has focussed on vertebrate taxa, but invertebrates also extensively bioturbate and stabilise sediment (Chapter 1). Caddisfly have prompted geomorphological research for their construction of nets and retreats, which stabilise gravel substrates (e.g. Albertson *et al.*, 2019; Johnson *et al.*, 2009), resulting in zoogeomorphic effects as important as those of much larger taxa (Albertson and Allen, 2015). However, documented zoogeomorphic effects of case-building caddisfly are limited to a few examples of bioturbation (*Sericostoma personatum*; De Nadaï-Monoury *et al.*, 2013), bioerosion (sp. unknown; Savrda, 2019) and the formation of calcium carbonate bioherms (e.g. Leggitt and Cushman, 2001). It is

The results of this thesis demonstrate that case-building caddisfly redistribute substantial quantities of fine sediment over riverbeds. This discussion chapter will first evaluate how each of the thesis objectives have been met (Section 6.2). Second, the chapter will consider the zoogeomophic effects of case-construction in more detail (Section 6.3) and how these vary between case architecture groups (Figure 1.3; Section 6.4) and the relative importance of stabilising and destabilising zoogeomorphic mechanisms (Section 6.5). The challenge presented by scaling zoogeomorphic processes is discussed in Section 6.6 and the possible implications for fluvial geomorphology and ecology are discussed in Section 6.7.

6.2 Fulfilment of thesis objectives

The primary aim of this thesis was to investigate and quantify the effect of case-building caddisfly on the distribution and transport of sediment in rivers. This was addressed via three specific objectives;

- 1) To quantify the mass and grain-size of sediment incorporated into caddisfly cases and identify the individual taxa responsible, in riffle habitats of a lowland stream (Chapter 2).
- 2) To determine the effect that case construction by caddisfly larvae has on the hydraulic force required to transport incorporated sediment (Chapter 3).
- 3) To ascertain the importance of caddisfly in transporting and redistributing sediment over the riverbed (Chapters 4 and 5).

The aim and objectives of this thesis were addressed by two field studies and two laboratory flume experiments.

6.2.1 Objective 1 – Quantifying sediment use by case-building caddisfly

The characteristics of sediment used in individual caddisfly cases had been studied for a range of species (e.g. De Gispert et al., 2018; Gaino et al., 2002; Otto and Svensson, 1980). Statzner et al., (2005) and Statzner (2011), however, were the only published studies which spatially quantified sediment use by case-building caddisfly and they only considered a few species. Previous research had also primarily investigated caddisfly case-building from an ecological perspective, to understand the process of case construction (Hansell, 1972; Rowlands and Hansell, 1987), the purpose of cases (Ferry et al., 2013; Otto and Johansson, 1995; Williams et al., 1987) or adaptation of case design to the biotic or abiotic environment (Boyero et al., 2006; Statzner, 2011; Statzner et al., 2005). To understand the geomorphological effects of case-building caddisfly it was necessary to measure sediment use at larger spatial scales, for all taxa living at the site. Objective 1 was addressed using a field study which quantified the sediment used by the caddisfly community found in riffle habitats in a lowland stream (Chapter 2). There was a wide variability in the mass of sediment used in caddisfly cases (mean = of 38 g m⁻², range 4 – 139 g m⁻²). Despite variability in local habitat and caddisfly sediment use, all sites supported diverse case-building caddisfly communities utilising mineral sediment. Consequently, geomorphological effects of case-building caddisfly are potentially widespread. Whilst caddisfly examined in Chapter 2 primarily used coarse sand in case construction ($D_{50} = 1 \text{ mm}$), the range of sediment caddisfly used varied from very fine sand to gravel (0.063 to 11 mm; Figure 2.3), reflecting the grain size requirements or preferences of different species (Figure 2.2.). Therefore, this chapter

indicated that case-building caddisfly may have zoogeomorphic effects across a wide range of grain sizes.

The results of Chapter 2 also demonstrated the diversity of case-building caddisfly behaviour and sediment use within and between species. The average mass of an individual caddisfly case varied between species by several orders of magnitude (0.001 g for early instar Leptoceridae and Lepidostomatidae cases to 0.828 g for Rhyacophilidae and Hydropsychidae pupal cases; Table 2.1). Further research (Chapters 3 & 4) focussed on taxa which used a large amount of sediment or exhibited behaviours which may have zoogeomorphic effects, including crawling with their cases. The taxa which accounted for the most sediment use (e.g. Glossosomatidae = 64%, Rhyacophilidae & Hydropsychidae = 15%, Limnephilidae = 11%) were primarily surface dwellers and therefore, may affect interactions between sediment and hydraulics at the sediment-water interface, potentially affecting the entrainment of this sediment.

Moore (2006) suggested that the size of an organism's zoogeomorphic effect is a function of its abundance, body size and behaviour. Caddisfly were abundant in Wood Brook (mean abundance = 2250 larvae m⁻²), and 90% of these individuals were case-builders rather than net building or free-living taxa. Whilst case-building caddisfly have a small body size (typically < 30 mm), other similar sized taxa have been shown to have important zoogeomorphic effects (e.g. net-building caddisfly - Albertson and Allen, 2015; stonefly - Statzner *et al.*, 1996; Chironomidae - Xing *et al.*, 2018). Chapter 2 concludes that caddisfly case-building effect a substantial amount of surface sediment over a wide range of grain sizes including sand and fine gravel.

6.2.2 Objective 2 – Effects of case construction on sediment entrainment

To understand the zoogeomorphic effect of case-building caddisfly it is necessary to know, not only how much sediment is used in cases, but also how case construction affects the mobility of sediment incorporated within them. Caddisfly cases can be considered composite particles which are substantially larger and comparatively less dense than their constituent sediment grains. Whilst some researchers have investigated the entrainment of caddisfly cases of different species (Otto, 1976; Otto and Johansson, 1995; Waringer, 1993, 1989), the effect of caddisfly bioconstruction on the geomorphological properties of amalgamated sediment had not been examined previously. Chapter 3 presented the result of a detailed flume experiment designed to understand the effect of case construction on sediment entrainment and address thesis objective 2. Chapter 3 compared the entrainment thresholds of empty cases and the loose sediment grains from which individual cases were constructed, for three species with different case designs. The tubular cases of *P. latipennis* and *S. personatum* required significantly less shear stress to move

130

(mean case $\tau_c = 0.06$ and 0.18 N m⁻² respectively), than their loose constituent sediment (mean sediment $\tau_c = 0.35$ and 0.49 N m⁻² respectively, Figure 3.7A). There was no difference in the entrainment threshold of *A. fuscipes* cases compared to the loose sediment from which they were constructed (mean case $\tau_c = 0.28$, sediment = 0.32 N m⁻²). This was hypothesised to be a result of *A. fuscipes* domed case shape and flat base, which increased friction with the riverbed.

The results of Chapter 3 suggest that empty cases of *P. latipennis* and *S. personatum* are more easily entrained than their constituent grains and therefore, by incorporating sediment into their cases, these species may increase the mobility of this sediment. However, these results are limited to empty caddisfly cases, which might occur following case abandonment or emergence of an adult caddisfly from a pupation case. Additional research is required to understand how the behaviour of a live occupant would affect case entrainment. Previous research suggests that caddisfly are effective at avoiding entrainment, by clinging on or moving to more sheltered locations (Waringer, 1989), however larvae may be regularly entrained short distances (Lancaster *et al.,* 2006).

The large numbers of case-building caddisfly collected in sediment traps in Chapter 5 also suggests that caddisfly and their cases are highly mobile (Figure 5.4A & Figure 5.9A). It is likely that entrainment of caddisfly was an important factor in their movement into traps, in addition to larvae crawling. This is supported by the fact that the number of Limnephilidae found in traps, including *P. latipennis*, was highest during periods with the highest discharges (April samples; Figure 5.3 & Figure 5.4), although this may also have coincided with increased locomotive activity prior to pupation. Although the act of case construction by *A. fuscipes* did not reduce the hydraulic force required to transport constituent sediment (as was the situation for *S. personatum* and *P. latipennis*), immediately after case construction, *A. fuscipes* typically transport this sediment vertically upwards, increasing its hydraulic exposure (Chapter 4). This may increase the vulnerability of cases to entrainment and explain why *A. fuscipes* also constituted a substantial proportion of sediment transport in bedload traps in Chapter 5 (Figure 5.7).

6.2.3 Objective 3 – Sediment redistribution by case-building caddisfly

Caddisfly are mobile animals and many species transport their cases with them. Chapters 4 and 5 investigated sediment redistribution by case-building caddisfly (objective 3). The flume experiment reported in Chapter 4, demonstrated a vertical flux of sediment by *A. fuscipes* larvae from sheltered interstices within an armoured gravel-bed, on to the exposed upper surface of gravel particles. On average, each Glossosomatidae larvae was found to transport 0.055 g of sediment upwards by 25.40 mm. Combining this result with the sediment use by

Glossosomatidae populations in Wood Brook (Chapter 2), this flux represents a mean vertical movement of 25.95 gm⁻² (up to 88.92 g m⁻²) of sediment; equivalent to 1 kg of sand for every 40 m² of riverbed. Therefore, Glossosomatidae relocate sediment to more exposed locations which may increase downstream transport of this sediment.

Chapter 5 examined the relative contribution of caddisfly and abiotic processes to bedload transport in a small gravel-bed river and therefore considered the net effect of caddisfly sediment transport via 1) locomotion (e.g. *A fuscipes*, Chapter 4) and 2) entrainment (Chapter 3). The importance of caddisfly to total annual bedload transport was not measured because the highest discharges did not occur during sampled periods. However, during the low to medium discharge conditions when caddisfly contributions to bedload were sampled, case-building caddisfly were unexpectedly mobile; an average of 82 larvae entered the pit traps m⁻¹ day⁻¹ and case-building caddisfly accounted for 30% of medium to coarse sand transport. Therefore, case-building caddisfly actively contributed to sediment transport in this gravel-bed stream.

6.3 Bioconstruction by case-building caddisfly larvae

Five mechanisms are commonly used to conceptualise an organism's biogeomorphic effects (Table 1.1; Corenblit *et al.*, 2011; Naylor *et al.*, 2002; Viles, 2019). The results of this thesis and previous published research provide evidence that caddisfly undertake at least four of these biogeomorphic mechanisms (Figure 6.1). Caddisfly nets provide *bioprotection* (e.g. Johnson *et al.*, 2009), whilst burrowing taxa cause *bioturbation* of soft substrates (De Nadaï-Monoury *et al.*, 2013) and *bioerosion* of hard substrates (Savrda, 2019). An example of caddisfly *bioweathering* remains to be found. This thesis has evaluated the zoogeomorphic effects of *bioconstruction* by caddisfly larvae (Figure 6.1).

Caddisfly cases are unique compared to most bioconstructions documented in the zoogeomorphic literature. First, cases are often transported by the larvae (Chapters 4 & 5). This is common for autogenic bioconstructions (e.g. mussel and snail shells), but far less common for allogenic engineering, (e.g. caddisfly cases). Second, bioconstructions are constructed by individual larvae rather than colonies (e.g. ants), although many individuals may exhibit similar behaviours (Chapter 4) or aggregate (e.g. for pupation; Otto and Svensson, 1981). Case-building caddisfly were found to have destabilising effects on sand distributions in rivers. The cases of some species were intrinsically more mobile than incorporated sediment (Chapter 3) and are

transported by the larvae (Chapter 4), processes which together result in a flux of sediment which can be directly attributed to caddisfly larvae (Chapter 5).

However, sediment destabilisation by case-building caddisfly is very different from most documented bioturbation. Previous research concerning bioturbation has chiefly described movements of particles short, vertical or horizontal distances (François *et al.*, 1997) causing mixing, which in rivers may affect hydraulic connectivity and sediment exposure in relation to potentially entraining flows in coarser river sediments (Johnson *et al.*, 2011; Rice *et al.*, 2019). In contrast, caddisfly transport their case sediment long distances (relative to organism length). Furthermore, caddisfly larvae may transport sediment in specific directions (e.g. Glossosomatidae move sediment vertically upwards). Consequently, the effects of case-building caddisfly on sediment transport processes are likely to be unique.



Figure 6.1. Conceptual zoogeomorphology of caddisfly larvae. Case construction (bioconstruction) may destabilise sediment by 1) modifying the entrainment threshold of the incorporated sediment and 2) moving the case sediment vertically and horizontally. However, case construction can also stabilise sediment if cases are moved to areas of lower hydraulic stress or if cases are fixed in place with silk (bioprotection). Cases are typically fixed for pupation so that many caddisfly species may destabilise sediment for much of their larval life but then stabilise sediment during pupation. The focus in this thesis on bioconstruction adds to the literature on existing caddisfly zoogeomorphic effects (bioprotection by caddisfly nets and retreats and bioturbation and bioerosion by burrowing larvae).

It is also expected that cases fixed to the substrate will stabilise sediment (Figure 6.2). This has been documented for pupal cases of *H. siltalai* (Statzner, 2012), and most species of tubular and dome case-building caddisfly also fix their cases for pupation (Figure 6.2). This process presumably stabilises the constituent sediment particles by fixing the small grains to large and immobile clasts. Therefore, case-building caddisfly may have either destabilising and stabilising effects on river sediment depending on species, life -stage and case-building behaviour.

6.4 Effect of case architecture on caddisfly zoogeomorphology

The behaviour of an organism is a key control on their zoogeomorphic effect (Moore, 2006). However, living organisms exhibit a wide variety of behaviours, which makes conceptualising and understanding their different biogeomorphic effects challenging (Allen *et al.*, 2014). Caddisfly are taxonomically and functionally a very diverse order of aquatic insects (Morse *et al.*, 2019). Nearctic Trichoptera have more genera and occupy a broader range of habitats and trophic categories than many other aquatic insect groups (e.g. Ephemeroptera, Odonata and Plecoptera; Wiggins and Mackay, 1978). Case-building caddisfly are diverse taxonomically and, in their behaviour, and sediment use (Chapter 2), as well as their zoogeomorphic effects (Chapter 3).

Case architecture, therefore, is an important caddisfly behaviour and control on their zoogeomorphic effect. Wiggins (1996) described five broad caddisfly architecture groups (Figure 1.3), four of which were found to have potential zoogeomorphic effects. The fifth group, purse case makers, were low in abundance and used very little sediment in the field studies (Table 2.2; Figure 5.7).



Figure 6.2. Examples of sediment stabilisation by case building caddisfly observed in Black Brook (location of the work reported in Chapter 5) September 2018. (A) Limnephilidae cases fixed to the underside of a brick for pupation. (B) Sericostomatidae case with a trail of larger gravel and fixed to the underside of a large cobble for pupation. (C) Fixed pupal cases of (i) free living or net building taxa like Rhyacophilidae and (ii) Glossosomatidae, left dry after sudden drop in water levels.

6.4.1 Tube case-builders

Tubular case-builders were the most species diverse of the case architecture types in Wood Brook (13 out of 24 taxa identified; Table 2.2). Tubular case-builders typically construct mobile cases which are carried around with the larvae (Figure 6.3) until pupation at which stage they are often fixed to larger, stable cobbles (Figure 6.2; Wiggins, 2004). Case architecture was found to be very variable between species in this group (case mass varied from 0.001 g for early instar Lepidostomatidae and Leptoceridae to 0.146 g for Limnephilidae; D_{50} from 0.17 to 1.12 mm; Table 2.1). Of the tube case-builders, Limnephilidae used most sediment (11% by mass; Figure 2.3) while other tube case-builders each used less than 5%.

Bioconstruction by tubular case-builders destabilises sediment, as tubular cases were inherently more mobile than their constituent grains (Figure 3.7A). Furthermore, larvae may crawl over the riverbed, transporting their case and constituent sediment with them (Figure 5.9). In Black Brook, Limnephilidae accounted for, on average 35% of sediment transport, and up to 70% in April prior to pupation (Figure 5.8B). The transport of sediment by tubular case-builders involves movement of a small number of grains, large distances (relative to organism size) over the riverbed surface. However, tube case-builders may also stabilise sediment. Tube case-builders typically secure their cases to large stable particles for pupation (Figure 6.1A &B).

Whilst bioturbation by burrowing tubular case taxa was not examined in this thesis, previous research has demonstrated that *S. personatum* and *Odontoceridae* larvae cause vertical mixing of fine sediment (De Nadaï-Monoury *et al.*, 2013). Sericostomatidae larvae burrow diurnally to approximately 12 cm depth (Wagner, 1990), increasing the organic matter content of the substrate (Wagner, 1991). *S. personatum* were widespread in riffle habitats (Chapter 2) as well as the pools studied by De Nadaï-Monoury *et al.*, (2013). Bioturbation by burrowing is widely documented in terrestrial, marine and freshwater habitats and therefore is likely to be important in rivers, particularly as sediment disturbed on the surface may then be transported by the flow.


Figure 6.3. Further examples of case construction architecture by tubular case-building caddisfly larvae. (A) Cases of early instar *Athripsodes* sp. (Leptoceridae, left) and *Lepidostoma hirtum* (Lepidostomatidae, right) were the lightest cases recorded in Chapter 2 (mean mass = 0.001 g; Table 2.1). (B) Lepidostomatidae larvae transition from fine mineral grains to a square sectioned organic case. (C) Many organic cases were also found during the work for Chapter 5, particularly for early to mid instar Limnephilidae, ventral view (D) dorsal view of the same case.

6.4.2 Dome case-builders

Glossosomatidae (*Agapetus fuscipes*) were the most abundant taxa in the two field studies (mean density = 1805, max = 6710 larvae m⁻²; Chapter 2). Glossosomatidae also used the most sediment, despite the mean mass of each case being relatively low (0.02 g; Chapter 2). *A. fuscipes* cases required more shear stress to move than cases of tubular caddisfly (Chapter 3), again despite their relatively low mass. The design of *A. fuscipes* cases is hypothesised to increase the hydraulic force required to entrain them, because their flat base increases friction with the substrate and hinders rolling, which is the main mechanism for transport of tubular cases (Chapter 3).

Glossosomatidae A. fuscipes were responsible for the upwards vertical displacement of sediment (Chapter 4), which opposes the geophysical processes partly responsible for bed armouring (downwards winnowing and infiltration of fine sediment). This increases the exposure of transported sediment, but the effect of this transport for sediment flux in rivers depends on the behaviour of larvae under high flows because they may be entrained, move to sheltered areas or fix their sediment to stable particles with silk. Unlike most tube-building taxa which only fix their cases prior to pupation, Glossosomatidae larvae may fix their cases earlier (Olden et al., 2004). Furthermore, Glossosomatidae often fix their cases into hydraulically exposed locations, rather than under stones (as for many tube-case makers; Figure 6.1Ci). Even Glossosomatidae adults use and cement sediment in rivers, by collecting small capstones which are secured on top of egg masses, presumably to provide protection (Figure 6.3; Anderson, 1973). A. fuscipes display a preference for high flow velocities and therefore their sediment interactions have the potential to be particularly zoogeomorphologically important because they move sediment around on the surface and into more exposed areas. In Chapter 5 Glossosomatidae were responsible for 56% of caddisfly sand transport. Therefore, due to their behaviour and abundance, Glossosomatidae are particularly important zoogeomorphic engineers.



Figure 6.4. Glossosomatidae (believed to be *A. fuscipes*) attach their eggs to small gravel particles in the shallow margins of Black Brook (Chapter 5) and cover them with a capstone, fastened in place. Each female *A. fuscipes* larvae caps an average of six egg masses in this manner (Anderson, 1973). Whilst unlikely to have zoogeomorphic effects, this behaviour provides another example of the close-knit interactions between invertebrate life histories and fluvial geomorphology.

6.4.3 Pupal cases of net and retreat building and free-living taxa

This thesis has focused on case construction so has not looked at the sediment stabilisation effect of nets and retreats which are well documented (Johnson *et al.*, 2009). Net and retreat building and free-living taxa have been combined for this discussion because both only build a case for pupation purposes (Figure 1.3). Very few pupal cases of these taxa were recorded in Wood Brook, but they were by far the heaviest cases found (mean mass = 0.83 g; Table 2.1). The caddisfly families Rhyacophilidae and Hydropsychidae, combined accounted for 15% of overall sediment use by mass (Figure 2.3). Pupal cases of these taxa also expanded the size range of sediment used by caddisfly into the fine gravel size range (mean $D_{50} = 4$ mm; Table 2.1).

Statzner, (2012) suggest that pupal cases of *Hydropsyche siltalai* fixed between cobbles decrease the entrainment of the fine gravel incorporated in the cases and the cobbles themselves (Figure 6.4). Therefore, net and retreat building and free-living taxa are primarily sediment stabilisers. In situations where these species are abundant (Statzner, 2011 found 250 *H. siltalai* larvae per m⁻²) they are likely to stabilise considerable mass of fine gravel, as well as the larger particles to which these pupal cases are fixed.



Figure 6.5. Sediment stabilisation by pupal cases of *Hydropsyche siltalai* (Hydropsychidae). Reproduced from Statzner (2012). These cases increase the entrainment thresholds of 1) the fine incorporated into cases, and 2) the ~5 cm diameter gravel particle secured to the much larger clast (and may also reduce the mobility of the larger stone).

Case-building caddisfly may therefore both destabilise and stabilise sediment, based on the case architecture and life stage. Therefore, the zoogeomorphic consequences of case-building caddisfly depend on the net effect of these potentially conflicting processes (Figure 6.1). If transport and stabilisation occur simultaneously, the net effect could be limited. However, if they are separated in space, time, or by the size of sediment affected, the impact could be large. The effects of stabilising and destabilising zoogeomorphic mechanisms may be separated by the hydraulic energy under which they are important (Figure 6.6). Chapter 5 identified sediment transport via caddisfly bioturbation during low to medium flow periods. During the high discharge events which caused general sediment movement (April, Figure 5.3), caddisfly are unlikely to have had a substantial effect, relative to abiotic transport. In contrast, sediment stabilisation behaviours (net/retreat building or fixing of pupal cases) are only geomorphologically important at hydraulic forces above the original entrainment threshold of incorporated sediment (i.e. sediment would have been transported in the absence of caddisfly activities). The effects of sediment stabilisation persist until the hydraulic force overwhelms the silk structure or transports the larger sediment particles that cases are fixed to. Therefore, as a

community, caddisfly larvae may increase sediment mobility under low flows but decrease it under high flows (Figure 6.6).

Furthermore, there is also likely to be a seasonal transition from sediment destabilisation to stabilisation as many species pupate (and fix their sediment) in spring/summer. For example, Limnephilidae (largely *P. latipennis*) were responsible for about 60 - 70 % of caddisfly sediment transport in traps in April but this substantially declined by August to less than 30% (Figure 5.8), presumably due to pupation and anchoring of this sediment.



Figure 6.6. Hypothesised relationship between zoogeomorphic effects (sediment transport) of mobile case building taxa (destabilisation) and fixed cases and nets (stabilisation) with increasing hydraulic power, in gravel-bed rivers. Abiotic sediment transport increases non-linearly with hydraulic power. The stabilising influence of fixed cases and nets becomes important when the entrainment threshold reaches that of fixed sediment and reduces when hydraulic power is sufficient to break apart the stabilisation or entrain the larger particles which sediment is secured to. Mobile case-building caddisfly transport sediment even at the lowest discharges (which occur most of the time). With increasing hydraulic power cases are potentially mobilised earlier than the surrounding riverbed, but their importance in high flows is unknown but probably minimal (Chapter 5).

The zoogeomorphic effects of caddisfly may also be separated by the diameter of sediment affected. Caddisfly used a bimodal grain size distribution of sediment in Wood Brook with mobile case building caddisfly responsible for the mode at 1.4 mm and pupal cases of Rhyacophilidae and Hydropsychidae the mode at 5.6 mm was (Figure 2.3B). This is likely related to the need for mobile taxa to transport their cases whilst caddisfly which build their pupal cases

in situ, do not need to move grains very far (Statzner, 2011). Therefore, caddisfly taxa with different biogeomophic mechanisms could affect different sized grains, with nets and retreats stabilising the gravel framework (e.g. 11- 22 mm; Albertson *et al.*, 2014a), whilst pupal cases of these taxa and free living species stabilise smaller gravel particles ($D_{50} = 4$ mm; Table 2.1) and mobile case-builders destabilise predominantly coarse sand and fine gravel ($D_{50} = 0.91$ mm; Chapter 5), until pupation. Therefore, caddisfly may predominantly increase the mobility of sand, but reduce entrainment of gravel and cobbles.

Destabilising and stabilising effects could also be spatially discrete. At the river segment scale (Figure 6.7), predominantly case-building and net and retreat-building taxa may prefer different environments. However, many individual caddisfly taxa found to be important in this thesis change their case architecture with life stage (e.g. fixing cases for pupation) and therefore, it is likely that both bioturbation and biostabilisation occur at the same sites. Consequently, future research is required to consider the relative importance of stabilisation and destabilising zoogeomorphic effects of case-building caddisfly over spatial, temporal and environmental gradients.

6.5 Scaling up caddisfly zoogeomorphology and future research directions.

This thesis has considered the zoogeomorphic effects of caddisfly across a range of scales (Figure 6.7). At the scale of individual sediment grains, caddisfly construct cases that can increase sediment mobility (Chapter 3; grain scale Figure 6.7), but which can also reduce grain entrainment (Figure 6.5). At the microhabitat scale, case-building caddisfly redistribute grains vertically and horizontally, sometimes altering grain exposure to entrainment (Chapter 4). Furthermore, sediment use by caddisfly larvae varied considerably at the habitat patch scale (the smallest spatial scale investigated in Chapter 2). A flux of caddisfly sediment was documented at the pool/riffle and reach scales (Chapter 5). However, in common with the effects of other zoogeomorphic agents where effects can be identified and demonstrated at small scales (Rice *et al.*, 2019; Statzner, 2012), the zoogeomorphic effects of case-building caddisfly at larger spatial scales remain unknown (Figure 6.7).



Figure 6.7. The role of spatial scale in this thesis and zoogeomorphology more broadly. (A) Scale in rivers, modified from Frissell *et al.* (1986). (B) The zoogeomorphic effects of case-building caddisfly at each scale. At larger spatial scales the effects are unknown. (C) The scale considered in each thesis chapter.

Case-building caddisfly are extremely widespread, found in all biogeographical regions except for the Antarctic (de Moor and Ivanov, 2008), often at high abundances (Morse *et al.*, 2019). Whilst the spatial distribution and abundance of caddisfly taxa are controlled by habitat characteristics (particularly flow velocity and bed sediment) and ecological processes such as dispersal and competition (Murphy and Davy-Bowker, 2005), the diversity of case-building caddisfly mean that taxa with similar case architecture to those found in this thesis are present in most lotic, lentic and even terrestrial habitats (Wallace *et al.*, 2003). It is therefore likely that the zoogeomorphic effects documented in this thesis are spatially widespread.

Similarly, Chapter 5 indicates that sediment transport by case-building caddisfly is not limited by season. Although the individual caddisfly taxa responsible changes (Figure 5.8B), zoogeomorphic effects continue over the entire year (Figure 5.4A). Furthermore, caddisfly cases may be long lasting constructions, surviving beyond pupation. Preliminary results from ongoing experiments

on caddisfly case decay indicate that unoccupied *A. fuscipes* cases fixed to the surface of bricks in rivers remain intact for several months in the absence of large-scale bed movement (Figure 6.8). Cases of *S. personatum* and *P. latipennis* are even more durable. The longevity of caddisfly cases is particularly high in areas with alkaline geology, where cases can become tufarised (Figure 6.9), further stabilising incorporated sediment. Therefore, the spatial and temporal scale of case-building caddisfly zoogeomorphological effects may be widespread and pervasive and long lasting; worthy of additional research.



Figure 6.8. Caddisfly case longevity. Preliminary results from an ongoing experiment to understand how quickly Glossosomatidae *A. fuscipes* cases break down in rivers. Empty larval cases were either attached to bricks using glue to simulate silk or placed within mesh bags and protected from physical erosion. Initial results suggest that, in the absence of high discharge and bed movement, cases break down predominantly by biological or chemical processes rather than physical erosion. Furthermore, for the whole case to degrade usually takes more than 90 days.



Figure 6.9. Empty caddisfly cases from Little Stour River, Kent UK provide surfaces for calcium carbonate deposition.

Scale is a substantial challenge for biogeomorphological research (Naylor *et al.*, 2002). Fluvial processes propagate upstream and downstream (Ward, 1989), and therefore, fluvial geomorphology is typically considered at large spatial scales, from pool-riffle sequences to whole catchments. However, biogeomorphology must consider the behaviour of organisms, often in controlled environments, requiring work at much smaller scales. The focus of fluvial geomorphology on large scale processes may be one reason that zoogeomorphic research in rivers has chiefly been concerned with larger taxa, whilst understanding of the geomorphic importance of invertebrates comes largely from other disciplines (De Nadaï-Monoury *et al.*, 2013b; Mermillod-Blondin, 2011; Mermillod-Blondin and Lemoine, 2010; Savrda, 2019). However, consideration of multiple scales is essential for successful geomorphological research (Thoms and Parsons, 2002). For example, sediment transport at the reach or catchment scale is the result of numerous particle interactions at the grain scale. The challenge is therefore to build a zoogeomorphic framework which can integrate geomorphology and ecology across scales and allow for the complex mechanisms and geomorphic effects of different taxa.

Future research therefore needs to understand the relative importance of stabilisation and destabilisation effects of case-building caddisfly at larger scales. Upscaling the quantification of sediment transport by case-building caddisfly larvae could be done using pit traps as for Chapter 5, considering more rivers and isolating the zoogeomorphic effects of case-building caddisfly during high bedload transport events. To investigate potential stabilisation of sediment by pupating caddisfly might be relatively easy to achieve in the flume at small scales (sensu. Johnson et al., 2009), but upscaling stabilisation effects in the field presents challenges. For example, it is difficult to quantify how much sediment would have been transported without caddisfly activity. The largest spatial scale caddisfly stabilisation experiment to date was conducted by Albertson et al. (2019) in 50 m² semi-natural outdoor river channels, facilities which are rare and would make the necessary spatial and temporal timescales prohibitively expensive. Mobile flumes which can manipulate water velocities over patches of river-bed in the field may provide the best mechanisms for upscaling the results of this thesis. For example, whether A. fuscipes increase the entrainment of sediment by moving it to more exposed locations (Chapter 4) could be determined by manipulating discharge above patches of river bed and measuring whether caddisfly cases or bed sediment moves first. The scale could be increased by undertaking these experiments at nested spatial scales (sensu. Chapter 2) and at various intervals over time (sensu Chapter 5).

Upscaling zoogeomorphic research via statistics and modelling should proceed with caution. There is likely to be some intrinsic bias upscaling from small-scale flume and field experiments because the latter are deliberately conducted at the spatial and temporal scales over which the zoogeomorphic effect in question is suspected to be important. For example, Chapter 4 evaluated the transport of sediment by *A. fuscipes* larvae at population densities equivalent to the average in Wood Brook (Chapter 2). However, densities in Wood Brook varied from 30 to 6710 larvae m⁻² and therefore their zoogeomorphic effects are also considerably more variable than considered in Chapter 4. A related challenge is intra-specific variability. Both field chapters indicated that *A. fuscipes* case mass varied considerably in space (Figure 2.5) and even at the same site (Figure 5.13). Therefore, more research is required to assess the difference in biogeomorphic effects between individuals of the same species. These effects are rarely considered, with researchers typically focusing on the largest (as in Chapter 4) or average-sized animals (Albertson and Allen, 2015).

6.6 Implications for river geomorphology and ecology

Case-building caddisfly affect the transport and distribution of coarse sand and fine gravel in rivers. Whilst it is unknown whether case-building caddisfly significantly modify total sediment flux over larger spatial and temporal scales and they are unlikely to affect large scale landform development, they may nevertheless, have equally important smaller scale effects on fluvial geomorphology and ecology (Figure 6.7). The potential geomorphological implications of caddisfly sediment engineering are discussed in more detail in Chapters 2-4, but a summary and additional ecological effects are proposed here.

Sand transport; Case building caddisfly transport sand during low to medium discharge conditions (Chapter 5). Although the magnitude of sand transported during high discharge events is unknown it is unlikely that caddisfly significantly affect the total sediment flux of Black Brook. However, caddisfly may increase the transport of coarse sand size fractions, changing the grain size distribution of the sediment flux.

Gravel transport; Changes to fine sediment distributions are also important for the transport of coarser, gravel particles. As the proportion of sand increases over 30%, riverbeds become dominated by the sand matrix which may partially bury gravel particles, reducing their exposure to entraining flows. However, the fine sediment reduces the roughness of the bed surface meaning that, if gravels are entrained, they may travel further (overpassing; Carling, 1990; Isla, 1993). Therefore, the sand fraction in gravel-bed rivers can also promote or inhibit the transport of larger gravel clasts, depending on fine sediment proportion (Grams *et al.*, 2007; Venditti *et al.*, 2010; Wilcock *et al.*, 2001). Statzner (2012) estimated that stonefly bioturbating and removing fine sediment from interstices could increase cobble critical entrainment thresholds by 50% (following Wilcox & Kenworthy, 2002). Therefore, it is likely that by creating agglomerate particles from sand (Chapter 3), case-building caddisfly affect the interaction between sand and gravel fractions in rivers and the subsequent transport of both sediment size fractions.

Colmation; Clogging of gravel-bed pore spaces with fine sediment is a widespread problem in rivers (Waters, 1995; Wharton *et al.*, 2017). Colmation reduces the porosity and flow connectivity of substrates and can disconnect riverbed surfaces from subsurface (hyporheic) habitats with negative consequences for biodiversity (Descloux *et al.*, 2013). Bioturbation by fish, crayfish and invertebrates can promote decolmation (Mermillod-Blondin, 2011; Nogaro *et al.*, 2006; Wharton *et al.*, 2017). It is likely that caddisfly also promote decolmation by moving substantial amounts of sediment over the bed surface. The upwards vertical transport of sand by Glossosomatidae larvae (Chapter 4) for example, will reduce the sand content of interstices. Furthermore, locomotive activity of case-building caddisfly will disturb fine sediment on the surface, reducing compaction and potentially increasing entrainment of these size fractions (*sensu*. Statzner, 1996).

Particle surface roughness; Caddisfly cases in exposed locations may change microscale hydraulics and facilitate habitation of these environments by other taxa. Glossosomatidae, in particular, usually inhabit exposed locations where they may add to flow resistance by increasing the roughness of otherwise smooth gravel particles. For example, Poff and Ward (1988) found that *Baetis* sp. (Baetidae Mayfly) nymphs occurred in and on occupied Glossosoma verdona (Glossosomatidae) cases in greater abundance than on rock surfaces without these cases. Furthermore, McCabe and Gotelli, (2003) found that aggregations of *Brachycentrus* (Brachycentridae) and *Neophylax* (Thremmatidae) pupal cases increased the species richness of macroinvertebrates compared with normal substrate. Personal observation of unoccupied A. *fuscipes* cases fixed to gravel surfaces, found a build-up of fine sediment in the lee, providing evidence for reduced flow velocities in the zone directly affected by the construction of a fixed case.

Zoochory; One further result of this thesis was the observation of seeds incorporated into the cases of caddisfly in the sediment traps in Chapter 5 (Figure 6.9). Seeds were mostly incorporated by Limnephilidae larvae (largely *P. Latipennis*). A wide range of seeds types were found. Some of the seeds are believed to be *Rubus* (blackberry; Figure 6.9E) and *Sambacus* (Elderflower; Figure 6.9G). This is the first recorded instance of seed use by caddisfly larvae in case construction. This could be important because *Potamophylax* larvae could disperse seeds (zoochory) within streams. Caddisfly larvae transport could be one of relatively few methods by which seeds could move upstream in rivers. It is unknown whether the seeds were viable as the cases were preserved in formaldehyde. Bio-transportation in caddisfly cases could also be important for other materials such as pollutants, including plastics (Ehlers *et al.*, 2019).



Figure 6.10. Examples of seeds in caddisfly cases observed in samples collected for Chapter 5. Several morphologically distinct seed types were found. Seeds of type G were the most common and were often used by caddisfly whose cases were otherwise largely mineral. Seeds were also used in strategic locations, such as along the sides of cases as in B. Caddisfly may disperse seeds within riverbeds and potentially allow plants to colonise upstream locations. Scale bar indicates approximately 2 mm.

6.7 Concluding remarks

At the beginning of this PhD research the effects of case-building caddisfly on sediment distribution and transport in rivers were unknown. This thesis has demonstrated via flume and field studies at a range of spatial and temporal scales that case-building caddisfly affect the distribution and transport of sediment within gravel-bed rivers.

Case construction by caddisfly larvae resulted in composite particles, which for tubular casebuilding species, reduced the hydraulic force required to mobilise incorporated sediment. Casebuilding caddisfly also transported sediment over the riverbed via crawling and entrainment. As a result of these processes, case-building caddisfly contributed substantially to the transport of coarse sand in low to medium discharge conditions. The zoogeomorphic effects of caddisfly were variable between case architecture types, species and even-within species.

Case-building caddisfly, with similar behaviours to those considered in this thesis are widespread, ubiquitous to rivers and zoogeomorphic effects are not limited to specific seasons. They therefore have the potential to be important zoogeomorphic taxa. Future research is required to understand these effects at larger spatial and temporal scales. The zoogeomorphic effects of case-building caddisfly may have implications for sediment transport and habitat provision in rivers.

Reference list

- Abramoff, M.D., Magalhaes, P.J., Ram, S.J., 2004. Image Processing with ImageJ. Biophotonics International 11, 36–42.
- Albertson, L.K., Allen, D.C., 2015. Meta-analysis: abundance, behavior, and hydraulic energy shape biotic effects on sediment transport in streams. Ecology 96, 1329–1339.
- Albertson, L.K., Sklar, L., Pontau, P., Dow, M., Cardinale, B.J., 2014a. A mechanistic model linking insect (Hydropsychidae) silk nets to incipient sediment motion in gravel-bedded streams. Journal of Geophysical Research: Earth Surface 119, 1833–1852. DOI:10.1002/2013JF003024.
- Albertson, L.K., Cardinale, B.J., Sklar, L.S., 2014b. Non-additive increases in sediment stability are generated by macroinvertebrate species interactions in laboratory streams. PLoS One 9, 1–11. DOI: 10.1371/journal.pone.0103417.
- Albertson, L.K., Daniels, M.D., 2016a. Effects of invasive crayfish on fine sediment accumulation, gravel movement, and macroinvertebrate communities. Freshwater Science 35, 644–653. DOI: 10.1086/685860.
- Albertson, L.K., Daniels, M.D., 2016b. Resilience of aquatic net-spinning caddisfly silk structures to common global stressors. Freshwater Biology 61, 670–679. DOI: 10.1111/fwb.12737.
- Albertson, L.K., Sklar, L.S., Cooper, S.D., Cardinale, B.J., 2019. Aquatic macroinvertebrates stabilize gravel bed sediment: A test using silk net-spinning caddisflies in semi-natural river channels. PLoS One 14, 1–17. DOI: 10.1371/journal.pone.0209087.
- Albertson, L.K., Sklar, L.S., Pontau, P., Dow, M., Cardinale, B.J., 2014. A mechanistic model linking insect (Hydropsychidae) silk nets to incipient sediment motion in gravel-bedded streams. Journal of Geophysical Research: Earth Surface 119, 1833–1852. DOI: 10.1002/2013JF003024.
- Allen, D.C., Cardinale, B.J., Wynn-Thompson, T., 2014. Toward a better integration of ecological principles into ecogeoscience research. Bioscience 64, 444–454. DOI: 10.1093/biosci/biu046.
- Anderson, N.H., 1973. Eggs and oviposition behaviour of *Agapetus fuscipes* Curtis (Trichoptera, Glossosomatidae). Entomologist's Monthly Magazine 109, 129–131.
- Anderson, N.H., Bourne, J., 1974. Bionomics of three species of glossosomatid caddis flies (Trichoptera: Glossosomatidae) in Oregon. Canadian Journal of Zoology 52, 405–411. DOI: 10.1139/z74-049.
- Arce, J.A., Diéguez-Uribeondo, J., 2015. Structural damage caused by the invasive crayfish *Procambarus clarkii* (Girard, 1852) in rice fields of the Iberian Peninsula: A study case. Fundamental and Applied Limnology 186, 259–269. DOI: 10.1127/fal/2015/0715.
- Ashworth, P.J., Ferguson, R.I., 1989. Size-selective entrainment of bed load in gravel bed streams. Water Resources Research 25, 627–634. DOI: 10.1029/WR025i004p00627.
- Atkinson, C.L., Allen, D.C., Davis, L., Nickerson, Z.L., 2018. Incorporating ecogeomorphic feedbacks to better understand resiliency in streams: A review and directions forward. Geomorphology 305, 123–140. DOI: 10.1016/j.geomorph.2017.07.016.

- Auerbach, D.A., Deisenroth, D.B., McShane, R.R., McCluney, K.E., Poff, L.N., 2014. Beyond the concrete: Accounting for ecosystem services from free-flowing rivers. Ecosystem Services 10, 1–5. DOI: 10.1016/j.ecoser.2014.07.005.
- Austen, I., Andersen, T.J., Edelvang, K., 1999. The influence of benthic diatoms and invertebrates on the erodibility of an intertidal mudflat, the Danish Wadden Sea. Estuarine, Coastal and Shelf Science 49, 99–111. DOI: 10.1006/ecss.1998.0491.
- Bae, Y.J., McCafferty, W.P., 1995. Ephemeroptera tusks and their evolution, in: Corkum, L., Ciborowski, J. (Eds.), Current Directions in Research on Ephemeroptera. Canadian Scholars' Press, Toronto, pp. 377–403.
- Balian, E.V., Segers, H., Lévèque, C., Martens, K., 2008. The Freshwater Animal Diversity Assessment: an overview of the results. Hydrobiologia 595, 627–637. DOI: 10.1007/s10750-007-9246-3.
- Barton, D.R., Kelton, N., Eedy, R.I., 2000. The effects of carp (Cyprinus carpio L.) on sediment export from a small urban impoundment. Journal of Aquatic Ecosystem Stress and Recovery 8, 155–159. DOI: 10.1023/A:1011423432727.
- Bates, D., Maechler, M., Bolker, B., Walker, S., 2014. Linear Mixed-Effects Models using "Eigen" and S4 [R package lme4 version 1.1-21].
- Becker, G., 2005. Life cycle of *Agapetus fuscipes* (Trichoptera, Glossosomatidae) in a first-order upland stream in central Germany. Limnologica 35, 52–60. DOI: 10.1016/j.limno.2005.01.003.
- Becker, G., 2001. Larval size, case construction and crawling velocity at different substratum roughness in three scraping caddis larvae. Fundamental and Applied Limnology 151, 317–334. DOI: 10.1127/archiv-hydrobiol/151/2001/317.
- Bergey, E.A., Ward, J. V, 1989. Upstream-downstream movements of aquatic invertebrates in a Rocky Mountain stream. Hydrobiologia 185, 11–82.
- Beschta, R.L., Jackson, W.L., 1979. The Intrusion of Fine Sediments into a Stable Gravel Bed. Journal of the Fisheries Research Board of Canada 36, 204–210. DOI: 10.1139/f79-030.
- Biron, P.M., Robson, C., Lapointe, M.F., Gaskin, S.J., 2004. Comparing different methods of bed shear stress estimates in simple and complex flow fields. Earth Surface Processes and Landforms 29, 1403–1415. DOI: 10.1002/esp.1111.
- Böck, K., Polt, R., Schülting, L., 2018. Ecosystem Services in River Landscapes, in: Schmutz, S., Sendzimir, J., (Eds), Riverine Ecosystem Management. Springer International Publishing, Cham, pp. 413–433. DOI: 10.1007/978-3-319-73250-3_21.
- Booth, M.T., Hairston, N.G., Flecker, A.S., 2020. Consumer movement dynamics as hidden drivers of stream habitat structure: suckers as ecosystem engineers on the night shift. Oikos 129, 194–208. DOI: 10.1111/oik.06396.
- Boyero, L., Rincón, P.A., Bosch, J., 2006. Case selection by a limnephilid caddisfly [Potamophylax latipennis (Curtis)] in response to different predators. Behavioral Ecology and Sociobiology 59, 364–372. DOI: 10.1007/s00265-005-0059-y.
- Brandt, S.A., 2000. Classification of geomorphological effects downstream of dams. Catena 40, 375–401.

- Breukelaar, A.W., Lammens, E., Breteler, J., Tátrai, I., 1994. Effects of benthivorous bream (*Abramis brama*) and carp (*Cyprinus carpio*) on sediment resuspension and concentrations of nutrients and chlorophyll a. Freshwater Biology 32, 113–121. DOI: 10.1111/j.1365-2427.1994.tb00871.x.
- Bridge, J.S., Bennett, S.J., 1992. A model for the entrainment and transport of sediment grains of mixed sizes, shapes, and densities. Water Resources Research 28, 337–363. DOI: 10.1029/91WR02570.
- Brooks, A.P., Brierley, G.J., 2002. Mediated equilibrium: The influence of riparian vegetation and wood on the long-term evolution and behaviour of a near-pristine river. Earth Surface Processes and Landforms 27, 343–367. DOI: 10.1002/esp.332.
- Brown, A.V., Brussock, P.P., 1991. Comparisons of benthic invertebrates between riffles and pools. Hydrobiologia 220, 99–108. DOI: 0.1007/BF00006542.
- Brown, S.A., Ruxton, G.D., Humphries, S., 2004. Physical properties of *Hydropsyche siltalai* (Trichoptera) net silk. Journal of the North American Benthological Society 23, 771–779. DOI: 10.1899/0887-3593(2004)023<0771:PPOHST>2.0.CO;2.
- Bunte, K., Abt, S.R., 2001. Sampling Surface and Subsurface Particle-Size Distributions in Wadable Gravel- and Cobble-Bed Streams for Analysis in Sediment Transport, Hydraulics, and Streambed Monitoring: General Technical Report RMRS-GTR-74. Rocky Mountain Research Station.
- Burner, C.J., 1951. Characteristics of spawning nests of Columbia River salmon. U. S. Fish and Wildlife Service Fishery Bulletin 61, 97–110.
- Butler, D.R., 1995. Zoogeomorphology: Animals as Geomorphic Agents. Cambridge University Press, Cambridge.
- Butler, D.R., 1992. The grizzly bear as an erosional agent in mountainous terrain. Zeitschrift für Geomorphologie 36, 179–189.
- Butler, D.R., Malanson, G.P., 2005. The geomorphic influences of beaver dams and failures of beaver dams. Geomorphology 71, 48–60. DOI: 10.1016/j.geomorph.2004.08.016.
- Butler, D.R., Sawyer, C.F., 2012. Introduction to the special issue-zoogeomorphology and ecosystem engineering. Geomorphology 157–158, 1–5. DOI: 10.1016/j.geomorph.2012.02.027.
- Buxton, T.H., Buffington, J.M., Yager, E.M., Hassan, M.A., Fremier, A.K., 2015. The relative stability of salmon redds and unspawned streambeds. Water Resources Research 51, 6074–6092. DOI: 10.1002/2015WR016908.
- Cadée, G.C., 2001. Sediment Dynamics by Bioturbating Organisms. Springer, Berlin, pp. 127– 148. DOI: 10.1007/978-3-642-56557-1_7.
- Cardinale, B.J., Duffy, J.E., Gonzalez, A., Hooper, D.U., Perrings, C., Venail, P., Narwani, A., Mace, G.M., Tilman, D., Wardle, D.A., Kinzig, A.P., Daily, G.C., Loreau, M., Grace, J.B., Larigauderie, A., Srivastava, D.S., 2012. Biodiversity loss and its impact on humanity. Nature 486, 59–67. DOI: 10.1038/nature11148.
- Cardinale, B.J., Gelmann, E.R., Palmer, M.A., 2004. Net spinning caddisflies as stream ecosystem engineers: the influence of *Hydropsyche* on benthic substrate stability. Functional Ecology 18, 381–387. DOI: 10.1111/j.0269-8463.2004.00865.x.

- Carling, P.A., 1990. Particle over-passing on depth-limited gravel bars. Sedimentology 37, 345–355. DOI: 10.1111/j.1365-3091.1990.tb00963.x.
- Carling, P.A., Kelsey, A., Glaister, M.S., 1992. Effect of Bed Roughness, Particle Shape and Orientation on Initial Motion Criteria, in: Billi, P., Hey, R.D., Thorne, C.R., Tacconi, P. (Eds.), Dynamics of Gravel -Bed Rivers, John Wiley and Sons, Chichester, pp 23-37.
- Cashman, M.J., Wharton, G., Harvey, G.L., Naura, M., Bryden, A., 2019. Trends in the use of large wood in UK river restoration projects: insights from the National River Restoration Inventory. Water and Environment Journal 33, 318–328. DOI: 10.1111/wej.12407.
- Carter, N., Viles, H., 2005. Bioprotection explored: The story of a little known earth surface process. Geomophology 67, 273–281. DOI: 10.1016/j.geomorph.2004.10.004.
- Castro, J.M., Thorne, C.R., 2019. The stream evolution triangle: Integrating geology, hydrology, and biology. River Research and Applications 35, 315–326. DOI: 10.1002/rra.3421.
- Chaalali, A., Brind'amour, A., Stanislas, Dubois, F., Le Bris, H., 2017. Functional roles of an engineer species for coastal benthic invertebrates and demersal fish. Ecology and Evolution 7, 5542–5559. DOI: 10.1002/ece3.2857.
- Charbonneau, P., Hare, L., 1998. Burrowing behavior and biogenic structures of mud-dwelling insects. Journal of the North American Benthological Society 17, 239–249. DOI: 10.2307/1467965.
- Church, M., 2010. Gravel-Bed Rivers, in: Burt, T.P., Allison, R.J. (Eds.), Sediment Cascades: An Integrated Approach. John Wiley & Sons, Chichester, pp. 241–269.
- Church, M., 2006. Bed Material Transport and the Morphology of Alluvial River Channels. Annual Review of Earth and Planetary Sciences 34, 325–354. DOI: 10.1146/annurev.earth.33.092203.122721.
- Ciutat, A., Weber, O., Gérino, M., Boudou, A., 2006. Stratigraphic effects of tubificids in freshwater sediments: a kinetic study based on X-ray images and grain-size analysis. Acta Oecologica 30, 228–237. DOI: 10.1016/j.actao.2006.04.004.
- Constanza, R., d'Arge, R., de Groot, R., Farber, S., Grasso, M., Hannon, B., Limburg, K., Naeem, S., O'Neill, R.V., Paruelo, J., Raskin, R.G., Sutton, P., van den Belt, M., 1997. The value of the world's ecosystem services and natural capital. Nature 387, 253–260. DOI: 10.1038/387253a0.
- Coombes, M.A., 2016. Biogeomorphology: diverse, integrative and useful. Earth Surface Processes and Landforms 41, 2296–2300. DOI: 10.1002/esp.4055
- Corenblit, D., Baas, A.C.W., Bornette, G., Darrozes, J., Delmotte, S., Francis, R.A., Gurnell, A.M., Julien, F., Naiman, R.J., Steiger, J., 2011. Feedbacks between geomorphology and biota controlling Earth surface processes and landforms: A review of foundation concepts and current understandings. Earth-Science Reviews 106, 307-331. DOI: 10.1016/j.earscirev.2011.03.002
- Corenblit, D., Tabacchi, E., Steiger, J., Gurnell, A.M., 2007. Reciprocal interactions and adjustments between fluvial landforms and vegetation dynamics in river corridors: A review of complementary approaches. Earth-Science Reviews 84, 56–86. DOI: 10.1016/j.earscirev.2007.05.004.
- Corey, A.T., 1949. Influence of shape on the fall velocity of sand grains. Unpublished M.S. Thesis, Colorado Agricultural and Mechanical College, Fort Collins.

- Corti, R., Larned, S.T., Datry, T., 2013. A comparison of pitfall-trap and quadrat methods for sampling ground-dwelling invertebrates in dry riverbeds. Hydrobiologia 717, 13–26. DOI: 10.1007/s10750-013-1563-0.
- Cotton, J.A., Wharton, G., Bass, J.A.B., Heppell, C.M., Wotton, R.S., 2006. The effects of seasonal changes to in-stream vegetation cover on patterns of flow and accumulation of sediment. Geomorphology 77, 320–334. DOI: 10.1016/j.geomorph.2006.01.010
- Cowan, J.A., Humphreys, G.S., Mitchell, P.B., Murphy, C.L., 1985. An assessment of pedoturbation by two species of mound-building ants, *Camponotus intrepidus* (Kirby) and *Iridomyrmex purpureus* (F. smith). Australian Journal of Soil research 23, 95–107. DOI: 10.1071/SR9850095.
- Cudney, M.D., Wallace, J.B., 1980. Life Cycles, Microdistribution and Production Dynamics of Six Species of Net-Spinning Caddisflies in a Large Southeastern (U.S.A.) River. Holarctic Ecology 3, 169–182. DOI: 10.2307/3682366.
- Cummings, E.R., 1932. Reefs or bioherms? Geological Society of America Bulletin 43, 331-352.
- Cummins, K.W., 1995. Invertebrates, in: Calow, P., Petts, G.E. (Eds.), The Rivers Handbook. Blackwell Scientific, Oxford, pp. 234–250.
- Cummins, K.W., Lauff, G.H., 1969. The Influence of Substrate Particle Size on the Microdistribution of Stream Macrobenthos. Hydrobiologia 34, 145–181.
- Curran, J.C., Waters, K.A., 2014. The importance of bed sediment sand content for the structure of a static armor layer in a gravel bed river. Journal of Geophysical Research: Earth Surface 119, 1484–1497. DOI: 10.1002/2014JF003143.
- Dafoe, L.T., Rygh, A.L., Yang, B., Gingras, M.K., Pemberton, S.G., 2011. A new technique for assessing Tubificid burrowing activities, and recognition of biogenic grading formed by these Oligochaetes. Palaios 26, 66–80. DOI: 10.2110/palo.2010.p10-023r.
- Dangerfield, J.M., Mccarthy, T.S., Ellery, W.N., 1998. The mound-building termite *Macrotermes michaelseni* as an ecosystem engineer. Journal of Tropical Ecology 14, 507–520. DOI: 10.1017/S0266467498000364.
- Darwin, C.R., 1881. The formation of vegetable mould, through the action of worms, with observations on their habits. John Murray, London.
- Davidson, T.M., Altieri, A.H., Ruiz, G.M., Torchin, M.E., 2018. Bioerosion in a changing world: a conceptual framework. Ecology Letters 21, 422–438. DOI: 10.1111/ele.12899.
- De Brouwer, J., Besse-Lototskaya, A., Ter Braak, C., Kraak, M., Verdonschot, P., 2017. Flow velocity tolerance of lowland stream caddisfly larvae (Trichoptera). DOI: 10.1007/s00027-016-0507-y.
- De Gispert, Q., Alfenas, G., Bonada, N., 2018. Grain size selection in case building by the mountain cased-caddisfly species *Potamophylax latipennis* (Curtis, 1834): a trade-off between building time and energetic costs 37, 33–45. DOI: 10.23818/limn.37.04.
- de Moor, F.C., Ivanov, V.D., 2008. Global diversity of caddisflies (Trichoptera: Insecta) in freshwater. Hydrobiologia 595, 393–407. DOI: 10.1007/s10750-007-9113-2.
- De Nadaï-Monoury, E., Lecerf, A., Canal, J., Buisson, L., Laffaille, P., Gilbert, F., 2013. A costeffective method to quantify biological surface sediment reworking. Hydrobiologia 713, 115–125. DOI: 10.1007/s10750-013-1497-6

- Delgado, J.A., Carbonell, R., 1997. Case features of caddisfly larvae (*Sericostoma selysi*) as related to water velocity and potential to drift. Journal of Freshwater Ecology 12, 193–197. DOI: 10.1080/02705060.1997.9663526.
- Demir, T., 2000. The influence of particle shape on bedload transport in coarse-bed river channels. Doctoral thesis, University of Durham.
- Descloux, S., Datry, T., Marmonier, P., 2013. Benthic and hyporheic invertebrate assemblages along a gradient of increasing streambed colmation by fine sediment. Aquatic Sciences 75, 493–507. DOI: 10.1007/s00027-013-0295-6.
- DeVries, P., 2012. Salmonid influences on rivers: A geomorphic fish tail. Geomorphology 157– 158, 66–74. DOI: 10.1016/J.GEOMORPH.2011.04.040.
- Dey, S., Papanicolaou, A., 2008. Sediment Threshold under Stream Flow: A State-of-the-Art Review. KSCE Journal of Civil Engineering 12, 45–60. DOI: 10.1007/s12205-008-8045-3.
- Dietrich, W.E., Kirchner, J.W., Ikeda, H., Iseya, F., 1989. Sediment supply and the development of the coarse surface layer in gravel-bedded rivers. Nature 340, 215–217.
- Dobson, M., Poynter, K., Cariss, H., 2000. Case Abandonment as a Response to Burial by *Potamophylax cingulatus* (Trichoptera: Limnephilidae) Larvae. Aquatic Insects 22, 99–107. DOI: 10.1076/0165-0424(200004)22:2;1-P;FT099.
- Dodds, G.S., Hisaw, F.L., 1925. Ecological studies on aquatic insects. III. Adaptations of caddisfly larvae to swift streams. Ecology 6, 123–137.
- Dorn, R., 2014. Ants as a powerful biotic agent of olivine and plagioclase dissolution. Geology 42, 771–774. DOI: 10.1130/G35825.1.
- Drake, C.M., Lott, D.A., Alexander, K.N.A., Webb, J., 2007. Surveying terrestrial and freshwater invertebrates for conservation evaluation (NERR005). Natural England Research Report 139.
- Dudgeon, D., 1990. Functional significance of selection of particles and their use by aquatic animals in the construction of external structures. In Wotton (Eds) The Biology of Particles in Aquatic Systems (2nd ed.). CRC Press Inc., Boca Raton. pp. 289–312.
- Dudgeon, D., Arthington, A.H., Gessner, M.O., Kawabata, Z., Knowler, D.J., Lévêque, C., Naiman, R.J., Prieur-Richard, A., Soto, D., Stiassny, M.L.S., Sullivan, C.A., 2006. Freshwater biodiversity: importance, threats, status and conservation challenges. Biological Reviews 81, 163–182. DOI: 10.1017/S1464793105006950.
- Eckman, J.E., Nowell, A.R.M., Jumars, P.A., 1981. Sediment destabilization by animal tubes. Journal of Marine Research 39, 361–374.
- Edington, J.M., Hildrew, A.G., 1995. Caseless caddis larvae of the British Isles with notes on their ecology (2nd ed). Freshwater Biological Association, Ambleside. pp. 136.
- Edmunds, G.F., Mccafferty, W.P., 1996. New Field Observations On Burrowing In Ephemeroptera From Around The World. Entomological News 107, 68–76.
- Edwards, R.W., 1962. Some Effects of Plants and Animals on the Conditions in Fresh-Water Streams With Particular Reference To Their Oxygen Balance. International Journal of Air and Water Pollution 6, 505–520. DOI: 10.1016/b978-1-4832-8391-3.50024-5.

- Edwards, T.K., Glysson, G.D., 1999. Field Methods for Measurement of Fluvial Sediment, in: Book 3, Applications of Hydraulics. Techniques of Water-Resources Investigations of the U.S. Geological Survey, p. 89.
- Ehlers, S.M., Manz, W., Koop, J.H.E., 2019. Microplastics of different characteristics are incorporated into the larval cases of the freshwater caddisfly Lepidostoma basale. Aquatic Biology 28, 67–77. DOI: 10.3354/ab00711.
- Einstein, H.A., 1944. Bed-load transportation in mountain creek. United States Department of Agriculture Technical Bulletin, 55. Clemson, Greenville.
- Elliott, J.M., 2003. A comparative study of the dispersal of 10 species of stream invertebrates. Freshwater Biology 48, 1652–1668. DOI: 10.1046/j.1365-2427.2003.01117.x.
- Elliott, J.M., 1971. The Distances Travelled by Drifting Invertebrates in a Lake District Stream. Oecologia 6, 350–379.
- Erman, N.A., 1986. Movements of self-marked caddisfly larvae, *Chyrnda centralis* (Trichoptera: Limnephilidae), in a Sierran spring stream, California, U.S.A. Freshwater Biology 16, 455–464. DOI: 10.1111/j.1365-2427.1986.tb00988.x.
- Everall, N.C., Johnson, M.F., Wood, P., Farmer, A., Wilby, R.L., Measham, N., 2017. Comparability of macroinvertebrate biomonitoring indices of river health derived from semi-quantitative and quantitative methodologies. Ecological. Indicators 78, 437–448. DOI: 10.1016/j.ecolind.2017.03.040.
- Faller, M., Harvey, G.L., Henshaw, A.J., Bertoldi, W., Bruno, M.C., England, J., 2016. River bank burrowing by invasive crayfish: Spatial distribution, biophysical controls and biogeomorphic significance. Science of the Total Environment 569–570, 1190–1200. DOI: 10.1016/j.scitotenv.2016.06.194.
- Fei, S., Phillips, J., Shouse, M., 2014. Biogeomorphic Impacts of Invasive Species. Annual Review of Ecology, Evolution, and Systematics 45, 69–87. DOI: 10.1146/annurevecolsys-120213-091928.
- Feldmeth, C.R., 1970. The respiratory energetics of two species of stream caddisfly larvae in relation to water flow. Comparative Biochemistry and Physiology 32, 193–202. DOI: 10.1016/0010-406X(70)90933-3.
- Feller, C., Brown, G.G., Blanchart, E., Deleporte, P., Chernyanskii, S.S., 2003. Charles Darwin, earthworms and the natural sciences: various lessons from past to future. Agriculture, Ecosystems & Environment 99, 29–49. DOI: 10.1016/S0167-8809(03)00143-9.
- Ferry, E.E., Hopkins, G.R., Stokes, A.N., Mohammadi, S., Brodie, E.D., Gall, B.G., 2013. Do all portable cases constructed by caddisfly larvae function in defense? Journal of Insect Science 13, 1–9. DOI: 10.1673/031.013.0501.
- François, F., Poggiale, J.C., Durbec, J.P., Stora, G., 1997. A new approach for the modelling of sediment reworking induced by a macrobenthic community. Acta Biotheoretica 45, 295– 319. DOI: 10.1023/A:1000636109604.
- Fremier, A.K., Yanites, B.J., Yager, E.M., 2017. Sex that moves mountains: The influence of spawning fish on river profiles over geologic timescales. Geomorphology 305, 163–172. DOI: 10.1016/j.geomorph.2017.09.033.

- Friedrichs, M., Graf, G., Springer, B., 2000. Skimming flow induced over a simulated polychaete tube lawn at low population densities. Marine Ecology Progress Series 192, 219–228. DOI: 10.3354/meps192219.
- Frissell, C.A., Liss, W.J., Warren, C.E., Hurley, M.D., 1986. A hierarchical framework for stream habitat classification: Viewing streams in a watershed context. Environmental Management 10, 199–214. DOI: 10.1007/BF01867358.
- Gabet, E.J., Reichman, O.J., Seabloom, E.W., 2003. The effects of bioturbation on soil processes and sediment transport. Annual Review of Earth and Planetary Sciences 31, 249–73. DOI: 10.1146/annurev.earth.31.100901.141314.
- Gaino, E., Cianficconi, F., Rebora, M., Todini, B., 2002. Case-building of some Trichoptera larvae in experimental conditions: Selectivity for calcareous and siliceous grains. Italian Journal of Zoology 69, 141–145. DOI: 10.1080/11250000209356451.
- Gani, M.R., Gani, N.D., 2011. River-margin habitat of *Ardipithecus ramidus* at Aramis, Ethiopia 4.4 million years ago. Nature Communications 2, 1–5. DOI: 10.1038/ncomms1610.
- Garcia, C., Laronne, J.B., Sala, M., 1999. Variable source areas of bedload in a gravel-bed stream. Journal of Sedimentary Research 69, 27–31. DOI: 10.2110/jsr.69.27.
- Garcia, M., (Ed.) 2008. Sedimentation Engineering: Processes, Measurements, Modeling, and Practice. American Society of Civil Engineers, Reston. DOI: 10.1061/9780784408148.
- Gautreau, E., Volatier, L., Nogaro, G., Gouze, E., Mermillod-Blondin, F., 2020. The influence of bioturbation and water column oxygenation on nutrient recycling in reservoir sediments. Hydrobiologia 847, 1027–1040. DOI: 10.1007/s10750-019-04166-0
- Gordon, N.D., McMahon, T.A., Finlayson, B.L., Gippel, C.J., Nathan, R.J., 2004. Stream Hydrology: An Introduction for Ecologists (2nd ed). John Wiley and Sons, Chichester.
- Goring, D.G., Nikora, V.I., 2002. Despiking Acoustic Doppler Velocimeter Data. Journal of Hydraulic Engineering 128, 117–126. DOI: 10.1061/(ASCE)0733-9429(2002)128.
- Gottesfeld, A.S., Hassan, M.A., Tunnicliffe, J.F., Poirier, R.W., 2004. Sediment dispersion in salmon spawning streams: the influence of floods and salmon redd construction. Journal of the American Water Resources Association 40, 1071–1086. DOI: 10.1111/j.1752-1688.2004.tb01068.x.
- Grabowski, R.C., Gurnell, A.M., Burgess-Gamble, L., England, J., Holland, D., Klaar, M.J., Morrissey, I., Uttley, C., Wharton, G., 2019. The current state of the use of large wood in river restoration and management. Water and Environment Journal 33, 366–377. DOI: 10.1111/wej.12465.
- Graça, M.A.S., 2001. The Role of Invertebrates on Leaf Litter Decomposition in Streams a Review. International Review of Hydrobiology 86, 383–393. DOI: 10.1002/1522-2632(200107)86:4/5<383::AID-IROH383>3.0.CO;2-D.
- Grams, P.E., Wilcock, P.R., 2007. Equilibrium Entrainment of Fine Sediment Over a Coarse Immobile Bed. Water Resources Research 43, W10420. DOI: 10.1029/2006WR005129.
- Gunn, R.J., 1985. The biology of *Brachycentrus subnubilus* Curtis (Trichoptera) in the River Frome, Dorset. Hydrobiologia 120, 133–140.
- Gurnell, A.M., England, J., Burgess-Gamble, L., 2019. Trees and wood: working with natural river processes. Water Environment Journal 33, 342–352. DOI: 10.1111/wej.12426.

- Gurnell, A.M., 1998. The hydrogeomorphological effects of beaver dam-building activity. Progress in Physical Geography 22, 167–189. DOI: 10.1177/030913339802200202.
- Gurnell, A.M., 2014. Plants as river system engineers. Earth Surface Processes and Landforms 39, 4–25. https://doi.org/10.1002/esp.3397.
- Gurnell, A.M., Piégay, H., Swanson, F.J., Gregory, S. V, 2002. Large wood and fluvial processes. Freshwater Biology 47, 601–619. DOI: 10.1046/j.1365-2427.2002.00916.x.Haidvogl, G., 2018. Historic Milestones of Human River Uses and Ecological Impacts, in: Schmutz, S., Sendzimir, J., (Eds), Riverine Ecosystem Management. Springer International Publishing, Cham, pp. 19–39. DOI: 10.1007/978-3-319-73250-3_2.
- Han, X., Fang, H.W., Johnson, M.F., Rice, S.P., 2019. The Impact of Biological Bedforms on Near-Bed and Subsurface Flow: A Laboratory-Evaluated Numerical Study of Flow in the Vicinity of Pits and Mounds. Journal of Geophysical Research: Earth Surface 124, 1939– 1957. DOI: 10.1029/2019JF005000.
- Hansell, M., 1968. The House Building Behaviour of the Caddis-fly Larva Silo Pallipes Fabricius: II. Description and Analysis of the Selection of Small Particles. Animal Behaviour 16, 562–577. DOI: 10.1016/0003-3472(68)90052-3.
- Hansell, M.H., 1972. Case building of the caddis fly larva, *Lepidostoma hirtum*. Journal of Zoology 167, 179–192. DOI: 10.1111/j.1469-7998.1972.tb01728.x.
- Hart, D.D., 1985. Grazing Insects Mediate Algal Interactions in a Stream Benthic Community. Oikos 44, 40. DOI: 10.2307/3544041.
- Hart, D.D., Resh, V.H., 1980. Movement patterns and foraging ecology of a stream caddisfly larva. Canadian Journal of Zoology 58, 1174–1185. DOI: 10.1139/z80-162.
- Harvey, G.L., Bertoldi, W., 2015. Dynamic riverine landscapes: The role of ecosystem engineers. Earth Surface Processes and Landforms 40, 1701–1704. DOI: 10.1002/esp.3757.
- Harvey, G.L., Henshaw, A.J., Brasington, J., England, J., 2019. Burrowing Invasive Species: An Unquantified Erosion Risk at the Aquatic-Terrestrial Interface. Reviews of Geophysics 57, 1018–1036. DOI: 10.1029/2018RG000635.
- Harvey, G.L., Henshaw, A.J., Moorhouse, T.P., Clifford, N.J., Holah, H., Grey, J., Macdonald, D.W., 2014. Invasive crayfish as drivers of fine sediment dynamics in rivers: field and laboratory evidence. Earth Surface Processes and Landforms 39, 259–271. DOI: 10.1002/esp.3486.
- Hassan, M.A., Gottesfeld, A.S., Montgomery, D.R., Tunnicliffe, J.F., Clarke, G.K.C., Wynn, G., Jones-Cox, H., Poirier, R., Macisaac, E., Herunter, H., Macdonald, S.J., 2008. Salmon driven bed load transport and bed morphology in mountain streams. Geophysical Research Letters 35 L04405. DOI: 10.1029/2007GL032997.
- Hassan, M.A., Petticrew, E.L., Montgomery, D.R., Gottesfeld, A.S., Rex, J.F., 2011. Salmon as Biogeomorphic Agents in Gravel Bed Rivers: The Effect of Fish on Sediment Mobility and Spawning Habitat. Geophysical Monograph Series 194, 337–352. DOI: 10.1029/2010GM000968.
- Hastings, A., Byers, J.E., Crooks, J.A., Cuddington, K., Jones, C.G., Lambrinos, J.G., Talley, T.S., Wilson, W.G., 2007. Ecosystem engineering in space and time. Ecology Letters 10, 153– 164. DOI: 10.1111/j.1461-0248.2006.00997.x.

- Hauer, C., Leitner, P., Unfer, G., Pulg, U., Habersack, H., Graf, W., 2018. The Role of Sediment and Sediment Dynamics in the Aquatic Environment, in: Schmutz, S., Sendzimir, J., (Eds), Riverine Ecosystem Management. Springer International Publishing, Cham, pp. 151–169. DOI: 10.1007/978-3-319-73250-3_8.
- He, X., Chen, Z.-Q., Lu, Z., Li, J., Hu, W., Li, S., Xu, Z., 2015. Exceptionally Preserved Caddisfly Larval Cases (Insecta) from the Lower Cretaceous of the Liupanshan Basin, Western China. Journal of Earth Science 26, 192–202. DOI: 10.1007/s12583-015-0525z.
- Hickin, N., 1967. Caddis Larvae. Hutchinson & Co, London.
- Holzenthal, R.W., Blahnik, R.J., Prather, A.L., Kjer, K.M., 2007. Order Trichoptera Kirby, 1813 (Insecta), Caddisflies. Zootaxa 1668, 639–698.
- Houghton, D.C., Stewart, K.W., 1998. Life History and Case-Building Behavior of *Culoptila cantha* (Trichoptera: Glossosomatidae) in the Brazos River, Texas. Annals of the Entomological Society of America 91, 59–70. DOI: 10.1093/aesa/91.1.59.
- Humphreys, G.S., Mitchell, P.B., 1983. A preliminary assessment of the role of bioturbation and rainwash on sandstone hillslopes in the Sydney Basin, in: Young, R.W., Nanson, G.C. (Eds.) Aspects of Australian Sandstone Landscapes. Australian and New Zealand Geomorphology Group, Special Publication No. 1, pp. 66–80.
- Ikeda, H., Iseya, F., 1988. Experimental study of heterogeneous sediment transport, Environmental Research Center paper No. 12, University of Tsukuba, Japan.
- Isla, F.I., 1993. Overpassing and armouring phenomena on gravel beaches. Marine Geology 110, 369–376. DOI: 10.1016/0025-3227(93)90094-C.
- Jackson, W.L., and Beschta, R.L., 1984. Influences of increased sand delivery on the morphology of sand and gravel channels. Water Resources Bulletin 20, 527–533. DOI:10.1111/j.1752-1688.1984.tb02835.x.
- Jackson, J.K., McElravy, E.P., Resh, V.H., 1999. Long-term movements of self-marked caddisfly larvae (Trichoptera: Sericostomatidae) in a California coastal mountain stream. Freshwater Biology 42, 525–536. DOI: 10.1046/j.1365-2427.1999.00503.x.
- Jansen, M.J.W., Metz, J.A.J., 1979. How many victims will a pitfall make? Acta Biotheoretica 28, 98–122. DOI: 10.1007/BF00046807.
- Jesson, M.A., Bridgeman, J., Sterling, M., 2015. Novel software developments for the automated post-processing of high volumes of velocity time-series. Advances in Engineering Software 89, 36–42. DOI: 10.1016/j.advengsoft.2015.06.007.
- Johansson, A., 1991. Caddis larvae cases (Trichoptera, Limnephilidae) as anti-predatory devices against brown trout and sculpin. Hydrobiologia 211, 185–194. DOI: 10.1007/BF00008534.
- Johnson, M.F., Reid, I., Rice, S.P., Wood, P.J., 2009. Stabilization of fine gravels by net-spinning caddisfly larvae. Earth Surface Processes and Landforms 34, 413–423. DOI: 10.1002/esp.1750.
- Johnson, M.F., Rice, S.P., Reid, I., 2011. Increase in coarse sediment transport associated with disturbance of gravel river beds by signal crayfish (*Pacifastacus leniusculus*). Earth Surface Processes and Landforms 36, 1680–1692. DOI: 10.1002/esp.2192.

- Johnson, M.F., Rice, S.P., Reid, I., 2010. Topographic disturbance of subaqueous gravel substrates by signal crayfish (*Pacifastacus leniusculus*). Geomorphology 123, 269–278. DOI: 10.1016/j.geomorph.2010.07.018.
- Johnson, M.F., Thorne, C.R., Castro, J.M., Kondolf, G.M., Mazzacano, C.S., Rood, S.B., Westbrook, C., 2019. Biomic river restoration: A new focus for river management. River Research and Applications 36, 3–12. DOI: 10.1002/rra.3529.
- Jones, C.G., 2012. Ecosystem engineers and geomorphological signatures in landscapes. Geomorphology 157–158, 75–87. DOI: 10.1016/j.geomorph.2011.04.039.
- Jones, C.G., Lawton, J.H., Shachak, M., 1994. Organisms as ecosystem engineers. Oikos 69, 373–386.
- Juras, M., Albertson, L.K., Cahoon, J., Johnson, E., 2018. Incorporating macroinvertebrate biological structures into gravel-bedded stream fluid dynamics using 3D CFD modelling. Ecological Engineering 119, 19–28. DOI: 10.1017/CBO9781107415324.004.
- Karman, T.V., 1931. Mechanical similitude and turbulence. Technical Memorandums, National Advisory Committee on Aeronautics, Washington DC. Reprint from Nachrichten von der Gesellschaft der Wissenschaften zu Gottingen - Fachgruppe I (Mathematik). No. 5, pp. 58–76, 1930.
- Katano, I., Doi, H., 2014. Stream grazers determine their crawling direction on the basis of chemical and particulate microalgal cues. PeerJ e503. DOI: 10.7717/peerj.503.
- Kaval, P., 2019. Integrated catchment management and ecosystem services: A twenty-five year overview. Ecosystem Services 37, 100912. DOI: 10.1016/j.ecoser.2019.100912.
- Kim, S.-C., Friedrichs, C.T., Member, J., Maa, P.-Y., Wright, L.D., 2000. Estimating Bottom Stress in Tidal Boundary Layer from Acoustic Doppler Velocimeter Data, Journal of Hydraulic Engineering 126, 399–406. DOI: 10.1061/(ASCE)0733-9429(2000)126:6(399).
- Kirkegaard, J., Wolters, G., Sutherland, J., Soulsby, R., Frostick, L., McLelland, S., Mercer, T., Gerritsen, H., 2011. Users Guide to Physical Modelling and Experimentation, CRC Press, Boca Raton. DOI: 10.1201/b11335.
- Kohler, S.L., 1992. Competition and the Structure of a Benthic Stream Community. Ecological Monographs 62, 165–188. DOI: 10.2307/2937092.
- Kohler, S.L., McPeek, M.A., 1989. Predation Risk and The Foraging Behavior of Competing Stream Insects. Ecology 70, 1811–1825. DOI: 10.2307/1938114.
- Kondolf, G.M., Boulton, A.J., O'Daniel, S., Poole, G.C., Rahel, F.J., Stanley, E.H., Wohl, E.,
 Bång, A., Carlstrom, J., Cristoni, C., Huber, H., Koljonen, S., Louhi, P., Nakamura, K.,
 2006. Process-based ecological river restoration: Visualizing three-dimensional
 connectivity and dynamic vectors to recover lost linkages. Ecology and Society 11, 5.
- Kovalak, W.P., 1976. Seasonal and diel changes in the positioning of *Glossosoma nigrior* Banks (Trichoptera: Glossosomatidae) on artificial substrates. Canadian Journal of Zoology 54, 1585–1594. DOI: 10.1139/z76-184.
- Kuhnle, R,A,, 1993. Fluvial transport of sand and gravel mixtures with bimodal size distributions. Sedimentary Geology 85, 17–24. DOI:10.1016/0037-0738(93)90072-D.
- Lake, P.S., 2000. Disturbance, patchiness, and diversity in streams, Journal of the North American Benthological Society 19, 573–592. DOI: 10.2307/1468118.

- Lancaster, J., 2008. Movement and dispersion of insects in stream channels: What role does flow play?, in: Lancaster, J., Briers, R. (Eds.), Aquatic Insects: Challenges to Populations. CABI, Wallingford, pp. 139–157. DOI: 10.1079/9781845933968.0139.
- Lancaster, J., Buffin-Bélanger, T., Reid, I., Rice, S., 2006. Flow and substratum-mediated movement by a stream insect. Freshwater Biology 51, 1053–1069. DOI: 10.1111/j.1365-2427.2006.01554.x.
- Laronne, J., Garcia, C., Reid, I., 2001. Mobility of patch sediment in gravel bed streams: Patch character and its implications for bedload, in. Mosley, P.M. Gravel-Bed Rivers V, New Zealand Hydrological Society Inc., Wellington. pp. 249–289.
- Laronne, J.B., Alexandrov, Y., Bergman, N., Cohen, H., Garcia, C., Habersack, H., Powell, D.M., Reid, I., 2002. The continuous monitoring of bed load flux in various fluvial environments. Erosion and Sediment Transport Measurement in Rivers; Technological and Methodological Advances (Proceedings of the Oslo Workshop. June 2002). IAHS Publ. 283. 2003.
- Law, A., Gaywood, M.J., Jones, K.C., Ramsay, P., Willby, N.J., 2017. Using ecosystem engineers as tools in habitat restoration and rewilding: beaver and wetlands. Science of the Total Environment 605–606, 1021–1030. DOI: 10.1016/j.scitotenv.2017.06.173.
- Leggitt, V.L., Cushman, R.A., 2001. Complex caddisfly-dominated bioherms from the Eocene Green River Formation. Sedimentary Geology 145, 377–396. DOI: 10.1016/S0037-0738(01)00155-5.
- Limm, M.P., Power, M.E., 2011. The caddisfly *Dicosmoecus gilvipes*: making a case for a functional role. Journal of the North American Benthological Society 30, 485–492. DOI: 10.1899/10-028.1.
- Luckenbach, M.W., 1986. Sediment stability around animal tubes: The roles of hydrodynamic processes and biotic activity. Limnology and Oceanography 31, 779–787. DOI: 10.4319/lo.1986.31.4.0779.
- Maguire, Z., Tumolo, B.B., Albertson, L.K., 2020. Retreat but no surrender: net-spinning caddisfly (Hydropsychidae) silk has enduring effects on stream channel hydraulics. Hydrobiologia 847, 1539–1551. DOI: 10.1007/s10750-020-04210-4.
- Malm, T., Johanson, K.A., Wahlberg, N., 2013. The evolutionary history of Trichoptera (Insecta): A case of successful adaptation to life in freshwater. Systematic Entomology 38, 459–473. DOI: 10.1111/syen.12016.
- Malmqvist, B., Rundle, S., 2002. Threats to the running water ecosystems of the world. Environmental Conservation 29, 134–153. DOI: 10.1017/S0376892902000097.
- Mao, L., Cooper, J.R., Frostick, L.E., 2011. Grain size and topographical differences between static and mobile armour layers. Earth Surface Processes and Landforms 36, 1321–1334. DOI: 10.1002/esp.2156.
- Marchant, R., 1988. Vertical Distribution of Benthic Invertebrates in the Bed of the Thomson River, Victoria. Australian Journal of Marine and Freshwater Research 39, 775–84.
- Maroulis, J.C., Nanson, G.C., 1996. Bedload transport of aggregated muddy alluvium from Cooper Creek, central Australia: a flume study. Sedimentology 43, 771–790. DOI: 10.1111/j.1365-3091.1996.tb01501.x.

- Martin, I.D., Barton, D.R., 1987. The formation of diapause aggregations by larvae of *Neophylax fuscus* Banks (Trichoptera: Limnephilidae) and their influence on mortality and development. Canadian Journal of Zoology 65, 2612–2618. DOI: 10.1139/z87-396.
- Martin, V., Fisher, T.S.R., Millar, R.G., Quick, M.C., 2002. ADV Data Analysis for Turbulent Flows: Low Correlation Problem, in: Proceedings of Hydraulic Measurements and Experimental Methods Conference, ASCE Conference Proceedings. pp. 770–779. DOI: 10.1061/40655(2002)101.
- Masteller, C.C., Finnegan, N.J., 2017. Interplay between grain protrusion and sediment entrainment in an experimental flume. Journal of Geophysical Research 122, 274–289. DOI: 10.1002/2016JF003943.
- Mathers, K.L., Rice, S.P., Wood, P.J., 2019. Predator, prey, and substrate interactions: the role of faunal activity and substrate characteristics. Ecosphere 10, e02545. DOI: 10.1002/ecs2.2545.
- Mathers, K.L., White, J.C., Fornaroli, R., Chadd, R., 2020. Flow regimes control the establishment of invasive crayfish and alter their effects on lotic macroinvertebrate communities. Journal of Applied Ecology 57, 886–902. DOI: 10.1111/1365-2664.13584.
- McCabe, D.J., Gotelli, N.J., 2003. Caddisfly diapause aggregations facilitate benthic invertebrate colonization. Journal of Animal Ecology 72, 1015–1026. DOI: 10.1046/j.1365-2656.2003.00779.x.
- Mclean R.F. (1982) Zingg shape. In: Schwartz, M. (Ed.), Beaches and Coastal Geology. Encyclopedia of Earth Science. Springer, Boston.
- McLelland, S.J., Nicholas, A.P., 2000. A new method for evaluating errors in high-frequency ADV measurements. Hydrological Processes 14, 351–366. DOI: 10.1002/(SICI)1099-1085(20000215)14:2<351::AID-HYP963>3.0.CO;2-K.
- Mcneely, C., Power, M.E., 2007. Spatial Variation in Caddisfly Grazing Regimes Within a Northern California Watershed. Ecology 88, 2609–2619. DOI: 10.1890/06-0796.1.
- Meadows, P.S., Tait, J., 1989. Modification of sediment permeability and shear strength by two burrowing invertebrates, Marine Biology 101, 75–82. DOI: 10.1007/BF00393480
- Meadows, P.S., Tait, J., Hussain, S.A., 1990. Effects of estuarine infauna on sediment stability and particle sedimentation, Hydrobiologia 190, 263–266. DOI: 10.1007/BF00008194
- Meentemeyer, R.K., Vogler, J.B., Butler, D.R., 1998. The geomorphic influences of burrowing beavers on streambanks, Bolin Creek, North Carolina. Zeitschrift für Geomorphologie 42, 453–468.
- Merrill, D., 1969. The distribution of case recognition behaviour in ten families of caddis larvae (Trichoptera). Animal Behaviour 17, 486–493. DOI: 10.1016/0003-3472(69)90151-1.
- Mermillod-Blondin, F., 2011. The functional significance of bioturbation and biodeposition on biogeochemical processes at the water-sediment interface in freshwater and marine ecosystems. Journal of the North American Benthological Society 30, 770–778. DOI: 10.1899/10-121.1.
- Mermillod-Blondin, F., Gérino, M., Creuzé des Châtelliers, M., Degrange, V., 2002. Functional Diversity among 3 Detritivorous Hyporheic Invertebrates: An Experimental Study in Microcosms. Journal of the North American Benthological Society 21, 132–149. DOI: 10.2307/1468305.

- Mermillod-Blondin, F., Lemoine, D.G., 2010. Ecosystem engineering by tubificid worms stimulates macrophyte growth in poorly oxygenated wetland sediments. Functional Ecology 24, 444–453. DOI: 10.1111/j.1365-2435.2009.01624.x
- Mermillod-Blondin, F., Rosenberg, R., 2006. Ecosystem engineering: the impact of bioturbation on biogeochemical processes in marine and freshwater benthic habitats. Aquatic Sciences 68, 434–442. DOI: 10.1007/s00027-006-0858-x.
- Meysman, F.J.R., Middelburg, J.J., Heip, C.H.R., 2006. Bioturbation: a fresh look at Darwin's last idea. Trends in Ecology & Evolution 21, 688–695. DOI: 10.1016/j.tree.2006.08.002.
- Michaud, E., Desrosiers, G., Mermillod-Blondin, F., Sundby, B., Stora, G., 2005. The functional group approach to bioturbation: The effects of biodiffusers and gallery-diffusers of the *Macoma balthica* community on sediment oxygen uptake. Journal of Experimental Marine Biology and Ecology 326, 77–88. DOI: 10.1016/j.jembe.2005.05.016.
- Montgomery, D.R., Buffington, J.M., Peterson, N.P., Schuett-Hames, D., Quinn, T.P., 1996. Stream-bed scour, egg burial depths, and the influence of salmonid spawning on bed surface mobility and embryo survival. Canadian Journal of Fisheries and Aquatic Sciences 53, 1061–1070. DOI: 10.1139/f96-028.
- Montgomery, D.R., Piégay, H., 2003. Wood in rivers: interactions with channel morphology and processes, Editorial. Geomorphology 51, 1–5. DOI:10.1016/S0169-555X(02)00322-7.
- Moore, J.W., 2006. Animal Ecosystem Engineers in Streams. Bioscience 56, 237–246. DOI: 10.1641/0006-3568(2006)056[0237:AEEIS]2.0.CO;2.
- Morris, M., Mohammadi, M.H., Day, S., Hondzo, M., Sotiropoulos, F., 2015. Prediction of *Glossosoma* biomass spatial distribution in Valley Creek by field measurements and a three-dimensional turbulent open-channel flow model. Water Resources Research 51, 1457–1471. DOI: 10.1002/2014WR015887.
- Morris, M., Hondzo, M., 2013a. *Glossosoma nigrior* (Trichoptera: Glossosomatidae) respiration in moving fluid. Journal of Experimental Biology 216, 3015–3022. DOI: 10.1242/jeb.082974.
- Morris, M., Hondzo, M., 2013b. Double-averaged rough-bed open-channel flow with high *Glossosoma* (Trichoptera: Glossosomatidae) abundance. Environmental Fluid Mechanics 13, 257–278. DOI: 10.1007/s10652-012-9265-0.
- Morris, M., Hondzo, M., Power, M.E., 2011. Scaling *Glossosoma* (Trichoptera) density by abiotic variables in mountain streams. Journal of the North American Benthological Society 30, 493–506. DOI: 10.1899/10-068.1.
- Morse, J., 2009. The Trichoptera World Checklist [Online]. Available from: https://entweb.sites.clemson.edu/database/trichopt/ (Accessed 7th June 2020).
- Morse, J.C., Frandsen, P.B., Graf, W., Thomas, J.A., 2019. Diversity and ecosystem services of trichoptera. Insects 10, 125. DOI: 10.3390/insects10050125.
- Murphy, J.F., Davy-Bowker, J., 2005. Spatial structure in lotic macroinvertebrate communities in England and Wales: relationship with physicochemical and anthropogenic stress variables. Hydrobiologia 534, 151–164. DOI: 10.1007/s10750-004-1451-8.
- Murray, J.M.H., Meadows, A., Meadows, P.S., 2002. Biogeomorphological implications of microscale interactions between sediment geotechnics and marine benthos: A review. Geomorphology 47, 15–30. DOI: 10.1016/S0169-555X(02)00138-1.

- Naimen, R.J., Johnston, C.A., Kelley, J.C., 1988. Alteration of North American Streams by Beaver. Bioscience 38, 753–762. DOI: 10.2307/1310784.
- Naman, S., Rosenfeld, J., Richardson J, 2016. Causes and consequences of invertebrate drift in running waters: from individuals to populations and trophic fluxes. Canadian Journal of Fisheries and Aquatic Sciences 73, 1292–1305. DOI: 10.1139/cjfas-2015-0363.
- Naylor, L.A., Viles, H.A., Carter, N.E.A., 2002. Biogeomorphology revisited: looking towards the future. Geomorphology 47, 3–14. DOI: 10.1016/S0169-555X(02)00137-X.
- Newson, M.D., Newson, C.L., 2000. Geomorphology, ecology and river channel habitat: mesoscale approaches to basin-scale challenges. Progress in Physical Geography 242, 195–217. DOI: 10.1177/030913330002400203.
- Nijboer, R., 2004. The ecological requirements of *Agapetus fuscipes* Curtis (Glossosomatidae), a characteristic species in unimpacted streams. Limnology Ecology and Environment of Inland Waters 34, 213–223. DOI: 10.1016/S0075-9511(04)80046-X.
- Nogaro, G., Mermillod-Blondin, F., Francois-Carcaillet, F., Gaudet, J.-P., Lafont, M., Gibert, J., 2006. Invertebrate bioturbation can reduce the clogging of sediment: an experimental study using infiltration sediment columns. Freshwater Biology 51, 1458–1473. DOI: 10.1111/j.1365-2427.2006.01577.x.
- Nogaro, G., Mermillod-Blondin, F., Valett, M.H., François-Carcaillet, F., Gaudet, J.-P., Lafont, M., Gibert, J., 2009. Ecosystem engineering at the sediment–water interface: bioturbation and consumer-substrate interaction. Oecologia 161, 125–138. DOI: 10.1007/s00442-009-1365-2
- Nortek, 2009. Vectrino User Guide. Nortek AS, Norway.
- Nunokawa, M., Gomi, T., Negishi, J.N., Nakahara, O., 2008. A new method to measure substrate coherent strength of Stenopsyche marmorata. Landscape and Ecological Engineering 4, 125–131. DOI: 10.1007/s11355-008-0044-5.
- Oakey, R.J., Green, M., Carling, P.A., Lee, M.W.E., Sear, D.A., Warburton, J., 2005. Grain-Shape Analysis--A New Method for Determining Representative Particle Shapes for Populations of Natural Grains. Journal of Sedimentary Research 75, 1065–1073. DOI: 10.2110/jsr.2005.079.
- Okano, J.-i., Kikuchi, E., Sasaki, O., 2010. The role of particle surface texture on case material selection and silk lining in caddis flies. Behavioral Ecology 21, 826–835. DOI: 10.1093/beheco/arq066.
- Okano, J.-i., Kikuchi, E., 2009. The effects of particle surface texture on silk secretion by the caddisfly *Goera japonica* during case construction. Animal Behaviour 77, 595–602. DOI: 10.1016/j.anbehav.2008.10.027.
- Okano, J.-i., Kikuchi, E., 2012. Effect of current velocity and case adaptations on the distribution of caddisfly larvae (*Glossosoma*, Trichoptera). Limnology 13, 37–43. DOI: 10.1007/s10201-011-0353-5.
- Okano, J.-i., Sasaki, O., Kano, H., 2016. The effects of surface roughness of sediment particles on the respiration of case-bearing caddisfly larvae. Freshwater science 35, 611–618. DOI: 10.1086/685713.

- Olden, J.D., Hoffman, A.L., Monroe, J.B., Poff, N.L.R., 2004. Movement behaviour and dynamics of an aquatic insect in a stream benthic landscape. Canadian Journal of Zoology 82, 1135–1146. DOI: 10.1139/Z04-094.
- Orlandini, S., Moretti, G., Albertson, J.D., 2015. Evidence of an emerging levee failure mechanism causing disastrous floods in Italy. Water Resources Research 51, 7995–8011. DOI: 10.1002/2015WR017426.
- Otto, C., 1971. Growth and Population Movements of *Potamophylax cingulatus* (Trichoptera) Larvae in a South Swedish Stream. Oikos 22, 293–301. DOI: 10.2307/3543853.
- Otto, C., 1976. Factors Affecting the Drift of *Potamophylax cingulatus* (Trichoptera) Larvae. Oikos 27, 93–100. DOI: 10.2307/3543437.
- Otto, C., 1975. Energetic Relationships of the Larval Population of *Potamophylax cingulatus* (Trichoptera) in a South Swedish Stream. Oikos 26, 159–169. DOI: 10.2307/3543705.
- Otto, C., Johansson, A., 1995. Why do some caddis larvae in running waters construct heavy, bulky cases? Animal Behaviour 49, 473–478. DOI: 10.1006/anbe.1995.0061.
- Otto, C., Svensson, B., 1980. The Significance of Case Material Selection for the Survival of Caddis Larvae. Journal of Animal Ecology 49, 855–865. DOI: 10.2307/4231.
- Otto, C., Svensson, B., 1981. Why do *Potamophylax Cingulatus* (Steph.) (Trichoptera) Larvae Aggregate at Pupation?, in: Moretti, G.P. (Ed.) Proceedings of the Third International Symposium on Trichoptera. Springer Netherlands, pp. 285–291. DOI: 10.1007/978-94-009-8641-1_37.
- Palmer, M., Ruhi, A., 2019. Linkages between flow regime, biota, and ecosystem processes: Implications for river restoration. Science 365, eaaw2087. DOI: 10.1126/science.aaw2087.
- Paphitis, D., 2001. Sediment movement under unidirectional flows: an assessment of empirical threshold curves, Coastal Engineering 43, 227–245. DOI: 10.1016/S0378-3839(01)00015-1.
- Parker, G., 2008. Transport of Gravel and Sediment Mixtures, Chapter 3, in: Garcia, M. (Ed.) Sedimentation Engineering: Theories, Measurements, Modelling and Practice. American Society of Civil Engineers, Reston. pp. 165–252. DOI: 10.1061/9780784408230.
- Parker, G., Sutherland, A.J., 1990. Fluvial armor. Journal of Hydraulic Research 28, 529–544. DOI: 10.1080/00221689009499044.
- Parkos III, J.J., Santucci, V.J., Wahl, D.H., 2003. Effects of adult common carp (*Cyprinus carpio*) on multiple trophic levels in shallow mesocosms. Canadian Journal of Fisheries and Aquatic Sciences 60, 182–192. DOI: 10.1139/F03-011.
- Perret, E., Berni, C., Camenen, B., Herrero, A., El Kadi Abderrezzak, K., 2017. Transport of moderately sorted gravel at low bed shear stresses: the role of fine sediment infiltration. Earth Surface Processes and Landforms 43, 1416–1430. DOI: 10.1002/esp.4322.
- Perret, E., Herrero, A., Berni, C., Abderrezzak, K. El, Camenen, B., 2015. Incipient motion of a bimodal mixture of gravel and silt: a laboratory experimental study, E-Proceedings of the 36th IAHR World Congress, The Hague, Netherlands. p. 11.

- Petit, F., 1994. Dimensionless critical shear stress evaluation from flume experiments using different gravel beds. Earth Surface Processes and Landforms 19, 565–576. DOI: 10.1002/esp.3290190608.
- Peyer, S.M., McCarthy, A.J., Lee, C.E., 2009. Zebra mussels anchor byssal threads faster and tighter than quagga mussels in flow. Journal of Experimental Biology 212, 2027–2036. DOI: 10.1242/jeb.028688.
- Pitlick, J., Mueller, E.R., Segura, C., Cress, R., Torizzo, M., 2008. Relation between flow, surfacelayer armoring and sediment transport in gravel-bed rivers. Earth Surface Processes and Landforms 33, 1192–1209. DOI: 10.1002/esp.1607.
- Pledger, A.G., Rice, S.P., Millett, J., 2016. Bed disturbance via foraging fish increases bedload transport during subsequent high flows and is controlled by fish size and species. Geomorphology 253, 83–93. DOI: 10.1016/j.geomorph.2015.09.021.
- Pledger, A.G., Rice, S.P., Millett, J., 2014. Reduced bed material stability and increased bedload transport caused by a foraging fish: a flume study with juvenile Barbel (*Barbus barbus*). Earth Surface Processes and Landforms 39, 1500–1513. DOI: 10.1002/esp.3592.
- Poff, N.L., Ward, J. V, 1988. Use Of Occupied Glossosoma verdona (Trichoptera, Glossosomatidae) Cases By Early Instars Of Baetis Spp (Ephemeroptera, Baetidae) In A Rocky mountain Stream. Entomological news 99, 97–101.
- Pollock, M., Lewallen, G., Woodruff, K., Jordan, C., Castro, J.M. (Eds.), 2015. The Beaver Restoration Guidebook Working with Beaver to Restore Streams, Wetlands, and Floodplains, (1.02. ed.). United States Fish and Wildlife Service, Portland.
- Pollock, M.M., Beechie, T.J., Wheaton, J.M., Jordan, C.E., Bouwes, N., Weber, N., Volk, C., 2014. Using Beaver Dams to Restore Incised Stream Ecosystems. Bioscience 64, 279– 290. DOI: 10.1093/biosci/biu036.
- Poreh, M., Sagiv, A., Seginer, I., 1970. Sediment Sampling Efficiency of Slots. Journal of the Hydraulics Division 96, 2065–2078.
- Powell, D.M., 1998. Patterns and processes of sediment sorting in gravel-bed rivers. Progress in Physical Geography 22, 1–32. DOI: 10.1177/030913339802200101.
- Pringle, C.M., Blake, G.A., Covich, A.P., Buzby, K.M., Finley, A., 1993. Effects of omnivorous shrimp in a montane tropical stream: sediment removal, disturbance of sessile invertebrates and enhancement of understory algal biomass. Oecologia 93, 1–11. DOI: 10.1007/BF00321183.
- Pringle, C.M., Hamazaki, T., 1998. The Role of Omnivory in a Neotropical Stream: Separating Diurnal and Nocturnal Effects. Ecology 79, 269–280. DOI: 10.1890/0012-9658(1998)079[0269:TROOIA]2.0.CO;2.
- Pringle, C.M., Naiman, R.J., Bretschko, G., Karr, J.R., Oswood, M.W., Webster, J.R., Welcomme, R.L., Winterbourn, M.J., 1988. Patch dynamics in lotic systems: the stream as a mosaic. Journal of the North American Benthological Society 7, 503–524. DOI: 10.2307/1467303.
- Puttock, A., Graham, H.A., Carless, D., Brazier, R.E., 2018. Sediment and nutrient storage in a beaver engineered wetland. Earth Surface Processes and Landforms 43, 2358–2370. DOI: 10.1002/esp.4398.

- R Core Team., 2017. A language and environment for statistical computing. Vienna, Austria. https://www.R-project.org/.
- R Core Team., 2018. A language and environment for statistical computing. Vienna, Austria. https://www.R-project.org/.
- Reid, A.J., Carlson, A.K., Creed, I.F., Eliason, E.J., Gell, P.A., Johnson, P.T.J., Kidd, K.A., MacCormack, T.J., Olden, J.D., Ormerod, S.J., Smol, J.P., Taylor, W.W., Tockner, K., Vermaire, J.C., Dudgeon, D., Cooke, S.J., 2019. Emerging threats and persistent conservation challenges for freshwater biodiversity. Biological Reviews 94, 849–873. DOI: 10.1111/brv.12480.
- Reid, I., Layman, J.T., Frostick, L.E., 1980. The Continuous Measurement Of Bedload Discharge. Journal of Hydraulic Research 18, 22–1686. DOI: 10.1080/00221688009499550
- Reid, S.C., Lane, S.N., Berney, J.M., Holden, J., 2006. The timing and magnitude of coarse sediment transport events within an upland, temperate gravel-bed river. Geomorphology 83, 152–182. DOI: 10.1016/j.geomorph.2006.06.030.
- Reinhardt, L., Jerolmack, D., Cardinale, B.J., Vanacker, V., Wright, J., 2010. Dynamic interactions of life and its landscape: feedbacks at the interface of geomorphology and ecology. Earth Surface Processes and Landforms 35, 78–101. DOI: 10.1002/esp.1912.
- Rennie, C.D., Millar, R.G., 2000. Spatial variability of stream bed scour and fill: a comparison of scour depth in chum salmon (*Oncorhynchus keta*) redds and adjacent bed. Canadian Journal of Fisheries and Aquatic Sciences 57, 928–938. DOI: 10.1139/cjfas-57-5-928.
- Ricciardi, A., Whoriskey, F.G., Rasmussen, J.B., 1997. The role of the zebra mussel (*Dreissena polymorpha*) in structuring macroinvertebrate communities on hard substrata. Canadian Journal of Fisheries and Aquatic Sciences 54, 2596–2608. DOI: 10.1139/cjfas-54-11-2596.
- Rice, S.P., Buffin-Bélanger, T., Lancaster, J., Reid, I., 2007. 24 Movements of a macroinvertebrate (*Potamophylax latipennis*) across a gravel-bed substrate: effects of local hydraulics and micro-topography under increasing discharge. in Habersack, H., Piegay, H., Rinaldi, M., eds. Gravel-Bed Rivers VI: From Process Understanding to River Restoration. Elsevier Science, Amsterdam. pp. 637–660.
- Rice, S.P., Church, M., 1998. Grain size along two gravel-bed rivers: statistical variation, spatial pattern and sedimentary links. Earth Surface Processes and Landforms 23, 345–363. DOI: 10.1002/(SICI)1096-9837(199804)23:4<345::AID-ESP850>3.0.CO;2-B.
- Rice, S.P., Johnson, M.F., Extence, C., Reeds, J., Longstaff, H., 2014. Diel patterns of suspended sediment flux and the zoogeomorphic agency of invasive crayfish. Cuadernos de Investigación Geográfica 40, 7–26. DOI: 10.18172/cig.2508.
- Rice, S.P., Johnson, M.F., Mathers, K., Reeds, J., Extence, C., 2016. The importance of biotic entrainment for base flow fluvial sediment transport. Journal of Geophysical Research: Earth Surface 121, 890–906. DOI: 10.1002/2015JF003726.
- Rice, S.P., Johnson, M.F., Reid, I., 2012. Animals and the Geomorphology of Gravel Bed Rivers, in: Church, M., Biron, P.M., André, G.R. (Eds.), Gravel-Bed Rivers: Processes, Tools, Environments. John Wiley and Sons, Oxford, pp. 225–241.

- Rice, S.P., Lancaster, J., Kemp, P., 2010. Experimentation at the interface of fluvial geomorphology, stream ecology and hydraulic engineering and the development of an effective, interdisciplinary river science. Earth Surface Processes and Landforms 35, 64– 77. DOI: 10.1002/esp.1838.
- Rice, S.P., Pledger, A., Toone, J., Mathers, K., 2019. Zoogeomorphological behaviours in fish and the potential impact of benthic feeding on bed material mobility in fluvial landscapes. Earth Surface Processes and Landforms 44, 54–66. DOI: 10.1002/esp.4541.
- Richards, P.J., Humphreys, G.S., 2010. Burial and turbulent transport by bioturbation: a 27-year experiment in southeast Australia. Earth Surface Processes and Landforms 35, 856–862. DOI: 10.1002/esp.2007.
- Richardson, K., Benson, I. and Carling, P.A. (2003) An instrument to record sediment movement in bedrock channels. in Bogen, J., Fergus, T. and Walling, D.E. (eds.) Erosion and Sediment Transport Measurement in Rivers: Technological and Methodological Advances. ICE/IAHS. pp. 228-235.
- Roche, A., Vennin, E., Bundeleva, I., Bouton, A., Payandi-Rolland, D., Amiotte-Suchet, P., Gaucher, E.C., Courvoisier, H., Visscher, P.T., 2019. The role of the substrate on the mineralization potential of microbial mats in a modern freshwater River (Paris Basin, France). Minerals 9, 359. DOI: 10.3390/min9060359.
- Rowe, C.L., Sadinski, W.J., Dunson, W.A., 1994. Predation on Larval and Embryonic Amphibians by Acid-Tolerant Caddisfly Larvae (*Ptilostomis postica*). Journal of Herpetology 28, 357. DOI: 10.2307/1564535.
- Rowlands, M., Hansell, M., 1987. Case design, construction and ontogeny of building in *Glyphotaelius pellucidus* caddisfly larvae. Journal of Zoology 211, 329356.
- Ruedemann, R., Schoonmaker, W.J., 1938. Beaver-dams as geologic agents. Science 88, 523–525. DOI: 10.1126/science.88.2292.523.
- Rust, B.R., Nanson, G.C., 1989. Bedload transport of mud as pedogenic aggregates in modern and ancient rivers. Sedimentology 36, 291–306. DOI: 10.1111/j.1365-3091.1989.tb00608.x.
- Savrda, C.E., 2019. Bioerosion of a modern bedrock stream bed by insect larvae (Conecuh River, Alabama): Implications for ichnotaxonomy, continental ichnofacies, and biogeomorphology. Palaeogeography, Palaeoclimatology, Palaeoecology 513, 3–13. DOI: 10.1016/j.palaeo.2017.06.030.
- Schumm, S.A., 1977. The Fluvial System. Wiley, New York.
- Scott, D., 1958. Ecological studies on the Trichoptera of the River Dean, Cheshire. Archiv für Hydrobiologie, 54, 340-392.
- Seal, J.N., Tschinkel, W.R., 2006. Colony Productivity of the Fungus-Gardening Ant *Trachymyrmex septentrionalis* (Hymenoptera: Formicidae) in a Florida Pine Forest. Annals of the Entomological Society of America 99, 673–682. DOI: 10.1603/0013-8746(2006)99[673:CPOTFA]2.0.CO;2.
- Sear, D.A., 1994. River restoration and geomorphology. Aquatic Conservation: Marine and Freshwater Ecosystems 4, 169–177. DOI: 10.1002/aqc.3270040207.
- Shields, A., 1936. Anwendung der Aehnlichkeitsmechanik und der Turbulenzforschung auf die Geschiebebewegung. PhD Thesis, Technical University Berlin.

- Shull, D.H., 2009. Bioturbation. in Steele, J.H. (Ed.) Encyclopedia of Ocean Sciences (2nd Ed.), Academic Press, Oxford. pp. 395-400, DOI: 10.1016/B978-012374473-9.00656-1.
- Simões, F.J.M., 2014. Shear velocity criterion for incipient motion of sediment. Water Science and Engineering 7, 183–193. DOI: 10.3882/j.issn.1674-2370.2014.02.006.
- Simon, S.S.T., Gibling, M.R., 2017. Pedogenic Mud Aggregates Preserved In A Fine-Grained Meandering Channel In the Lower Permian Clear Fork Formation, North-Central Texas, U.S.A. Journal of Sedimentary Research 87, 230–252. DOI: 10.2110/jsr.2017.12.
- Skelly, D.K., 2002. Experimental Venue and Estimation of Interaction Strength. Ecology 83, 2097–2101. DOI: 10.1890/0012-9658(2002)083[2097:EVAEOI]2.0.CO;2.
- Skubinna, J.P., Coon, T.G., Batterson, T.R., 1995. Increased Abundance and Depth of Submersed Macrophytes in Response to Decreased Turbidity in Saginaw Bay, Lake Huron. Journal of Great Lakes Research 21, 476–488. DOI: 10.1016/S0380-1330(95)71060-7.
- Soluk, D.A., Craig, D.A., 1990. Digging with a vortex: Flow manipulation facilitates prey capture by a predatory stream mayfly. Limnology and Oceanography 35, 1201–1206. DOI: 10.4319/lo.1990.35.5.1201.
- Soulsby, R.L., 1983. The bottom boundary layer of shelf seas. In Johns, B. (Ed.) Physical Oceanography of Coastal and Shelf Seas, Elsevier Science Publishers, Amsterdam. pp. 189-266. DOI: 10.1016/S0422-9894(08)70503-8.
- Southwood, T.R.E., 1977. Habitat, the templet for ecological strategies? Journal of Animal Ecology 46, 337–365. DOI: 10.2307/3817.
- Stallins, J.A., 2006. Geomorphology and ecology: Unifying themes for complex systems in biogeomorphology. Geomorphology 77, 207–216. DOI: 10.1016/j.geomorph.2006.01.005.
- Stapleton, K.R., Huntley, D.A., 1995. Seabed stress determinations using the inertial dissipation method and the turbulent kinetic energy method. Earth Surface Processes and Landforms 20, 807–815. DOI: 10.1002/esp.3290200906.
- Statzner, B., 2012. Geomorphological implications of engineering bed sediments by lotic animals. Geomorphology 157–158, 49–65. DOI: 10.1016/j.geomorph.2011.03.022.
- Statzner, B., 2011. Mineral grains in caddisfly pupal cases and streambed sediments: assessing resource use and its limitation across various river types. International Journal of Limnology 47, 103–118. DOI: 10.1051/limn/2011004.
- Statzner, B., Arens, M.F., Champagne, J.Y., Morel, R., Herouin, E., 1999. Silk-producing stream insects and gravel erosion: Significant biological effects on critical shear stress. Water Resources Research 35, 3495–3506. DOI: 10.1029/1999WR900196.
- Statzner, B., Dolédec, O., Sagnes, P., 2009. Recent Low-Cost Technologies to Analyse Physical Properties of Cases and Tubes Built by Aquatic Animals. International Review of Hydrobiology 94, 625–644. DOI: 10.1002/iroh.200911165.
- Statzner, B., Dolédec, S., 2011. Mineral grain availability and pupal-case building by lotic caddisflies: Effects on case architecture, stability and building expenses. Limnologica 41, 266–280. DOI: 10.1016/j.limno.2011.04.002.

- Statzner, B., Fièvet, E., Champagne, J.-Y., Morel, R., Herouin, E., 2000. Crayfish as geomorphic agents and ecosystem engineers: Biological behavior affects sand and gravel erosion in experimental streams. Limnology and Oceanography 45, 1030–1040. DOI: 10.4319/lo.2000.45.5.1030.
- Statzner, B., Fuchs, U., Higler, L.W.G., 1996. Sand erosion by mobile predaceous stream insects: Implications for ecology and hydrology. Water Resources Research 32, 2279–2287. DOI: 10.1029/96WR00977.
- Statzner, B., Gore, J.A., Resh, V.H., 1988. Hydraulic Stream Ecology: Observed Patterns and Potential Applications. Journal of the North American Benthological Society 7, 307–360. DOI: 10.2307/1467296.
- Statzner, B., Higler, B., 1986. Stream hydraulics as a major determinant of benthic invertebrate zonation patterns. Freshwater Biology 16, 127–139. DOI: 10.1111/j.1365-2427.1986.tb00954.x.
- Statzner, B., Holm, T.F., 1989. Morphological adaptation of shape to flow: Microcurrents around lotic macroinvertebrates with known Reynolds numbers at quasi-natural flow conditions. Oecologia 78, 145–157. DOI: 10.1007/BF00377150.
- Statzner, B., Mérigoux, S., Leichtfried, M., 2005. Mineral grains in caddisfly pupal cases and streambed sediments: Resource use and its limitation through conflicting resource requirements. Limnology and Oceanography 50, 713–721. DOI: 10.4319/lo.2005.50.2.0713.
- Statzner, B., Peltret, O., 2006. Assessing potential abiotic and biotic complications of crayfishinduced gravel transport in experimental streams. Geomorphology 74, 245–256. DOI: 10.1016/j.geomorph.2005.08.007.
- Statzner, B., Sagnes, P., Champagne, J.-Y., Viboud, S., 2003. Contribution of benthic fish to the patch dynamics of gravel and sand transport in streams. Water Resources Research 39, 1309. DOI: 10.1029/2003WR002270.
- Sterling, S.M., Church, M., 2002. Sediment trapping characteristics of a pit trap and the Helley-Smith sampler in a cobble gravel bed river. Water Resources Research 38, 19-1–19-11. DOI: 10.1029/2000WR000052.
- Stevens, D.J., Hansell, M.H., Freel, J.A., Monaghan, P., 1999. Developmental trade-offs in caddis flies: increased investment in larval defence alters adult resource allocation. Proceedings of the Royal Society of London 266, 1049–1054. DOI: 10.1098/rspb.1999.0742.
- Strayer, D., Caraco, N.F., Cole, J.J., Findlay, S., Pace., M.L., 1999. Transformation of Freshwater Ecosystems by Bivalves: A case study of zebra mussels in the Hudson River. Bioscience 49, 19–27. DOI: 10.1525/bisi.1999.49.1.19.
- Strayer, D.L., Dudgeon, D., 2010. Freshwater biodiversity conservation: recent progress and future challenges. Journal of the North American Benthological Society 29, 344–358. DOI: 10.1899/08-171.1.
- Surber, E.W., 1937. Rainbow Trout and Bottom Fauna Production in One Mile of Stream. Transactions of the American Fisheries Society 66, 193–202. DOI: 10.1577/1548-8659(1936)66[193:RTABFP]2.0.CO;2.

- Taccari, M.L., Van Der Meij, R., 2016. Investigation of the influence of animal burrowing on the failure of the levee of San Matteo along the Secchia river. 3rd European Conference on Unsaturated Soils 9, 19001. DOI: 10.1051/e3sconf/20160919001.
- Taghon, G.L., Nowell, A.R.M., Jumars, P.A., 1984. Transport and breakdown of faecal pellets: Biological and sedimentological consequences. Limnology and Oceanography 29, 64–72. DOI: 10.4319/lo.1984.29.1.0064.
- Takao, A., Negishi, J.N., Nunokawa, M., Gomi, T., Nakahara, O., 2006. Potential influences of a net-spinning caddisfly (Trichoptera: *Stenopsyche marmorata*) on stream substratum stability in heterogeneous field environments. Journal of the North American Benthological Society 25, 545–555. DOI: 10.1899/0887-3593(2006)25[545:PIOANC]2.0.CO;2.
- Tal, M., Paola, C., 2010. Effects of vegetation on channel morphodynamics: results and insights from laboratory experiments. Earth Surface Processes and Landforms 35, 1014–1028. DOI: 10.1002/esp.1908.
- Tashiro, T., Watanabe, S., Tsujimoto, T., 2005. Numerical analysis of the population dynamics of netspinning caddis larvae and the substratum adhesion due to their inhabitation in a cobble river with fewer disturbances. Proceedings of Hydraulic Engineering 49, 1453– 1458. DOI: 10.2208/prohe.49.1453.
- Thomas, R.E., Johnson, M.F., Frostick, L.E., Parsons, D.R., Bouma, T.J., Dijkstra, J.T., Eiff, O., Gobert, S., Henry, P.-Y., Kemp, P., Mclelland, S.J., Moulin, F.Y., Myrhaug, D., Neyts, A., Paul, M., Ellis Penning, W., Puijalon, S., Rice, S.P., Stanica, A., Tagliapietra, D., Tal, M., Tørum, A., Vousdoukas io, M.I., 2014. Physical modelling of water, fauna and flora: knowledge gaps, avenues for future research and infrastructural needs. Journal of Hydraulic Research 52, 311–325. DOI: 10.1080/00221686.2013.876453.
- Thoms, M.C., Parsons, M., 2002. Eco-geomorphology: an interdisciplinary approach to river science. International Association of Hydrological Sciences, Publication No 276, pp. 113-119.
- Tolkamp, H.H., 1980. Organism-substrate relationships in lowland streams. Agricultural Research Reports 907, Wageningen, p. 211.
- Townsend, C.R., Hildrew, A.G., 1994. Species traits in relation to a habitat templet for river systems. Freshwater Biology 31, 265–275. DOI: 10.1111/j.1365-2427.1994.tb01740.x.
- Tszydel, M., Sztajnowski, S., Michalak, M., Wrzosek, H., Kowalska, S., Krucińska, I., Lipp-Symonowicz, B., 2009. Structure and physical and chemical properties of fibres from the fifth larval instar of caddis-flies of the species *Hydropsyche angustipennis*. Fibres and Textiles of Eastern Europe 6, 7–12.
- Uchman, A., Mikuláš, R., Stachacz, M., 2017. Mayfly Burrows in Firmground of Recent Rivers from the Czech Republic and Poland, with Some Comments on Ephemeropteran Burrows in General. Ichnos 24, 191–203. DOI: 10.1080/10420940.2016.1257488.
- Van Der Lee, G.H., Michiel, ·, Kraak, H.S., Verdonschot, R.C.M., Verdonschot, P.F.M., 2020. Persist or perish: critical life stages determine the sensitivity of invertebrates to disturbances 82, 24. DOI: 10.1007/s00027-020-0698-0.
- Vandenbruwaene, W., Temmerman, S., Bouma, T.J., Klaassen, P.C., De Vries, M.B., Callaghan, D.P., Van Steeg, P., Dekker, F., Van Duren, L.A., Martini, E., Balke, T., Biermans, G., Schoelynck, J., Meire, P., 2011. Flow interaction with dynamic vegetation patches:
Implications for biogeomorphic evolution of a tidal landscape. Journal of Geophysical Research: Earth Surface 116, 1–13. DOI: 10.1029/2010JF001788.

- Vannote, R.L., Minshall, G.W., Cummins, K.W., Sedell, J.R., Gushing, C.E., 1980. The River Continuum Concept. Canadian Journal of Fisheries and Aquatic Sciences 37, 130–137. DOI: 10.1139/f80-017.
- Vaughn, C.C., Spooner, D., 2006. Unionid mussels influence macroinvertebrate assemblage structure in streams. Journal of the North American Benthological Society 25, 691–700. DOI: 10.1899/0887-3593(2006)25.
- Venditti, J.G., Dietrich, W.E., Nelson, P.A., Wydzga, M.A., Fadde, J., Sklar, L., 2010. Mobilization of coarse surface layers in gravel-bedded rivers by finer gravel bed load. Water Resources Research 46, 1–10. DOI: 10.1029/2009WR008329.
- Viles, H.A., 2019. Biogeomorphology: Past, present and future. Geomorphology, 106809. DOI: 10.1016/j.geomorph.2019.06.022.
- Viles, H.A., 1988. Biogeomorphology. Basil Blackwell, Oxford.
- Voulgaris, G., Trowbridge, J.H., 1998. Evaluation of the Acoustic Doppler Velocimeter (ADV) for Turbulence Measurements. Journal of Atmospheric and Oceanic Technology 15, 272–289. DOI: 10.1175/1520-0426(1998)015<0272:eotadv>2.0.co;2.
- Wagner, R., 1991. The influence of the diel activity pattern of the larvae of Sericostoma personatum (Kirby & Spence) (Trichoptera) on organic matter distribution in stream-bed sediments a laboratory study. Hydrobiologia 224, 6570. DOI: 10.1007/BF00006862.
- Wagner, R., 1990. A laboratory study on the life cycle of Sericostoma personatum (Kirby & Spence), and light dark-dependent food consumption. Hydrobiologia 208, 201–212. DOI: 10.1007/BF00007785.
- Wagner, R.H., 1987. Effects of an artificially silted stream bottom on species composition and biomass of Trichoptera in Breitenbach, in: Bournaud, M., Tachet, H. (Eds.), Proceedings of the Fifth International Symposium on Trichoptera, Lyon, France. Dr W. Junk Publishers, Dordrecht.
- Wallace, I.D., 2016. A review of the status of the caddis flies (Trichoptera) of Great Britain -Species Status No.27. Natural England Commissioned Reports, Number191.
- Wallace, I.D., Wallace, B., Philipson, G.N., 2003. Keys to the case-bearing caddis larvae of Britain and Ireland. Freshwater Biological Association, Ambleside. DOI: 10.1111/j.1365-2427.2004.01195.x.
- Wallace, J.B., Webster, J.R., 1996. The Role of Macroinvertebrates in Stream Ecosystem Function. Annual Review of Entomology 41, 115–139. DOI: 10.1146/annurev.en.41.010196.000555.
- Wallace, J.B., Webster, J.R., Woodall, W.R., 1977. The Role of Filter Feeders in Flowing Waters. Archiv für Hydrobiologie 79, 506–532.
- Wang, J.Q., Zhang, X.D., Jiang, L.F., Bertness, M.D., Fang, C.M., Chen, J.K., Hara, T., Li, B., 2010. Bioturbation of burrowing crabs promotes sediment turnover and carbon and nitrogen movements in an estuarine salt marsh. Ecosystems 13, 586–599. DOI: 10.1007/s10021-010-9342-5.

- Ward, J. V, 1989. The four-dimensional nature of lotic ecosystems. Journal of the North American Benthological Society 8, 2–8. DOI: 10.2307/1467397.
- Waringer, J.A., 1993. The drag coefficient of cased caddis larvae from running waters: experimental determination and ecological applications. Freshwater Biology 29, 419–427. DOI: 10.1111/j.1365-2427.1993.tb00776.x.
- Waringer, J.A., 1989. Resistance of a cased caddis larva to accidental entry into the drift: the contribution of active and passive elements. Freshwater Biology 21, 411–420. DOI: 10.1111/j.1365-2427.1989.tb01373.x
- Waters, T.F., 1995. Sediment in Streams: Sources, Biological Effects and Control. Monograph 7. American Fisheries Society, Bethesda.
- Webster, D.A., Webster, P.C., 1943. Influence of water current on case weight in larvae of the caddisfly, *Goera Calcarata* Banks. The Canadian Entomologist 75, 105–108. DOI: 10.4039/Ent75105-6.
- Wellnitz, T.A., Poff, N.L., Cosyleón, G., Steury, B., 2001. Current velocity and spatial scale as determinants of the distribution and abundance of two rheophilic herbivorous insects. Landscape Ecology 16,111–120. DOI: 10.1023/A:1011114414898.
- Wentworth, C.K., 1922. A Scale of Grade and Class Terms for Clastic Sediments. The Journal of Geology 30, 377–392.
- Wharton, G., Mohajeri, S.H., Righetti, M., 2017. The pernicious problem of streambed colmation: a multi-disciplinary reflection on the mechanisms, causes, impacts, and management challenges. Wiley Interdisciplinary Reviews: Water 4, e1231. DOI: 10.1002/wat2.1231.
- Whitford, W.G., Eldridge, D.J., 2013. Effects of Ants and Termites on Soil and Geomorphological Processes. Treatise on Geomorphology 12, 281–292. DOI: 10.1016/B978-0-12-374739-6.00335-3.
- Wiggins, G.B., 2004. Caddisflies: The Underwater Architects. University of Toronto Press, Toronto.
- Wiggins, G.B., 1996. Trichoptera Families, in: Merritt, R.W., Cummings, K.W. (Eds.), An Introduction to the Aquatic Insects of North America. Kendall/Hunt Publishing Company, Dubuque, pp. 309–349.
- Wiggins, G.B., Mackay, R.J., 1978. Some Relationships between Systematics and Trophic Ecology in Nearctic Aquatic Insects, with Special Reference to Trichoptera. Ecology 59, 1211–1220. DOI: 10.2307/1938234
- Wilcock, P.R., Crowe, J.C., 2003. Surface-based Transport Model for Mixed-Size Sediment. Journal of Hydraulic Engineering 129, 120–128. DOI: 10.1061/(ASCE)0733-9429(2003)129:2(120).
- Wilcock, P., Pitlick, J., Cui, Y., 2009. Sediment transport primer: estimating bed-material transport in gravel-bed rivers. General Technical Report RMRS-GTR-226, U.S. Department of Agriculture, Forest Service, Fort Collins. 78 p.
- Wilcock, P.R., Kenworthy, S.T., 2002. A two-fraction model for the transport of sand/gravel mixtures. Water Resources Research 38, 12-1-12–12. DOI: 10.1029/2001WR000684.

- Wilcock, P.R., Kenworthy, S.T., Crowe, J.C., 2001. Experimental study of the transport of mixed sand and gravel. Water Resources Research 37, 3349–3358. DOI: 10.1029/2001WR000683.
- Wilkes, M.A., Gittins, J., Mathers, K.L., Mason, R., Murray-Bligh, J., England, J., 2019. Physical and biological controls on fine sediment transport and storage in rivers. WIRESs Water 6, e1331. DOI: 10.1002/wat2.1331.
- Wilkinson, M.T., Richards, P.J., Humphreys, G.S., 2009. Breaking ground: Pedological, geological, and ecological implications of soil bioturbation. Earth-Science Reviews 97, 257–272. DOI: 10.1016/j.earscirev.2009.09.005.
- Williams, D.D., Tavares, A.F., Eric, B., 1987. Respiratory device or camouflage? A case for the caddisfly. Oikos 50, 42–52. DOI: 10.2307/3565400.
- Williams, D.D., Williams, N.E., 1993. The upstream/downstream movement paradox of lotic invertebrates: quantitative evidence from a Welsh mountain stream. Freshwater Biology 30, 199–218. DOI: 10.1111/j.1365-2427.1993.tb00802.x.
- Wohl, E., 2013. Floodplains and wood. Earth-Science Reviews 123, 194–212. DOI: 10.1016/J.EARSCIREV.2013.04.009.
- Wohl, E., Lane, S.N., Wilcox, A.C., 2015. The science and practice of river restoration. Water Resources Research 51, 5974–5997. DOI: 10.1002/2014WR016874.
- Wolman, M.G., 1954. A method of sampling coarse river-bed material. Eos, Transactions American Geophysical Union 35, 951–956. DOI: 10.1029/TR035i006p00951.
- Wood, P.J., Armitage, P.D., 1999. Sediment deposition in a small lowland stream-management implications. Regulated Rivers Research & Management 15, 199–210. DOI: 10.1002/(sici)1099-1646(199901/06)15:1/3<199::aid-rrr531>3.0.co;2-0.
- Wotton, R.S., 2011. EPS (Extracellular Polymeric Substances), silk, and chitin: vitally important exudates in aquatic ecosystems. Journal of the North American Benthological Society 30, 762–769. DOI: 10.1899/10-120.1.
- Wright, J., Jones, C., Flecker, A., 2002. An ecosystem engineer, the beaver, increases species richness at the landscape scale. Oecologia 132, 96–101. DOI: 10.1007/s00442-002-0929-1.
- WWF, 2016. Living Planet Report 2016: Risk and resilience in a new era. WWF International, Gland.
- Xie, M., Wang, N., Gaillard, J.F., Packman, A.I., 2018. Interplay between flow and bioturbation enhances metal efflux from low-permeability sediments. Journal of Hazardous Materials 341, 304–312. DOI: 10.1016/j.jhazmat.2017.08.002.
- Xing, X., Liu, L., Yan, W., Wu, T., Zhao, L., Wang, X., 2018. Bioturbation effects of Chironomid Larvae on nitrogen release and ammonia-oxidizing bacteria abundance in sediments. Water 10, 512. DOI: 10.3390/w10040512.
- Yang, Y., Gao, S., Wang, Y.P., Jia, J., Xiong, J., Zhou, L., 2019. Revisiting the problem of sediment motion threshold. Continental Shelf Research 187, 103960. DOI: 10.1016/j.csr.2019.103960.

- Zanetell, B.A., Peckarsky, B.L., 1996. Stoneflies as ecological engineers Hungry predators reduce fine sediments in stream beds. Freshwater Biology 36, 569–577. DOI: 10.1046/j.1365-2427.1996.00132.x.
- Zhang, L., Shang, J., He, W., You, B., Fan, C., 2014. The role of tubificid worms (*Limnodrilus hoffmeisteri*) in sediment resuspension: a microcosm study. International Journal of Limnology 50, 253–260. DOI: 10.1051/limn/2014013.
- Zimmerman, G.F., de Szalay, F.A., 2007. Influence of unionid mussels (Mollusca: Unionidae) on sediment stability: an artificial stream study. Archiv für Hydrobiologie 168, 299–306.
 DOI: 10.1127/1863-9135/2007/0168-0299.Zwick, P., Wagner, R., Reidelbach, J., Christl, H., 2011. The Fauna of the Breitenbach, in: Wagner, R., Marxsen, J., Zwick, P., Cox, E. (Eds.), Central European Stream Ecosystems: The Long Term Study of the Breitenbach. Wiley-Blackwell, Weinheim. Pp. 195–485. DOI: 10.1002/9783527634651.ch7.